

Groundwater flows in an urbanised floodplain and implications for environmental management

A thesis submitted for the degree of Doctor of Philosophy School of Archaeology, Geography and Environmental Science, University of Reading

> David M.J. Macdonald June 2019

Declaration

I confirm that this is my own work and the use of all material from other sources has been properly and fully acknowledged.

24 June 2019

Contents

Declaration	i
List of Figure	siii
List of Tables	siv
Supporting p	vublicationsv
Acknowledg	ementsvii
Abstract	ix
Chapter 1: Ir	ntroduction1
1.1 Riv	verine floodplains3
1.1.1	Riverine floodplain services
1.1.2	Groundwater flows in floodplain aquifers5
1.1.3	Modelling flooding in permeable floodplains9
1.1.4	Pollution11
1.1.5	Environmental management drivers relevant to floodplain environments
1.2 Re	search gaps addressed, and thesis aims and structure16
1.2.1	Research gaps addressed16
1.2.2	Thesis aims and structure
1.3 Stu	udy area21
Chapter 2: W C	/ater and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications25
Chapter 3: N	itrogen sources, transport and processing in peri-urban floodplains
Chapter 4: G	roundwater flooding within an urbanised flood plain55
Chapter 5: P g	ersistence of flooding on permeable lowland floodplains: the role of shallow roundwater and surface infiltration71
Chapter 6: C	onclusions
6.1 Ad	dressing aims, and implications for environmental management105
6.2 Fu	rther work
References	

List of Figures

Figure 1	River management structures in England and Wales provided by the Environment Agency
	(EA), March 2014
Figure 2	Historical landfill sites in England and Wales, licensed by the Environment Agency 2014,
	including those located on riverine floodplains
Figure 3	Bedrock geology within the River Thames catchment, categorised as carbonate, sandstone
	and mudstone, and overlying superficial alluvial deposits, coloured according to bedrock
	category
Figure 4	Features of the floodplain of the River Thames in the vicinity of Oxford, UK

List of Tables

Supporting publications

Refereed publications

Included in the thesis

- Macdonald, D.M.J., Dixon, A.J., & Gooddy, D.C. (2018). Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications. *Ecological Engineering*, 123, 226-237.
- Gooddy, D.C., Macdonald, D.M.J., Lapworth, D.J., Bennett, S.A., & Griffiths, K.J. (2014). Nitrogen sources, transport and processing in peri-urban floodplains. *Science of the Total Environment*, 494, 28-38.
- Macdonald, D.M.J., Dixon, A., Newell, A., & Hallaways, A. (2012). Groundwater flooding within an urbanised flood plain. *Journal of Flood Risk Management*, 5(1), 68-80.
- Macdonald, D.M.J., Jackson, C.R., Wade, A.J. (in preparation). Persistence of flooding on permeable lowland floodplains: the role of shallow groundwater and surface infiltration.

Complementary to the thesis

- Ascott, M.J., Marchant, B.P., Macdonald, D.M.J., McKenzie, A.A., Bloomfield, J.P. (2017). Improved understanding of spatio-temporal controls on regional scale groundwater flooding using hydrograph analysis and impulse response functions. *Hydrological Processes*, 31(25), 4586-4599.
- Punalekar, S., Verhoef, A., Tatarenko, I.V., van der Tol, C., Macdonald, D.M.J., Marchant, B., Gerard, F., White, K., Gowing, D., 2016. Characterization of a highly biodiverse floodplain meadow using hyperspectral remote sensing within a plant functional trait framework. *Remote Sensing*, 8(2), 112.

Articles and reports

- Macdonald, D. M. J., Jackson, C. R., & Mansour, M. M. (2018). A groundwater flow model of the alluvial sands and gravels aquifer of the River Thames floodplain in the vicinity of Oxford, UK. *British Geological Survey Open Report*, OR/18/067.
- NERC (2017). Groundwater: the threat beneath our feet. *Planet Earth NERC*, Winter 2016-17.
- ESI (2016). Oxford Flood Alleviation Scheme: Groundwater Flood Modelling. *ESI Ltd*. Confidential Report to CH2MHill.
- White, D., & Macdonald, D.M.J. (2015). Review of data relevant to the characterisation of leachate from Burgess Field waste dump, Oxford. *British Geological Survey Internal Report*, IR/15/037.
- Macdonald, D.M.J., Griffiths, K.J., Lapworth, D.J., Williams, P.J., Stuart, M.E., Gooddy, D.C. (2012). Hydrogeochemical characterisation of a peri-urban floodplain: initial findings. *British Geological Survey Internal Report*, IR/12/049.

Acknowledgements

I would like to thank all those colleagues from the British Geological Survey (BGS) who have helped collect data within the Oxford floodplain during my PhD or supported my research, including Peter Williams, Debbie White, Barry Townsend, Dan Lapworth, Kate Griffiths, Sarah Bennett, Andy Newell, Majdi Mansour and Carole Sharratt. I am very grateful to the Learning & Development Section within BGS which has helped to fund my PhD though the payment of fees and the provision of some of my staff time. I would like to thank senior BGS colleagues, John Bloomfield, Fiona McEvoy and Rob Ward, for their encouragement, and particularly Daren Gooddy whose suggestion it was to undertake this PhD, who provided advice and support throughout, and with whom much of the research was undertaken. I would also like to thank my BGS PhD supervisor Chris Jackson. His help has been invaluable, and the time he contributed to the studies, so much more than I could have hoped for.

I would also like to thank those others that have contributed to the gathering of data and evidence that was used in my PhD, to: the Environment Agency for the data provided and for the opportunity to collaborate with them on flood risk management studies in Oxford; Oxford City Council, for access to its land to carry out research activities; the residents of Oxford who have helped in the collection of monitoring data and in the sharing of knowledge on flooding events in the city; and the staff of Groundwater Monitoring and Drilling Ltd. (GMD), who for many years have persevered to help produce a unique water level dataset for the Oxford floodplain. I would like to say a special thanks to Andy Dixon of GMD. His dedication to improve the understanding of groundwater hydrology in the Oxford area has been outstanding and I have learnt so much from him.

I would like to thank my University of Reading supervisors Stuart Black and, especially, Andrew Wade for his time, guidance and encouragement throughout the PhD. I would also like to thank my PhD examiners, Stefan Krause of the University of Birmingham, and Elizabeth Stephens of the University of Reading for their suggested revisions to this thesis.

Finally, I would like to thank my close friends and family for their support, in particular my parents, and my sons Willem and Ben, who have been so patient with their father who has spent so much of the last seven years upstairs in the study. But most of all I have to give my greatest thanks to my wonderful wife, Anne. She has been so encouraging, supportive and understanding, and without her this PhD would not have been completed.

Abstract

With population growth, the large lowland floodplains of our major rivers have become increasingly urbanised. Environmental issues have arisen with the juxtaposition of these urban developments and the aquatic and terrestrial ecosystems associated with the rivers and their floodplains. The floodplain sediments are often highly permeable, and hydraulically well-connected to the water courses, and therefore the interaction of the urban environment and groundwater is very important.

The overall aim of this PhD is improved understanding of the hydrological regime of urbanised floodplains, in particular groundwater hydrology, leading to better environmental management. The PhD uses as a case study, the floodplain of the River Thames in the city of Oxford. Through surveys, data collection via an extensive monitoring network, and the development of conceptual and numerical models, the floodplain has been characterised and the hydrological processes better understood. Focussed studies, working with key stakeholders, have been undertaken relating to fluxes of pollutants into and through the subsurface, and to the role of groundwater in urban flooding.

The research undertaken has resulted in a better understanding of:

- the impact of river management structures on water and nitrate exchange between rivers and floodplain aquifers;
- the influence of legacy waste dumps on water quality in floodplain aquifers, and quantification of the fluxes of associated pollutants to rivers via the subsurface;
- the conditions and mechanisms that control the occurrence of groundwater flooding in urbanised floodplains; and
- the role groundwater and shallow geology play in controlling the duration of flooding in urbanised floodplains, through the development and application of a model system for simulating flooding that links flood inundation and groundwater flow models.

Through these focussed studies a range of generic recommendations are made for environmental managers, as well as recommendations for future work.

Chapter 1: Introduction

1 Introduction

1.1 Riverine floodplains

1.1.1 Riverine floodplain services

Riverine floodplains are areas of low-lying ground adjacent to rivers, formed mainly of unconsolidated river sediment. These floodplains are highly complex natural systems of high biodiversity and societal value, but are often severely degraded and in urgent need of protection and rehabilitation (Erős et al., 2019). Modification and degradation of floodplains is ongoing due to urbanization, navigation, increasing levels of agriculture and the development of major hydropower projects, making large riverine floodplains one of the most threatened ecosystems on Earth (Arthington et al., 2010; Sommerwerk et al., 2010).

Historically floodplains have been attractive locations for urban development for reasons such as their coincidence with sources of water and food and their proximity to transportation routes (Montz, 2000). However, in recent decades there has been a rapid increase in the rate of urbanisation of floodplains (Monk et al., 2019). As a proxy for urban floodplain population, Jongman et al. (2012) estimated the global population in 2010 exposed to a 1 in 100 year return period fluvial flood as 805 million, approximately twice the equivalent population in 1970. EEA (2018) states that 15% of Europe's population is located on floodplains; in Austria, the Netherlands, Slovakia and Slovenia this figure rises to more than 25%. Kummu et al. (2011) estimated that more than 50% of the world's population lives within three kms of freshwaters. Grizzetti et al. (2017), through an assessment of multiple pressures on European rivers, identified the area of floodplains that have been urbanised as a key predictor of ecological degradation.

As riparian zones, floodplains are defined as ecotones between terrestrial and aquatic realms (Gregory et al., 1991) extending from water bodies and including terrestrial vegetation associated with shallow groundwater (Naiman et al., 2000; Nilsson & Berggren, 2000). In lowland regions, floodplains associated with major rivers can occupy large areas; globally riverine floodplains cover more than 2 x 10^{6} km² (Ramsar & IUCN, 1999). Floodplains are dynamic systems shaped by repeated erosion and deposition of sediment, flood water inundation, and complex groundwater-surface water exchange processes (Junk et al., 1989; Thorp et al., 2006). This dynamic nature makes floodplains highly biologically productive and diverse ecosystems.

Tockner and Stanford (2002) estimated the worldwide value of the services provided by floodplains as US\$3.9 trillion annually, more than 25% of the value of all terrestrial ecosystem services although they cover only 1.4% of the land surface (Mitsch & Gosselink, 2000). In these services, Tockner and Stanford (2002) included flood regulation (37%), water supply (39%) and waste treatment (9%). However, anthropogenic development has had a devastating effect on the ecosystem services provided through floodplains and floodplain ecology, both in the developed and, increasingly, the developing world (Friberg et al., 2017). Expansion in agriculture is a primary reason for the alteration of floodplains. This includes removal of natural vegetation, mobilisation of sediment, drainage of land, and input of pollutants (Poff et al., 1997; Blann et al., 2009; Krause et al., 2011a). In Europe and North America, up to 90% of floodplains are already cultivated (Tockner & Stanford, 2002). Entwistle et al. (2019) report an increase in intensive agriculture in England from 38% of floodplain zones in 1990, to 64% in 2015. In tandem, they report that floodplain wetland areas in the form of fen, marsh, swamp and bog have been all but lost. Substantial efforts are ongoing to address the degradation of floodplain ecosystems and to improve the ecosystem services of floodplains, such as flood mitigation, by returning rivers to a more natural state (Palmer et al., 2014; Guida et al., 2016; Brown et al., 2018). Roni and Beechie (2013) estimated a global expenditure of approximately US\$ 3 billion annually on interventions including the remeandering of rivers, riparian revegetation, the removal of flood embankments and weirs to return the connectivity of river habitats, and riverine wetland creation (Gilvear et al., 2013).

In addition to habitat alteration, Tockner et al. (2010) identified the main threats to floodplains and their ecosystems as pollution (discussed in Section 1.1.4), measures for flow and flood control (Poff et al., 1997) and species invasion. Changes to natural flow regimes within river—floodplain ecosystems have come about through the regulation of rivers for the purposes of water supply, hydropower generation, flood management and the development of transport routes. Nilsson and Berggren (2000), at the time, estimated that two-thirds of the fresh water flowing to oceans globally was obstructed by approximately 40,000 large dams and 800,000 smaller ones. CIA (2002) stated that more than 600,000 kms of inland waterways have been altered for navigation worldwide. Nearly all large rivers and their floodplains in central Europe and the USA are affected by dykes, with an estimated 40,000 km of dykes in the USA alone (Johnston Associates, 1989). In central Europe, the floodplain landscapes of the Rhine, the Lower Danube and the Middle Elbe Rivers, for example, have lost more than three-quarters of their natural inundation area as a result of dykes (Leyer, 2005). Within England and Wales, records from the Environment Agency, accessed in 2014, showed that on the 68,755 km of the river network there were 17,569 locks, weirs and control gates (Figure 1).



Figure 1 River management structures in England and Wales provided by the Environment Agency (EA), March 2014. These sites appeared on lists of: flood risk management assets, extracted from the EA Asset Information Management System; and locks, held as part of the EA navigation function. Contains Environment Agency data licensed under the Open Government Licence v3.0, and Ordnance Survey data © Crown copyright and database right (2018).

Flow has a major control on physical habitat conditions in streams, which in turn is a primary determinant of biotic composition. Aquatic species have evolved life strategies in direct response to natural flow regimes and the maintenance of natural patterns of longitudinal and lateral connectivity is essential to the viability of populations of many riverine species (Bunn & Arthington, 2002). The impact of the control on river flows is a reduction in flood peaks, flooding frequency and duration, and a change in the nature of dry periods (Kingsford, 2000; McMahon & Finlayson, 2003). These flow changes reduce lateral connectivity between the floodplain and the river (Diaz-Redondo et al., 2018), disrupt sediment transport (Wohl et al., 2015) and reduce channel-forming flows (Ward & Stanford, 1995).

Inundation–duration curves used to assess the hydroecological integrity of floodplain ecosystems indicate an almost linear relationship between water level and inundated area in natural floodplains (Benke et al., 2000; Van der Nat et al., 2002). In regulated rivers, floodplain inundation tends to be short and the increase in inundation area is abrupt. Along channelled rivers, floodplains only receive surface flooding from the river during major flood events. Therefore, large-scale floodplain development and water-resource development often lead to major decreases in the active floodplain area.

According to Tockner and Stanford (2002), species invasion is the second most important cause of the overall decline in aquatic biodiversity. Less susceptibility to processes that are highly restrictive to natural species is a primary pathway to the dominance of invasive species (MacDougall & Turkington, 2005); this can relate to groundwater conditions. For example, in examining the key drivers of compositional shifts in riparian plant species, Stromberg et al. (2007) identified that the anthropogenic alteration of stream-flow regime in the Gila and Lower Colorado basins in Arizona, USA, favoured introduced species; these species, in contrast to native species, have deep roots and have a narrow germination window, which mean they cope better than the native species with the lower groundwater levels and shifts in flood timing that have been a consequence of river flow regulation. Leyer (2005) demonstrated that reduced water level fluctuation caused by the construction of dams and dykes on the River Elbe, Germany, led to substantial changes in the spatial distribution of floodplain plant species due to native-species preference for highly fluctuating, over stable, water tables. Native species need high groundwater levels at the beginning of their growing period, as they have a strong requirement for soil moisture for growth. In the floodplain study area for this thesis, increases in invasive wetland species in a highly biodiverse and protected floodplain meadow have been linked with long-term changes in groundwater levels and flood frequency (Punalekar et al., 2016).

1.1.2 Groundwater flows in floodplain aquifers

Groundwater-surface water interaction

Floodplains are complex systems where rivers erode and deposit sediments ranging in size from clays to gravels (Bridge, 2009). Significant groundwater flows occur where there are appreciable thicknesses of coarser-grained and more permeable sediments. These sediments can be in hydraulic connection with underlying permeable bedrock, in which case they may form part of a regional groundwater flow system, or they may be isolated by low permeability bedrock, in which case, in riverine floodplains, flows are driven by the interaction with surface water bodies (Woessner, 2000). The latter is the case within the study area used in this thesis.

During low precipitation periods, groundwater tends to discharge to rivers and streams. In periods of high precipitation when flows increase, higher river levels, especially in the lower river reaches, can cause the river to change from influent to effluent condition, infiltrating its banks and recharging the aquifer (Sophocleous, 2002). The interaction of river and floodplain aquifer is influenced by the hydraulic conductivity of river bed material. Naganna et al. (2017) review factors that influence river bed hydraulic conductivity. The factors are many and include: sediment particle size, substratum heterogeneity, longitudinal variations in impervious surfaces such as bedrock and sills, bed material

depth, channel geometry, variations in hydraulic radius, and roughness due to natural and anthropogenic modifications. As a result, the range of hydraulic conductivity varies over several orders of magnitude, reflecting variability that can be caused by settling, clogging of fine mineral or organic particles (colmation), and compaction due to the weight of eroded materials. Given the range of controlling factors, river bed hydraulic conductivity is highly spatially variable (Calver, 2001; Irvine et al., 2012) and can result in zones of focussed interaction between river and aquifer (Heeren et al., 2013).

The process of colmation occurs after extended periods of low flow velocity. It can also be induced by algal mats in eutrophic streams and by cohesive depositions in rivers receiving sewage effluent (Brunke, 1999). The removal of this layer, or decolmation, can result from the erosion that occurs during high flow conditions (Wong et al., 2015). Doppler et al. (2007) provide field evidence of the temporal variability of the river bed hydraulic conductivity, identifying that, depending on geometry and hydraulic characteristics of the riverbanks, it can be a function of the river stage, as at higher river levels more permeable zones of the river bed can contribute to infiltration of river water.

The alternating phases of colmation and decolmation are natural processes of sedimentation and erosion, however, the balance may be altered anthropogenically towards enhanced siltation (Petts, 1988; Doppler et al., 2007). The aggregation of fine-grained material associated with lower river velocity upstream of engineered river management structures and scouring of the river bed downstream of these structures, in combination with head gradients that increase and decrease the propensity for colmation, mean river bed hydraulic conductivity under the influence of river structures can be highly variable (Hatch et al., 2010).

As discussed in Section 1.1.1, river flows are often controlled by river management structures. Studies have shown how structures can cause groundwater levels in the associated aquifer to be raised and river reaches to switch from gaining water from the adjacent aquifer to losing water, when structures are introduced (Krause et al., 2007; Hill & Duval, 2009; Matula et al., 2014; Lee et al., 2015). There is some evidence that the influence of river management structures on groundwater levels can be linked to nutrient attenuation in the associated floodplain aquifers, but a limited number of studies have been undertaken on this topic (see section 1.2.1).

River-aquifer-floodplain interaction during high flows

This PhD has focussed primarily on high river flows and flooding on floodplains. Periods of high river flow result in raised river levels and, in some circumstances, out-of-bank flows onto the adjacent floodplain. Field studies have shown that rises in the river stageassociated with high river flow often cause a hydraulic gradient from river to aquifer resulting in temporary storage of river water in the riparian zone. For example, through a series of studies on a stretch of floodplain on the River Severn in the UK (Bates et al., 2000; Burt et al., 2002a; Claxton et al., 2003; Jung et al., 2004), observations and modelling were used to examine groundwater flows in a floodplain aquifer during high river flow events, involving both in-bank and overbank flooding. In both cases, river stage rise was shown to induce rapid responses of the water table over many tens of meters across the floodplain, which switched off hillslope inputs to the riparian zone. In overbank flood events, the rise in groundwater level that occurred beneath the floodplain was a response to the river stage rise rather than a response to flood water recharge (Claxton et al., 2003).

Vidcon (2012) reports a similar study in a floodplain in a second-order stream draining a mostly agricultural watershed in Indiana, USA. During storms, larger water table fluctuations (approximately 100 cm) occurred near the stream, compared with near the toe of the hillslope at the edge of the floodplain (10 - 25 cm). A quick rise in the water table near the stream occurred for all storms studied. Water table fluctuations, groundwater flow velocities and electrical conductivity data indicated that riparian zone water table responses to precipitation were primarily regulated by pressure wave processes. Regardless of the storm, high water tables persisted for at least 2 days after the cessation

of precipitation. Malzone et al. (2016) showed, through the analysis of a series of storm events, that the interaction between river and aquifer was dependent on antecedent conditions, with the time of year dictating the duration and magnitude of the exchange of water within the hyporheic zone.

Cloutier et al. (2014) and Buffin-Bélanger et al. (2016) in studies in a high hydraulic conductivity, unconfined, gravelly river floodplain aquifer in Eastern Canada, identified for a series of in-bank river floods, well-defined groundwater flood waves that propagated quickly through the entire floodplain (250 m). The largest flood event recorded in the studies affected local groundwater flow orientation by generating an inversion of the hydraulic gradient for 16 hours.

The diffusivity of the floodplain aquifer is a key controlling factor in the groundwater flood wave response (Pinder et al., 1969; Reynolds, 1987). The propagation of the flood wave is therefore very fast in confined aquifers with high hydraulic conductivity. García-Gil et al. (2015) used the equation for flood wave propagation derived by Pinder et al. (1969) to examine the influencing factors in groundwater flooding in the city of Zaragoza in north-eastern Spain, which is located on the floodplain of the Ebro River. These factors were: river level rise rate; absolute–relative height of the maximum stage; aquifer parameters; river–aquifer exchange rates; pre-event state of the aquifer; and distance to the aquifer boundary. Synthetic models derived by García-Gil et al. (2015) were used to assess risk to subsurface infrastructure from groundwater flooding and to improve their design.

The importance of groundwater recharge to floodplain aquifers from overbanked fluvial flood waters has been highlighted by Doble et al. (2012). Through the application of a fully coupled, surfacegroundwater flow model they identified that the infiltration volume increased with the floodplain flood level and duration, and was limited by low values of: hydraulic conductivity of the river bed and surface layer of the floodplain; aquifer transmissivity; and unsaturated aquifer storage. Land development that has occurred in recent human history, such as the stripping of natural vegetation to allow agricultural activities, has contributed to higher loads of fine sediments within rivers globally (Walling & Fang, 2003; Macklin et al., 2010). The deposition of this material downstream has produced floodplain sediments that commonly have a low hydraulic conductivity surface layer. Ramberg et al. (2006) show in a study from Okavango Basin, Botswana, that with the absence of this low hydraulic conductivity layer, as well as deep groundwater levels, very substantial groundwater recharge can occur. This was also the case for a delta region to the south west of Barcelona, Spain, where overabstraction has caused groundwater levels to fall well below the bed of the Llobregat River (Vázquez-Suñé et al., 2007). As a result of a lack of a low hydraulic conductivity surface layer, a large proportion of fluvial flood waters recharge the unconsolidated aquifer, representing 40% of the total aquifer water inputs.

Accounting for groundwater recharge from overbank flooding is required to reduce uncertainty and error in river-loss terms and groundwater sustainable-yield calculations (Wang et al., 2015). However, continental- and global-scale models of surface water–groundwater interactions rarely include an explicit process to account for overbank flood recharge. Doble et al. (2014) calculated the proportion of overbank flood recharge to be at least 4% of the total change in groundwater storage in a modelled catchment, and at least 15 % of the riparian recharge. Accounting for overbank flood recharge is an important, but often overlooked, requirement for closing water balances in both the surface water and groundwater domains.

Groundwater and flooding

Groundwater flooding is the emergence of groundwater at the ground surface away from perennial river channels but can also involve the rising of groundwater into man-made ground and subsurface assets, including the basements of building and other infrastructure such as sewers (Macdonald et al., 2008; Booth et al., 2016). The impact of groundwater flooding can be severe under conditions where the 'normal' ranges of groundwater level and groundwater flow are exceeded. This is a form of flooding on which a limited amount of research has been undertaken, as its relevance has only been widely acknowledged in the last two decades, subsequent to widespread groundwater flooding in the

UK and France in 2000/01 (Finch et al., 2004). The only national study that could be found on the scale of impact from groundwater flooding is that of McKenzie and Ward (2015), which estimated that in England between 122,000 and 290,000 properties are located in areas of high groundwater flood risk. In an assessment of future groundwater flood risk in the UK, Sayers et al. (2015) estimated increases of 71% and 90% in the significant chance of flooding for residential and non-residential properties, respectively, by the 2080s.

Groundwater flooding takes a number of forms (Macdonald et al., 2008). *Clearwater flooding* is longlasting, often regionally extensive groundwater flooding caused by the water table in an unconfined bedrock aquifer rising above the land surface as a response to extreme rainfall (Hughes et al., 2011; Naughton et al., 2017). It is this form of flooding that caused the significant damage to properties on the Chalk outcrop of southern England and northern France in recent years, often as the result of anomalous spring flows (Pinault et al., 2005; Ascott et al., 2017). It is of relevance to urbanised lowland floodplains where the alluvial floodplain sediments are in good hydraulic connection with underlying bedrock aquifers.

Flooding from *urban groundwater rebound* occurs where there has been a reduction in abstraction from large aquifers underlying major urban centres due to a decrease in industrial groundwater-reliant activities and a move away from groundwater as a source for domestic water supply (Soren, 1976; Lerner & Barrett, 1996; Jones, 2007). This allows lowered groundwater levels to recover causing the risk of flooding to subsurface infrastructure, such as tunnels and the basements of buildings, as well as changes in geotechnical and geochemical properties that can result in settlement and corrosion of deeply founded structures.

A form of groundwater flooding particularly relevant to urbanised floodplains is *permeable superficial deposit (PSD)* flooding. This flooding is associated with shallow unconsolidated sedimentary aquifers in hydraulic connection with rivers. These aquifers are susceptible as the storage capacity is often limited, direct rainfall recharge can be relatively high, and the sediments are very permeable. Few examples of the impact of this type of flooding have been reported in the international literature (see Section 1.2.1 for further detail).

In many cases, including in the study area on which this thesis is based, groundwater flooding is associated with floodplain aquifers isolated by underlying very low hydraulic conductivity bedrock. However, there are examples of locations where the lateral inflow of groundwater to floodplain aquifers from adjacent superficial and bedrock aquifers contributes to groundwater flooding, resulting in longer-lasting flood events. Gotkowitz et al. (2014) report on groundwater flooding that occurred in a floodplain terrace of the Wisconsin River, fed laterally and from beneath by a regional sandstone aquifer. This flow, in response to a period of intense rainfall and snow melt, caused flooding that lasted for a period of six months. Ó Dochartaigh et al. (2018) describe the groundwater dynamics of an upland floodplain aquifer in Scotland which although hydraulically well-connected to the river, with rapid groundwater level rise and recession over hours, is also strongly coupled with highly permeable extensive superficial deposits on the adjacent hillslopes. As a result of the lateral groundwater flow these deposits, high heads can be maintained in the floodplain aquifer for weeks, sometimes with artesian conditions, with important implications for drainage and infrastructure.

Urban groundwater flooding also occurs in cities where there is *excess importation of water* to meet urban needs for both public water supply and industrial activities (Foster, 2001). Where large volumes of water are imported and a high proportion is lost from the distribution network and from wastewater collection systems, the resulting aquifer recharge can cause long-term groundwater level rise (e.g.: urban centres of the Middle East, George (1992); Riyadh, Rushton & Al-Othman (1994); urban centres in north-eastern Ukraine, Jakovljev et al. (2002)). This situation often occurs in cities in the developing world where the water supply and wastewater collection networks are not sufficiently well constructed or maintained. This is a particular issue in large urbanised floodplains where the river drainage network is insufficient to maintain groundwater at low levels. The imbalance of aquifer recharge and discharge in urban areas, causing localised waterlogging, is a potential consequence of policies that promote *sustainable drainage systems* (SuDS; Dearden & Price, 2012). SuDS have a range of environmental aims which include: the alleviation of pluvial flooding caused by the sealing of the ground surface in urban areas; and the inability of the engineered piped drainage networks to deal with periods of intense rainfall. Pluvial flood alleviation is addressed by enhancing subsurface infiltration through a diffuse network of trenches and soakaways. However, the technique is often applied in settings that are not suitable, where the floodplains have low permeability and shallow water tables, causing localised waterlogging (Potter, 2006; Zheng et al., 2015).

The subsurface plays a key role in the growth of the built environment. It is the location for foundations, basements and subsurface infrastructure, such as sewerage networks and telecommunications. These constructions, which either create barriers to or conduits for groundwater, affect groundwater flows and, in turn, levels (Vázquez-Suñe et al., 2004; Pujades et al., 2012). Changing groundwater levels due to new construction can have implications for existing nearby property and infrastructure, potentially causing subsurface groundwater flooding (Paris et al., 2010). Other impacts associated with the rise of heads include: reduction of the bearing capacity of shallow foundations; expansion of heavily compacted fills under the foundation of structures; settlement of poorly compacted fills; increase in loads on basement walls of buildings; and increase in the need for drainage in temporary excavations (Marinos & Kavvadas, 1997). This may also cause the mobilisation of legacy pollutants located at shallow depths (Miller & Hutchins, 2017).

In fluvial flood events, much of the water that flows out of river channels onto the associated floodplain will find its way back to the river as its level recedes. However, natural levees and anthropogenic changes to topography in urban areas, may mean that shallow water is retained on the floodplain (Lewin & Ashworth, 2014), extending the period of flooding (McMillan & Brasington, 2007). In this context, Moftakhari et al. (2018) define nuisance flooding as shallow depths of flood water of 3-10 cm that do not pose significant threat to public safety or cause major property damage but do impact transport and public health. This impact may be caused by direct contact with flood waters but may also be indirect, for example by blocking the road network (Hammond et al., 2015). Moftakhari et al. (2017) showed that the cumulative exposure to frequent, relatively small flood events could cause a greater economic impact than exposure to less frequent extreme events. The recession of trapped flood waters will be slowed by low permeable surface geology and saturated ground associated with high groundwater levels. This is an aspect of flood inundation that has had little attention from the research community, however, with the potential for a greater frequency of flooding under future climate (Arnell & Gosling, 2016), nuisance flooding may become more prevalent. The economic drivers may be strong enough to require this form of flooding to be considered routinely in the modelling of flood hazards.

1.1.3 Modelling flooding in permeable floodplains

Although groundwater and surface water are often hydraulically well-connected, they are commonly considered as two separate systems and analysed independently. This is partly due to the difficulties in modelling their interactions but also because groundwater is considered to move over longer timescales than that of surface water (Liang et al., 2007). Evidence from Section 1.1.2 questions this stance in relation to flooding in permeable floodplains. Flood events in these settings can involve a range of interacting flooding mechanisms. Fluvial, groundwater and pluvial flooding, along with the performance of urban drainage systems, can all play a role in determining the nature and impact of urban flood events (Jha et al., 2012).

Complex, distributed models with detailed physics-based process representations, such as Mike-SHE, have been used to simulate variations in groundwater levels in two dimensions across floodplains, and their interactions with channels and the land surface (Bernard-Jannin et al., 2016; Clilverd et al., 2016; House et al., 2016; Thompson et al., 2017). These models have been used primarily to examine

exchanges at the interface between surface water and groundwater in the context of floodplain ecosystems. An alternative approach to the modelling of surface water-groundwater interaction is via an exchange flux that appears in both the groundwater and surface water flow equations as general sink/source terms (Kollet & Maxwell, 2006). In this approach, the exchange rate is often expressed in terms of the conductance concept, which assumes an interface connecting the two domains (Anderson et al., 2015). Recent studies have included additional processes into the conductance concept to account for the influence of microtopography on surface saturation (VanderKwaak & Loague, 2001; Panday & Huyakom, 2004). The application of the conductance concept to natural systems can be problematic where a distinct interface between the surface and subsurface is absent (Cardenas & Zlontik, 2003).

While modelling studies commonly integrate a surface and subsurface component as well as surface water-groundwater exchanges, overbank flood events are rarely included in the analysis (Bernard-Jannin et al., 2016; Teng et al., 2017). However, the importance of including overbank flood water infiltration has been illustrated in floodplain modelling studies (Doble et al., 2012). For example, both Hester et al. (2014) and Claxton et al. (2003), show its importance in determining the overall residence time of water within floodplain environments.

However, even where groundwater is recognised as relevant to flooding, the modelling codes that incorporate groundwater-surface water interactions are not generally used by the flood forecasting community (Teng et al., 2017; Jain et al., 2018), given their highly parameterised nature and relatively long run-times. This limits their use in probabilistic ensemble forecasting frameworks. Bernard-Jannin et al. (2016) undertook a comprehensive study of surface water-groundwater exchanges during overbank flood events by coupling 2D modelling of surface flow with the Saint-Venant equation and 3D modelling of unsaturated groundwater flow with Richard's equation. One finding from the modelling of this unconfined system was that significant differences in exchanges occurred in response to floods of different magnitude. However, the distributed model was computationally expensive and could only be applied at the reach scale for a short time.

Conversely, groundwater fluxes are often neglected, or poorly represented, in reduced-complexity models, using kinematic or diffusive wave approximations, which are widely used to simulate inundated floodplains (see Teng et al., 2017 for a review), such as FloodMap (Yu & Lane, 2006), JFLOW (Bradbrook, 2006), and LISFLOOD-FP (Bates & De Roo, 2000). For example, in modelling groundwater flooding caused by the water table rising to the land surface in response to rainfall recharge, Morris et al. (2018) calculated groundwater discharge rates along a chalk valley using a simple Darcian calculation. These estimated flows drove their JFLOW flood inundation model but were calibrated during the modelling process by comparing simulated floodplain flows against targeted spot flow measurements at observed points of groundwater emergence and at a point near the perennial head of the river downstream, which were taken during the flood event. Such an approach does not allow for predictive simulation.

Flood mechanisms in urbanised river floodplains also involve flows via subsurface drainage (Hammond et al., 2015). Drainage networks are means to remove flood waters but can also act as pathways for flood waters. Few modelling studies that have addressed this issue have included groundwater (Elga et al., 2015) due to the complexity of linking models that incorporate flows in rivers, aquifers and drainage networks, as well as interactions between these components. There is a growing number of studies that integrate models that simulate the interaction of drainage and groundwater during pluvial flooding or to examine urban aquifer water balance (e.g. Domingo et al., 2010; Kidmose et al., 2015; Locatelli et al., 2017). Sommer et al. (2009) describe a model system developed to obtain a holistic understanding of flood risk in the city of Dresden on the floodplain of the River Elbe. The system incorporates models of river, groundwater and sewerage systems using model-inking software (MpCCI; (http://www.mpcci.de). The aquifer and the sewerage system are linked via Darcy flow equations. The coupling between the flood and sewerage system works on a smaller time step than with groundwater. The linking software interpolates between the individual model grids. It is not clear

how river and aquifer are linked but flow here is conceptualised as being a minor proportion of the flow within the floodplain. Sommer et al. (2009) conclude that in the setting they have investigated: the inclusion of groundwater does not affect surface flood levels; groundwater infiltration is not significant in the overloading of the sewerage in comparison to storm runoff; problems can persist with the sewerage network, in part due to groundwater, if the river into which it is discharging does not recess quickly; and modelling helps to identify areas where high groundwater levels pose a risk to buildings.

1.1.4 Pollution

Floodplains are the locations of a range of receptors that are sensitive to pollution, in particular: aquatic ecosystems in rivers, lakes and ponds; wetland terrestrial ecosystems; and rivers and aquifers used for water supply. Rivers and floodplains are particularly exposed to high anthropogenic stress as they are located in the low-lying areas of the landscape in which the whole range of catchment modifications and impacts accumulate (Tockner et al., 2010). Urbanisation on floodplains also inevitably introduces contamination, particularly at the peri-urban interface. This is a zone of interaction, where urban and rural activities are juxtaposed, landscape features are subject to rapid modification, and often support large industrial developments (McGranahan et al., 2004; Douglas, 2006).

There are many potential sources of pollutants in urban environments, for example leaking sewers and other wastewater collection systems, accidental discharges from commerce and industry, and the maintenance of parks and other green spaces (Foster et al., 1999; Fetter et al., 2017). Peri-urban floodplains may also support some agricultural activity. In addition, many legacy landfills and waste dumps are located within urban and peri-urban floodplains (Laner et al., 2009; Brand et al., 2018).

Rivers themselves may be a source of pollution where flow into floodplain aquifers occurs in losing reaches and through overbank flooding during high flows (e.g. Stewart et al., 1998; Cabrera et al., 1999; Ciszewski & Grygar, 2016). Rivers can transport substantial amounts of pollutant associated with activities upstream, in particular excess nutrients used in agriculture and discharges from sewage treatment works (Mainstone & Parr, 2002; Withers & Lord, 2002; Jarvie et al., 2006; Bouraoui & Grizzetti, 2011).

Floodplain aquifers can be important as sources of drinking water (Ascott et al., 2016). These are often river bank infiltration (RBF) schemes (Hoehn, 2002) that take advantage of the improvement to water quality resulting from the attenuation that takes place in the riparian zone. RBF schemes can provide a high proportion of public water supply, for example 50% in Slovakia, 45% in Hungary and 16% in Germany (Hiscock and Grischek, 2002). Although these schemes can be highly effective at attenuating river pollution when in-bank (Ramli et al., 2017), Ascott et al. (2016) provide evidence that inundation of schemes from polluted overbanked flood waters can result in significant deterioration of public groundwater supplies for periods of weeks.

Outside of RBF schemes, groundwater sources within urban areas are less common in developed countries, given ready access to piped water systems and concerns about urban groundwater quality (Howard & Gelo, 2002). However, in developing countries, boreholes are a growing component of domestic water supply infrastructure, including community groundwater supplies for low income populations in peri-urban areas outside of the reach of public water supply networks (Foster et al., 2011). These communities are often located on marginal land on floodplains (Güneralp et al., 2015) and vulnerable to pollution associated with poor borehole construction and proximity to on-site sanitation (Back et al., 2018).

Nitrogen pollution in floodplain aquifers

Specific studies included in this thesis (Chapter 2 and 3) focus on nitrogen as a pollutant in floodplain environments. In recent decades anthropogenic inputs of nitrogen, for food production by intensive agriculture, and urbanisation, have caused increases in macronutrient fluxes, and have led to

widespread nitrogen pollution of aquatic systems (Foster et al., 1982; Burt et al., 2011; Whitehead & Crossman, 2012; Lapworth et al., 2013). This is a global issue with implications for food production and security, water quality and land management/planning (Galloway et al., 2004; Krause et al., 2008).

Characterising redox zones constitutes an important framework for understanding the behaviour of nutrients in floodplain aquifers. Organic carbon is very important for the evolution of different redox regimes in aqueous environments, and the amount and reactivity of the organic carbon in the aquifers is therefore an important parameter when evaluating the state and trends of groundwater quality (Gooddy & Hinsby, 2008). In floodplain environments there is a ready supply of organic matter from river water inundation and groundwater-surface water exchange (Lapworth et al., 2009). This provides a seasonal impetus for microbial action and the formation of transitory redox zones.

Nitrate, the predominant oxidised form of nitrogen, is readily transported in water. High concentrations of nitrate are typically associated with diffuse agricultural pollution from fertilisers (Oakes et al., 1981; Addiscott et al., 1991), although oxidation of anthropogenic ammonium sources also causes high nitrate concentrations (Gooddy et al., 2002). Anaerobic carbon-rich sediments, characteristic of floodplains, have the potential to support large populations of denitrifying bacteria. The denitrification potential generally increases with organic matter content towards the soil surface (Burt et al., 1999) with consequent potential for denitrification increasing as the water table rises. Shallow water tables also help to create anaerobic conditions, as the aerobic unsaturated zone of the sediments is small (Burt et al., 2002b; Kellogg et al., 2005).

Studies have shown a relationship between denitrification potential and factors such as soil texture, discrete flow zones, channel shapes and fluctuating water levels. Dahm et al. (1998) found that hydrological and biogeochemical dynamics were linked to the sediment characteristics of the floodplain and stream bed interface and also to the degree of channel constraint, the availability of specific chemical forms of electron donors and electron acceptors and temporal changes in discharge. Pinay et al. (2000) found a significant relationship between denitrification rates in the floodplain soils and their texture; highest rates were measured in fine textured soils with high silt and clay content. Pfeiffer et al. (2006) found evidence of nitrate attenuation at depth which depended on topographic features and temporal variability. This flow system provided sources of dissolved organic carbon to deeper groundwater flow paths, leading to the consumption of dissolved oxygen and generating redox conditions suitable for denitrification and subsequently iron reduction. McCarty et al. (2007) also showed that the denitrification potential of a riparian wetland was both stratified and was limited by the presence of discrete seepages or upwelling zones rather than a uniform distribution. Kolbe et al. (2019) provide evidence of pronounced biogeochemical reactivity with depth at a series of locations in six aquifer settings across the USA and suggest that previous estimates of denitrification had underestimated the capacity of deep aquifers to remove nitrate, while overestimating nitrate removal in shallow flow paths. They argue that the increased oxygen and nitrate reduction identified related to relatively little organic carbon in agricultural soils and the excess nitrate input that has depleted solid phase electron donors near the surface. Burt et al. (2002b) showed how fluctuating water levels can control the degree of denitrification. Stating that denitrification potential generally increases towards the soil surface, they propose water table elevation can control the degree to which nitrate reduction is optimised.

Nitrogen in the form of ammonium is present naturally in groundwater as a result of anaerobic degradation of organic matter but also occurs in groundwater impacted by anthropogenic sources, for example as a result of domestic and agricultural waste water disposal practices as well as landfill leachates (Gooddy et al., 1998; Lawrence et al., 2000; Christensen et al., 2001; Heaton et al., 2005). A large number of landfill sites are located on floodplains, as is highlighted for England and Wales by combining data obtained from the Environment Agency on licensed landfill sites, with floodplain extent; 3797 of the 21030 sites, 18%, are located on floodplains (Figure 2). Siting of landfills on floodplains is not current practice and therefore it is assumed these sites are historical and, as the examples in the case study area used within this thesis (Chapter 3) would suggest, without adequate

arrangements for the containment of leachate. Ammonium transport in the subsurface may be retarded by sorption, as it exhibits a high affinity for ion exchange and may be incorporated into clay mineral lattices (Ceazan et al., 1989; Buss et al., 2004), and ammonium can be attenuated through microbially induced transformations (DeSimone & Howes, 1998).

McClain et al. (2003) showed that rates and reactions of biogeochemical processes vary in space and time and these variations are often enhanced at terrestrial-aquatic interfaces. They defined biogeochemical 'hot spots' as patches that show disproportionately high reaction rates relative to the surrounding matrix, whereas 'hot moments' were short periods of time that exhibit disproportionately high reaction rates relative to longer intervening time periods. Hot spots occur where hydrological flow paths converge with substrates or other flow paths containing complementary or missing reactants. Hot moments occur when episodic hydrological flow paths reactivate and/or mobilize accumulated reactants.



Figure 2 Historical landfill sites in England and Wales, licensed by the Environment Agency 2014, including those located on riverine floodplains. Floodplains as defined by the British Geological Survey Geological Indicators of Flooding dataset, Class 1 and 2 Fluvial Zones (Booth & Linley, 2010). Contains Environment Agency data licensed under the Open Government Licence v3.0, and Ordnance Survey data © Crown copyright and database right (2018).

Krause et al. (2017) highlight ecohydrological interfaces, the dynamic transition zones that often develop within ecotones or boundaries between adjacent ecosystems, as hot spots of ecological, biogeochemical, and hydrological processes. The hyporheic zone, the interface between river and groundwater (Burt et al., 2013), is a hot spot for nutrient processing (McClain et al., 2003; Hester & Gooseff, 2010; Krause et al., 2011b; Antiguedad et al., 2017). Exchanges of water, nutrients, and organic matter here occur in response to variations in discharge and bed topography and porosity (Boulton et al., 1998). Upwelling groundwater can supply stream organisms with nutrients while downwelling stream water can provide dissolved oxygen and organic matter to microbes and invertebrates. The hyporheic zone hosts a wide range of hydrologically controlled processes that are potentially coupled in complex ways. Understanding these processes and the connections between them is critical since these processes are not only important locally but integrate to impact increasingly larger-scale biogeochemical functioning of the river corridor up to the river network scale (Cardenas, 2015; Abbott et al., 2016; Pinay et al., 2017).

1.1.5 Environmental management drivers relevant to floodplain environments

The UK has a wide portfolio of environmental regulations, most of which reflect the requirements of European Union Directives, and many of which impinge on activities in urban and peri-urban floodplains (de Sosa et al., 2018). Although difficult to quantify, a link is identified between the amount of natural floodplains and achieving the key objectives of European policies (EEA, 2018). Floodplain management or protection is encouraged but only indirectly required under European environmental policies, but floodplain health is important for achieving multiple European policy objectives. The following policies are particularly relevant to floodplains. The relevance of research undertaken within the PhD to the environmental management requirements associated with these policies is highlighted.

EU Water Framework Directive (2000/60/EC)

The Directive establishes a framework for the protection of inland surface waters (rivers and lakes), transitional waters (estuaries), coastal waters and groundwater (Chave, 2001). It sets out to ensure that all aquatic ecosystems and, with regard to their water needs, terrestrial ecosystems and wetlands meet 'good status' by 2027. Achieving the goals of the Directive is facilitated through a series of River Basin Management Plans (RBMP) in which those water bodies not at good status have been identified, the causes investigated and, where not disproportionately costly, measures put in place to address these. Floodplain aguifers, at least in the UK, where extensive and sufficiently permeable, can form groundwater bodies on their own, for example, the Upper Thames Gravels in the Thames River Basin. Floodplains can also form a groundwater body with the underlying bedrock aquifer, if together they are sufficiently large and in good hydraulic connection, for example the River Colne Valley Groundwater Body in the Thames River Basin, which combines Lower Thames Gravels with the Chalk aquifer (Environment Agency, 2016). Where floodplains do not form groundwater bodies on their own, activities related to the floodplain deposits are likely to have a bearing on the status of associated river bodies (Dahl et al, 2007). In the context of floodplains, the Water Framework Directive provides protection for component water courses and groundwaters but also the terrestrial ecosystems that they support. The Groundwater Directive (2006/118/EC) complements the Water Framework Directive by establishing a regime which sets groundwater quality standards and introduces measures to prevent or limit inputs of pollutants into groundwater.

Pieces of legislation designed to protect groundwater against pollution and deterioration are part of a larger regulatory framework that can be traced back to the 1990s. The *Nitrates Directive* (91/676/EEC) aims to reduce and prevent water pollution caused by nitrates from agricultural sources. The *Plant Protection Products Directive* (91/414/EEC) and the *Biocides Directive* (98/8/EC) ensures that commercial plant protection and biocidal products, such as pesticides, herbicides, or fungicides, have no harmful effect on human health or on groundwater. These three Directives are relevant in peri-urban floodplains where agricultural activities can take place, and in urban and peri-urban floodplains

if pollutants occur as a result of interaction with rivers that are sourced from heavily agricultural catchments (Pärn et al., 2012).

The Urban Wastewater Treatment Directive (91/271/EEC) aims to protect the environment from the adverse effects of discharges of urban wastewater and wastewaters from certain industrial sectors. The Integrated Pollution Prevention and Control (IPPC) Directive (96/61/EC) lays down measures designed to prevent or reduce air, water or soil pollution. The Directive applies to a significant number of mainly industrial activities with a high pollution potential. The Landfill Directive (99/31/EC) seeks to prevent or reduce the negative effects of landfill waste on the environment, including groundwater. However, this does not address the legacy contamination that is ubiquitous in the urban and peri-urban floodplains across Europe. These sites are addressed in the UK through legislation such as the Contaminated Land (England) Regulations 2006.

To help achieve good status of floodplain groundwater bodies and to contribute to the improvement of associated water bodies, contaminants from current and historic land use may need to be removed through improved land use practices or remediation, although in some situations natural attenuation of pollutants may be sufficient to reduce concentrations to acceptable levels (Ranalli & Macalady, 2010; Weber et al., 2011; Ciszewski & Grygar, 2016). In Chapters 2 and 3 of the thesis two aspects of water quality are addressed that are relevant to meeting the requirements of environmental regulations. In Chapter 2, an investigation is undertaken of the flux of nitrate into the floodplain aquifer in the case study area via the main river bed, and the floodplain during flood inundation. The natural attenuation of nitrate in the floodplain and in the river bed sediments is estimated. In Chapter 3, the leachate from an historical landfill on the floodplain is characterised, the degree of attenuation of ammonium in the leachate in the floodplain sediments examined, and the potential input to the case study river assessed.

EU Habitats and Birds Directives (1992/43/EEC & 2009/147/EC)

These Directives address nature conservation, protecting the highest value sites (Special Areas of Conservation) and species. In the UK, the EU legislation is enacted through the Wildlife and Countryside Act (1981) which also incorporates lower-protection Sites of Special Scientific Interest. This and other legislation underpin Biodiversity 2020, the strategy for England's wildlife and ecosystem services. Floodplain meadows, an important type of groundwater-dependent terrestrial ecosystem located within river floodplains, are a valuable resource, not only supporting some of the most diverse vegetation in the UK (up to 40 species per square metre) and associated fauna, but also performing key ecosystem services, such as flood storage and sediment retention (Wheeler et al., 2004). However, the UK now has less than 1500 ha of this unique habitat remaining (Jackson & Mcleod, 2002), most of which is protected under the EU Habitats Directive. Species composition is known to be tightly correlated to the hydrological regime (Silvertown et al., 2015), and related temperature and nutrient regime, both of which, in floodplain environments, are highly dependent on shallow groundwaters. In some cases it has been the long-term manipulation of the hydrological regime through water management systems that has established the ecosystems; changes to these hydrological regimes through lack of maintenance or through measures to address other aspects of the water environment, such as flooding, may have serious deleterious impacts on these vulnerable plant communities. Environmental regulation sets out to protect these ecosystems and provide a framework within which the competing societal needs can be optimised.

EU Floods Directive (2007/60/EC)

Environmental management regulation also aims to protect property and infrastructure. The Floods and Water Management Act (2010) in England, and equivalents in other regions of the UK, were a response to the increasing economic impact of flooding on urban areas within floodplains (Nones, 2015). The Act and associated regulations flow from the EU Floods Directive (2007/60/EC), established in 2007 (Klijn et al., 2008). Traditionally flood alleviation has focussed on flood defence to reduce overbank flooding from rivers, however, since major flooding in the late 1990s and early 2000s the whole-catchment approach has received more attention across Europe (Lane, 2017). Initiatives

related to this approach, such as Making Space for Water (Defra, 2005) and Working with Natural Processes (Barlow et al., 2014) in the UK, Room for the River in the Netherlands (Warner & van Buuren, 2011), and pan-European Nature Based Solutions (EC, 2015) have shaped flood policy.

Groundwater and pluvial flooding have received more attention in the UK since the Pitt Review (Pitt, 2008) and subsequently from their inclusion in the EU Floods Directive. The Directive includes provisions for groundwater and pluvial flooding (Cobby et al., 2009): assessing risk; producing 'flood hazard maps'; and introducing measures to address any significant risk. In England, the responsibility for developing measures which address significant risk associated with the main river network falls with the Environment Agency (EA). For minor water courses, pluvial flooding and groundwater flooding, the responsibility resides with the Lead Local Flood Authority (LLFAs), with guidance provided by the EA. This has proved a challenge, as often the LLFAs do not have the capacity in terms of skillset or resources. Developing measures to address all forms of flooding, which are often interlinked, whether this is through improvements in flood defences, flood conveyance or upstream land management, is challenging but a basic understanding of the mechanisms and the risk are essential. Improving the understanding of groundwater flooding mechanisms is the focus of Chapter 4 of the thesis. A grey area in terms of responsibility is where flooding results from water from a range of sources, for example from the main river in combination with groundwater. Flood processes relevant to this situation are the topic of Chapter 5.

1.2 Research gaps addressed, and thesis aims and structure

1.2.1 Research gaps addressed

This PhD addresses research gaps related to the understanding of the hydrological regime of urbanised floodplains, in particular groundwater hydrology. It uses the case study of the lowland floodplain of the River Thames in the vicinity of the city of Oxford in the southern UK (see Section 1.3 for further detail).

Impact of river management structures on water and nitrate exchange in urbanised floodplain aquifers

A large number of studies have investigated nutrient pollution within alluvial aquifers associated with river floodplains (e.g. Haycock and Pinay, 1993; Correll et al., 1997; Clement et al., 2003; Forshay and Stanley, 2005; Krause et al., 2008). These investigations relate primarily to the quality of water in associated rivers, and measures that can be undertaken to bring nutrient concentrations below levels that are detrimental to the ecological status of the aquatic environment. Within this body of research, few studies have examined the influence of river management structures on processes that relate to nutrient cycling and, where undertaken, have addressed relatively simple hydrological settings.

In an assessment of the influence of dam operations on groundwater/surface water interaction, Hucks Sawyer et al. (2009) monitored groundwater levels, temperature, and specific conductivity along a transect perpendicular to the Colorado River (Texas, USA), downstream of a major dam. They report that the stage fluctuations, associated with the dam operation, that force river water into and out of the banks, fundamentally change the hydrological, thermal, and geochemical dynamics of riparian aquifers and their hyporheic zones. Matula et al. (2014) modelled the potential impact of the proposed construction of a weir on the River Labe in the Czech Republic to aid river transport. Their simulations indicated that as a result of the raised river level, the groundwater level in the adjacent aquifer would rise significantly and the increased flow of water, contaminated by industrial effluent, from river to aquifer, could potentially result in the pollution of 16 weirs as part of a 'rivers restoration' project in Korea that aimed to: secure water resources; introduce comprehensive flood control measures; improve water quality; and restore river ecosystems. The study showed immediate increase in the groundwater levels associated with the weir installation but at the time of reporting had not seen any change in groundwater quality.

Specifically in relation to NO_3^- , Cisowska and Hutchins (2016) examined the effect of weirs on river NO_3^- concentrations by measuring river flow, water temperature and water quality data both before and after a weir was removed from the River Nidd in Yorkshire, UK. Modelling indicated that the removal of the weir reduced the annual fraction of the upstream NO_3^- load being retained along the river reach by ~2% over the two years monitored, however, the study incorporated only processes in the river bed sediment. Hill and Duval (2009) examined the hydrology and nitrogen biogeochemistry of a riparian zone before and after the construction of beaver dams along an agricultural stream in southern Ontario, Canada. The beaver dams increased surface flooding and raised the riparian water table by up to 1·0 m. Increased hydraulic gradients inland from the stream limited the entry of oxic nitrate-rich subsurface water from adjacent cropland. Permeable riparian sediments remained saturated during summer and autumn, whereas before dam construction a large area of the riparian zone was unsaturated in these seasons each year. In turn, beaver dam construction produced significant changes in riparian groundwater chemistry, including NO_3 -N concentrations in autumn and spring, which were lower in the post-dam (0·03–0·07 mg L⁻¹) versus the pre-dam period (0·1–0·3 mg L⁻¹).

The studies described here show the impact of river structures on groundwater levels in adjacent aquifers and the potential implications for groundwater quality. With efforts to improve the ecological status of rivers there is growing interest in understanding the implications of the removal of dams and weirs on ecological status of rivers. Tullos et al. (2016), in a synthesis of common management concerns associated with dam removal, highlight the implications on groundwater supplies of removing structures, that cause associated groundwater levels and storage to decline. However, generally the focus of research in this context is on rivers, and not associated aquifers. The need to take a more holistic view of the environmental implications has been identified in reviews of research studies undertaken on this topic (Bellmore et al., 2017a, b; Ding et al., 2019).

In Chapter 2 of this thesis, a study is presented which aims to contribute to the research on the topic of the impact of river management structures on groundwater flows, and associated nutrient dynamics. It uses the case study floodplain where a high density of long-standing river management structures are located. It was not possible to assess conditions with and without structures, however, the research provides an overall understanding of the hydrology of the river network and interaction with groundwater within the study area, and uses this to quantify water and nutrient exchange, specifically focussing on NO_{3}^{-} , the predominant oxidised form of nitrogen. It uses a combination of water level observations, water chemistry sampling and a simple numerical model.

Influence of waste dumps on floodplain groundwaters and fluxes of pollutants to associated rivers As described in Section 1.1.1, globally, floodplains are increasingly being encroached and developed due to anthropogenic pressures, including urbanisation and intensive agriculture (Tockner and Stanford 2002; Pinter 2005; Werritty 2006). In many parts of world there is a historical legacy of change in land use and associated historical pollution loading to floodplain groundwaters and surface waters (Burt et al., 2011; Stuart et al., 2011a). Urban and peri-urban floodplains are therefore complex, in terms of spatial heterogeneity in land use and topography, and temporal variability in recharge and redox processes (Burt et al., 2002; Burt and Pinay 2005; Macdonald et al. 2012a; MacDonald et al., 2014).

Due to fluctuating redox conditions, floodplains are considered hot-spots for nutrient attenuation (Devito et al., 1999; McClain et al., 2003; Harms and Grimm 2008). Groundwater levels are typically shallow and responsive to recharge events, and soil moisture conditions are also highly variable; together these have important implication for attenuation of oxidised N (Burt et al., 1999). Periods of inundation may stimulate denitrification due to the mobilisation of C pools at shallow depths; during periods of low groundwater level, denitrification may be restricted due to reduced C pools, low soil moisture and more oxidising conditions.

The important ecosystem services provided by floodplains and their general proximity to potentially damaging agricultural and urban nutrient sources means there is an imperative to understand associated nutrient processes, fluxes and attenuation mechanisms. Due to the complex hydrogeology of floodplain settings, estimating the residence times of groundwaters and legacy nitrogen pollutants is challenging. There is therefore uncertainty as to the timescale for changes in land management practices reducing baseflow nutrient fluxes to surface waters (Wang et al., 2011; Stuart et al., 2011b; Jackson et al., 2008).

Despite the environmental importance of NO_3^- and NH_4^+ , there are few studies documenting their transport and reaction processes in aquifers (Heaton et al., 2005). Isotopic fractionation studies can provide an excellent tool for understanding N transport and speciation (Wassenaar, 1995; Böhlke et al., 2006). Isotopic fractionations have been reported for NH4⁺ sorption to clays (Karamanos and Rennie, 1978), with the remaining NH_4^+ in solution relatively depleted in ¹⁵N. By contrast, nitrification results in a substantial increase in ¹⁵N for the remaining NH₄⁺ (Delwich and Steyn, 1970). Stable isotope ratios in NO₃⁻ have often been used to distinguish various sources of NO₃⁻ in groundwater, such as synthetic fertilisers and animal wastes (Gormley and Spalding, 1979; Flipse and Bonner, 1985). Denitrification causes an isotopic enrichment in the remaining NO₃⁻ (Mariotti et al., 1988). A disadvantage of the single isotope approach to denitrification studies is that processes such as ammonia volatilisation can also lead to enrichment of δ^{15} N in the residual NH₄⁺ source material and in the NO₃⁻ produced during nitrification. However, by combining ¹⁵N with an analysis of the δ^{18} O of the NO_3 , a more reliable indicator of the denitrification process is achieved (Böttcher et al., 1990; Wassenaar et al., 1995; Kendall., 1998; Fukada et al., 2004). Through evaluating variations in concentration and isotopic composition, the opposing isotope fractionation effects make it possible to distinguish between sorption, nitrification and denitrification as major processes affecting N distribution in a field setting.

In Chapter 3, the temporal and spatial variations in N species is examined within the case study floodplain using geochemical indicators together with $\delta^{15}N$ of NH₄⁺ and $\delta^{15}N$ and $\delta^{18}O$ of NO₃⁻. The study focusses on an area of the floodplain down-gradient of a large waste dump. The peri-urban hydrological setting, legacy of pollution, and dependent terrestrial ecosystems within the wider floodplain mean the study is relevant across many industrialised and rapidly industrialising regions. The objectives are to evaluate the origin of N in groundwater and ascertain whether sorption, nitrification or denitrification are contributing to the attenuation of the nutrient load. In turn, through the understanding of the water movement and redox status across the floodplain, an assessment is made of the cumulative influence of nutrient fluxes from waste dumps on the adjoining river.

Conditions and mechanisms that control groundwater flooding in urbanised floodplains

As stated in Section 1.1.2, a form of groundwater flooding particularly relevant to urbanised floodplains is permeable superficial deposit flooding (PSD), which is associated with shallow unconsolidated sedimentary aquifers in hydraulic connection with rivers. Section 1.1.2 also details a number of studies that investigate river-aquifer-floodplain interaction during high flows (Bates et al., 2000; Burt et al., 2002a; Claxton et al.; Jung et al., 2004; Vidcon, 2012; Cloutier et al., 2014; Buffin-Bélanger et al., 2016), and explains the significance of floodplain aquifer diffusivity as a key controlling factor in groundwater flood wave propagation (Pinder et al., 1969; Reynolds, 1987). However, these studies focus on non-urbanised floodplain settings.

A number of examples of the impact on urbanised areas of PSD flooding have been reported. Kreibich and Thieken (2008) describe damage that occurred to buildings in the floodplain of the Rivers Elbe and Danube in Germany due to water pressure on basements, resulting from abnormally high groundwater levels associated with persistent high river levels. In the modelling studies referenced in Section 1.1.3, carried out in Dresden by Sommer et al. (2009) and Karpf et al. (2011), water flows in the urbanised floodplain were examined, identifying groundwater flow as one mechanisms resulting

in flood impact, although surface water flooding were shown to dominate. García-Gil et al. (2015) references other studies from Germany in which PSD flooding accounted for ~20% of the costs of flood damage.

In a study of basement flooding along the Elbow River in Calgary, Canada, Abboud et al. (2018) identified groundwater ingress as the initial cause in 88% of the houses surveyed, while the remainder were flooded exclusively by groundwater. Of the 19 surveyed homes located outside of the 100-year overland flood zone, 47% were flooded by groundwater, indicating that groundwater flooding reached beyond overland water-flooded areas. Modelling linked groundwater flooding to the raised river stage in this PSD environment. Flood resilience strategies were proposed including groundwater level monitoring for early flood warning and design criteria for new homes.

These were the only examples within the international peer-reviewed literature that could be found that clearly address PSD groundwater flooding. However, given that urbanised floodplains are common, it is likely that PSD groundwater flooding is the cause of flood impact in many locations, with a substantial socio-economic cost. Without an understanding of the flood process, measures to reduce flood risk cannot be properly designed or instigated. Further studies are therefore required to highlight the mechanism to raise awareness, and also to examine how raised groundwater levels interact with property and infrastructure on urbanised floodplains.

The aim of the research presented in Chapter 4 of this thesis is to improve the understanding of groundwater flooding in urbanised permeable floodplains, to better understand the controls and mechanisms, to characterise the locations where this form of flooding is likely to occur and how it relates to other forms of flooding. A conceptual model is developed using a combination of floodplain characterisation, water level observations and local knowledge of flood impact, and the relevance to flood mitigation measures is discussed.

Role of groundwater and shallow geology in the persistence of flooding in urbanised floodplains

The author's personal experience of flooding within the case study city, underpinned by the data available from the comprehensive water level monitoring network there (see Section 1.3), has highlighted that flood waters can linger on the urban floodplain for many days after the peak in river levels. The greatest apparent impacts of the extended period of flooding are disruption to the transport network, a compromised sewerage network due to groundwater exfiltration and, in a few cases, persistent inundation of property. As was discussed in Section 1.1.2, natural levees and anthropogenic changes to topography in urban areas can cause a shallow depth of flood water to be retained on the floodplain after fluvial flood events (McMillan and Brasington, 2007; Lewin & Ashworth, 2014). This shallow flooding comes within the definition of 'nuisance flooding' (Moftakhari et al., 2018), not posing a significant threat to public safety or causing major property damage, but impacting on transport (Hammond et al., 2015) and public health.

In urban areas, engineered drainage may help remove flood waters (Hibbs & Sharp, 2012). Where drainage is absent, it is expected that low permeability surface geology and saturated ground associated with shallow groundwater levels will slow the recession of water trapped on the floodplain. Few surface water-groundwater interaction studies have examined the dynamics between inundated floodplains and shallow groundwater. As a consequence, the physics of the hydrological processes in areas with shallow groundwater is not well understood, and typically it is not included in river or groundwater models (Doble et al., 2012). As highlighted in Section 1.1.2, using a model of a conceptualised floodplain-aquifer system, Doble et al. (2012) showed that the surface layer of the floodplain is a key factor in controlling the period of flood inundation. They also showed that irregularities in the floodplain elevation have a large impact on the infiltration volume due to local ponding, but given the structure of the model, did not represent flow across the floodplain in 2D. However, they acknowledged that further work was required to investigate the spatio-temporal

variability of vertical floodplain-aquifer fluxes in more complex settings, for example, with irregular floodplain topography in two dimensions and adjacent to non-linear channels.

As already discussed, complex, distributed models with detailed physics-based process representations that have been used within floodplain environments to simulate the interaction between groundwater and surface waters (Bernard-Jannin et al., 2016; Clilverd et al., 2016; House et al., 2016; Thompson et al., 2017) are not generally used by the flood forecasting community (Teng et al., 2017; Jain et al., 2018). Their highly parameterised nature and relatively long run-times, limit their use in probabilistic ensemble forecasting frameworks. Conversely, potentially relevant vertical subsurface fluxes are often neglected, or poorly represented, in reduced-complexity models, that are widely used to simulate inundated floodplains (see Teng et al., 2017).

Recognising the need for flood inundation models that incorporate groundwater and infiltration processes, Chapter 5 reports on the development of a model system that integrates the popular cellular, or 'storage cell', flood inundation model, LISFLOOD-FP, with a finite-difference groundwater model, allowing vertical exchange of water through a surface layer. The coupled model is used to estimate vertical fluxes across the case study floodplain and to test the influence that low permeability surface layers and shallow groundwater have on flood duration, and as a result explore an aspect of flood hazards with potential significant socio-economic impact that has until now been largely ignored.

1.2.2 Thesis aims and structure

To summarise Section 1.2.1, the aims of the research presented in this thesis are to:

- 1) improve the understanding of the impact of river management structures on water and nitrate exchange in urbanised floodplain aquifers;
- 2) better understand the influence of waste dumps on water quality in floodplain aquifers and to quantify the fluxes of associated pollutants to rivers via the subsurface;
- 3) determine conditions and mechanisms that control the occurrence of groundwater flooding in urbanised floodplains;
- 4) develop a flood model system that incorporates groundwater processes, to be used to understand the role groundwater and shallow geology play in controlling the persistence of flooding in urbanised floodplains.

The structure of the thesis is as follows:

Chapter 1

The context for the PhD research has already been set out in Sections 1.1 and 1.2.1 of this thesis. In the remainder of this chapter, the study area is described. It is recognised that in Chapters 2 to 5, aspects of the study area are described in the published and prepared papers and that as a result there is unavoidable repetition within the thesis.

Chapters 2 to 4

Chapters 2 to 4 are formed by three papers that have already been published in peer-reviewed journals. Each paper addresses individually, in the same order, the first three aims set out above.

Chapter 5

This chapter addresses the fourth aim above. It is written in the format of a journal paper to a standard that means it can be submitted to an international peer-reviewed journal in its current form.

Chapter 6

In this chapter the insights from the papers in Chapters 2 to 5 are brought together and the implications for environmental management in urban and peri-urban floodplains are discussed, making reference to issues raised in Sections 1.1 and 1.2.1. Remaining knowledge gaps are highlighted

and recommendations are made for further research, building on the work presented in this thesis. Recommendations are also made as to how the results in this thesis could be translated into policy and practice for environmental managers.

Note that the PhD was written over the period of a number of years and also that Chapters 2 to 4 were not published in the order they are presented. Relevant developments in the conceptual understanding that happened subsequent to the writing of individual chapters are discussed in Chapter 6.

Individual reference lists are provided in Chapters 2 to 5. These references, as well as those used in Chapter 1, are pulled together into a complete reference list at the end of the thesis.

1.3 Study area

The city of Oxford is located in the south of the United Kingdom, in the upper reaches of the River Thames catchment (Figure 3). The main River Thames flows to the west of the historic centre of the city, which is located on high ground above the current floodplain (Figure 4). The river forms a series of channels as it passes Oxford. The River Cherwell joins the River Thames within the Oxford valley, downstream of the city centre. The mean annual flow of the River Thames upstream of Oxford is 18.48 m³/s (Marsh & Hannaford, 2008). The baseflow index for the river at this located in the headwaters, and the extensive floodplain gravel aquifers (Figure 3). The Thames in Oxford has a large number of river management structures associated with it, for historical navigation and hydropower purposes. Six locks and associated bypass flow channels are located on the main river. The study area for the PhD was defined as the River Thames floodplain in the Oxford valley between the upstream lock at Eynsham and downstream lock at Sandford (Figure 4).

The geology of the River Thames floodplain in Oxford is characterised by alluvial sediments (Figure 2 in Chapter 2), which are underlain by Oxford Clay bedrock throughout most of the study area (Newell, 2008). This is typical of much of the River Thames Catchment, where the extensive areas of alluvial deposits (alluvium and river terrace deposits) are associated with the wide, low-gradient bedrock mudstones, rather than the more steeply incised carbonate and sandstone bedrock (Figure 3; Bricker & Bloomfield, 2014).

The alluvium that underlies the floodplain typically comprises of silty clay. This overlies glacio-fluvial sands and gravels deposited by Late Devensian rivers (Newell, 2008). High ground surrounding the floodplain is formed of higher terrace gravel deposited during the latest Anglian Glaciation, and thin alternating layers of Late Jurassic limestones, sandstones and mudstones. The floodplain has an area of 20.4 km² and ranges in width from 410 to 2170 m. The floodplain narrows substantially downstream as it leaves the Oxford valley. The floodplain gravels are typically 2 to 4 m thick within the Oxford valley but can be up to 8 m thick in places. The alluvium forms a relatively uniform blanket around 1 m thick across the floodplain although in some areas it thins to a few decimetres and exceptionally can be up to 4 m thick (Newell, 2008).

Nine Sites of Special Scientific Interest are located on the floodplain in the Oxford valley (Figure 4), covering a total area of including 3.62 km². Four of these sites, including Port Meadow, a focus for research within this PhD, form the Oxford Meadows Special Area of Conservation. This is designated on the basis of the lowland hay meadows within and, in the case of Port Meadow, as it is one of only two known sites in the UK where creeping marshwort *Apium repens* is found (Gowing & Youngs, 2005). Port Meadow is also designated as a Scheduled Monument by Historic England due to evidence of Bronze Age and Iron Age activity (Lambrick, 1992). Its primary use is for the grazing of cattle and horses.

Fourteen historic waste dumps, licensed by the Environment Agency, are located on the Oxford floodplain, with a total area of 1.05 km² (Figure 4). At these sites waste has been deposited directly on to the floodplain, so does not fill excavations, and there is no lining beneath. They contain primarily

domestic but also some industrial waste. The largest of these is Burgess Field (White & Macdonald, 2015), with an area of 0.34 km², and a maximum height of 4 m above the natural floodplain. This waste dump is located up-gradient of Port Meadow and the River Thames. It was the destination for domestic waste from the city from 1937 to 1980. There is a legacy of anthropogenic contamination in this part of the floodplain, with documented waste sites dating back to the late 1800s. There are a number of gravel pits within the floodplain, primarily within the northern areas (Figure 4), but there is currently no operating gravel excavation, with the last activity ending in the early 2000s.



Figure 3 Bedrock geology within the River Thames catchment, categorised as carbonate, sandstone and mudstone, and overlying superficial alluvial deposits, coloured according to bedrock category. Contains Ordnance Survey data © Crown copyright and database right (2019).

The population of Oxford is approximately 171,000 with around 3400 domestic and commercial properties located on the River Thames floodplain. Approximately 14% of the floodplain is urbanised (Figure 4). The floodplain is also the location of key transport routes. The extensive river channel network limits the number of roads into the city centre. Flooding is a significant problem for the properties and transport routes on the floodplain, including main railway lines to London and Birmingham. The transport routes and urban development restrict the movement down-catchment of overbanked river waters during flood events, and reduce flood storage, raising fluvial flood levels (Figure 3 in Chapter 4). The 1 in 100 year flood envelope, as defined by the Environment Agency (Figure 4), covers approximately 85% of the floodplain. In the past two decades, flood frequency has increased with major events, in 1998, 2000, 2003, 2007, 2012 and 2014. The July 2007 flood was the largest of these events; identified as a 1 in 20 year flood, it inundated over 200 properties (Environment Agency, 2009). A significant number of properties affected by flooding this century have been as a result of rising groundwater, which was either the sole cause of flooding or the initial cause prior to inundation from fluvial waters. Flood waters lingering on the floodplain after the main river recession, extending the duration of flooding is also a major issue, especially for vehicular access to the city centre.

The Seacourt Stream, which becomes the Hinksey Stream in the south of the city (Figure 4), has few water management structures along its length. During flood events, this is the water course that sees the greatest rise in water levels and the most extensive flooding. Since the mid-2000s investigations have taken place in Oxford, led by the Environment Agency, to assess if a major scheme to address flooding is economic. The Oxford Flood Risk Management Study (Ball et al., 2009) concluded that the economic justification for a scheme could not be made within the rules set out by Government, however, following the 2014 floods the issue was revisited and as a result the Oxford Flood Alleviation Scheme was designed and funding for its construction found (Environment Agency, 2018). The scheme is due to begin construction in 2019, with the aim to improve conveyance of high flows through the Oxford valley, primarily along the course of the Seacourt Stream, and to better contain overbanked waters where flooding does occur.



Figure 4 Features of the floodplain of the River Thames in the vicinity of Oxford, UK. Areas in white are higher ground above the 1 in 100 year flood extent. Contains Ordnance Survey data © Crown copyright and database right (2019)

Studies within the Oxford area relating to the status of floodplain ecosystems and flood risk have resulted in an extensive groundwater and surface water level network being developed (Figure 2 in Chapter 2). Groundwater levels in the aquifer are very shallow, typically fluctuating within 2 m of ground level. At some sites monitoring has been undertaken since the early 1980s (Institute of Hydrology, 1987). Manual monitoring of water levels was undertaken on a monthly basis between 2005 and 2009; the total number of monitoring points at the end of 2009 was 235. Digital water level recorders have been installed at some point within 110 of these monitoring points. At a number of sites there are combinations of two or more of the following: piezometers completed in the gravel aquifer; piezometers completed in the overlying alluvium; surface water stilling wells; and flood water level recorders. Data were made available from recorders installed in the upstream and downstream waters at four of the Environment Agency locks in the study area. During the July 2007 flood, a further six temporary manual flood water monitoring points were established, which provided additional data in urban areas for the rise and recession of flood waters.

References are made to a selection of the water level monitoring sites in Chapters 2 to 5. Each chapter has taken a different approach to naming the sites, as the papers within will normally be read in isolation. In Table 1 the monitoring sites are identified that are used in more than one chapter (n.b. none of the sites referenced in Chapter 4, apart from the locks, are included in the other chapters).

SITES	
Chapter 3	Chapter 5
	S1
	P2
PTM26D	
PTM29D	
	S2
PTM11	P3
PTM21	P1
	SITES Chapter 3 PTM26D PTM29D PTM11 PTM11

Table 1	Naming of monitoring sites used in more than one of Chapters 2, 3 and 5. Each row
	relates to the same site.

As well as water levels, River Thames flows are routinely measured within the study area by the Environment Agency. Rainfall is measured on a 15-minute interval by the Environment Agency at Eynsham and Osney Locks. There is also a long climate record from the Radcliffe Meteorological Station in Oxford city centre (Burt & Shahgedanova, 1998).
Chapter 2: Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications

The work presented in this chapter has been published in the following journal paper: Macdonald, D.M.J., Dixon, A.J. & Gooddy, D.C. (2018). Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications. *Ecological Engineering*, 123, 226-237. Contents lists available at ScienceDirect

Ecological Engineering

journal homepage: www.elsevier.com/locate/ecoleng

Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications

David M.J. Macdonald^{a,*}, Andy J. Dixon^b, Daren C. Gooddy^a

^a British Geological Survey, Maclean Building, Wallingford, Oxfordshire OX10 8BB, UK
^b Groundwater Monitoring & Drilling Ltd, Chalgrove, Oxfordshire, UK

ARTICLE INFO

Keywords: Groundwater Groundwater-surface water interaction River management structure Floodplain Nitrate

ABSTRACT

The management of rivers for navigation, hydropower and flood risk reduction involves the installation of inchannel structures. These influence river levels and can affect groundwater flow within hydraulically-connected riparian floodplain aquifers. A comprehensively monitored, peri-urban, lowland river floodplain in the southern United Kingdom was used to explore these dependencies and to examine the implications for the flux exchange of water and nitrate between the river and the floodplain alluvial aquifer. The study demonstrated that rivers maintained at high levels by management structures, result in raised groundwater levels in the adjacent aquifer and complex groundwater flow patterns. Engineered river management structures were shown to promote flow from river to aquifer through the river bed but the majority of the associated nitrate was removed in the hyporheic zone. High-nitrate groundwater recharge to the alluvial aquifer also occurred through overbank flood flows. Across the floodplain, substantial denitrification occurred due to anaerobic conditions resulting from carbon-rich sediments and the shallow water table, the latter linked to the river management structures. An upper limit on the total annual mass of nitrate removed from river water entering the floodplain aquifer was estimated for the study site $(2.9 \times 10^4 \text{ kg})$, which was three orders of magnitude lower than the estimate of annual in-channel nitrate flux $(1.8 \times 10^7 \text{ kg})$. However, this capacity of lowland floodplains to reduce groundwater nitrate concentrations has local benefits, for example for private and public water supplies sourced from alluvial aquifers. The insights from the study also have relevance for those considering schemes that include the installation, removal or redesign of river management structures, as the resultant change in groundwater levels may have consequences for floodplain meadows and the nutrient status of the aquatic system.

1. Introduction

Floodplains are locations of complex interactions between river water, groundwater and overland flow (Burt et al., 2002). The degree of interaction is dependent on a number of factors, including: the magnitude and direction of the head gradient between river and aquifer; the permeability of the alluvial sediments and the river bed material; and the capacity of the river channel to retain high flows (Sophocleous, 2002). Naganna et al. (2017) provide a comprehensive review of the controls on river bed permeability identifying the importance of the particle size and depth of the bed material, the river channel geometry and upstream sediment supply to the river. Colmation and bioclogging of macropores and associated lower bed permeabilities is more likely to occur in river reaches losing water to adjacent aquifers (Battin and Sengschmitt, 1999; Brunke, 1999; Krause et al., 2007; Younger et al., 1993). Given the range of controlling factors, river bed permeability will be highly spatially variable (Calver, 2001; Irvine et al., 2012). Bed scouring resulting from floods can induce temporal changes in streambed elevation and particle size composition, increasing hydraulic conductivity (Blasch et al., 2007; Doppler et al., 2007; Hatch et al., 2010). Where permeable near surface floodplain sediments occur, Doble et al. (2012) showed overbanking river water can result in substantial groundwater recharge.

The management of rivers for navigation, hydropower and flood risk reduction involves the installation of in-channel structures (Gregory, 2006). These structures are ubiquitous in many countries (Davies and Walker, 1986; Downs and Gregory, 2014). For example, within England and Wales, records from the Government environment regulator, the Environment Agency, accessed in 2014, showed that 17,569 locks, weirs and control gates were located on the 68,755 km of the river network. The operation of engineered river management structures disrupts the natural interaction of surface water and

* Corresponding author.

E-mail address: dmjm@bgs.ac.uk (D.M.J. Macdonald).

https://doi.org/10.1016/j.ecoleng.2018.09.005

Received 16 February 2018; Received in revised form 28 August 2018; Accepted 3 September 2018

0925-8574/ © 2018 United Kingdom Research and Innovation, as represented by its component body, the British Geological Survey. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/BY/4.0/).









Fig. 1. River Thames floodplain in the vicinity of the city of Oxford, UK. Areas in white are higher ground above the 1-in-100 year flood extent. Contains Ordnance Survey data © Crown copyright and database right (2018).

groundwater. Studies have shown how structures can cause groundwater levels in the associated aquifer to be raised and river reaches to switch from gaining water from the adjacent aquifer to losing water when structures are introduced (Krause et al., 2007; Matula et al., 2014; Lee et al. 2015). The aggregation of fine-grained material associated with lower river velocity upstream of structures and scouring of the river bed downstream, in combination with head gradients that increase and decrease the propensity for colmation, mean bed permeability of rivers under the influence of river structures can be highly variable (Hatch et al., 2010; Naganna et al., 2017). Attempts to address poor river ecology have included the removal of weirs to return the connectivity of river habitats (Gilvear et al., 2013) with likely changes to potentially long-standing groundwater flow patterns and levels.

Groundwater levels are a factor in determining reduction-oxidation (redox) conditions within the subsurface that in turn are a major control on the processing of nutrients (Rivett et al., 2008). Nitrate (NO_3^-) , the predominant oxidised form of nitrogen, is readily transported in water and is stable under a range of conditions. However, anaerobic carbonrich sediments, characteristic of floodplains, have the potential to support large populations of denitrifying bacteria. Shallow water tables help to create these anaerobic conditions, as the aerobic unsaturated

zone of the sediments is small (Burt et al., 2002; Kellogg et al., 2005). The rate of denitrification increases with organic matter (OM) content towards the soil surface (Burt et al., 1999) and there is a ready supply of OM to floodplains through inundation by sediment-laden river water. Pinay et al. (2000) found a significant relationship between denitrification rates in floodplain sediments and their texture; highest rates were measured in fine-textured soils with high silt and clay content. These finer-grained floodplain sediments are often found at the surface, as a result of historical clearance of natural vegetation and increased agriculture upstream (Macklin et al., 2010).

The hyporheic zone interface between river and groundwater (Burt et al., 2013) is also a hotspot for nutrient processing (Antiguedad et al., 2016; McClain et al., 2003). Exchanges of water, nutrients, and OM here occur in response to variations in discharge and bed topography and porosity (Boulton et al., 1998). Upwelling groundwater can supply stream organisms with nutrients while downwelling stream water can provide dissolved oxygen and OM to microbes and invertebrates in the hyporheic zone. The improvement to water quality resulting from the actions of the hyporheic zone are the basis of river bank infiltration schemes (Hoehn, 2002) and water sourced from such schemes can provide a large proportion of public groundwater supplies (Ascott et al.,

2016). Although many authors acknowledge the part played by denitrification processes in the riparian zone in decreasing NO_3^- concentrations, the important role of dilution is also reported (Baillieux et al., 2014; Bernard-Jannin et al., 2016; Pinay et al., 1998).

A large number of studies have investigated nutrient pollution within alluvial aquifers associated with river floodplains (e.g. Haycock and Pinay, 1993; Correll et al., 1997; Clément et al., 2003; Forshay and Stanley, 2005; Krause et al., 2008). These investigations relate primarily to the quality of water in associated rivers, and measures that can be undertaken to bring nutrient concentrations below levels that are detrimental to the ecological status of the aquatic environment. Within this body of research few studies have examined the influence of river management structures on processes that relate to nutrient cycling (e.g. Hucks Sawyer et al., 2009; Cisowska and Hutchins, 2016) and where undertaken address relatively simple hydrological settings.

The aim of the study reported here was to examine flow and nutrient dynamics within the floodplain of a large lowland river system with a high density of long-standing river management structures. Hydrogeological and water level data from the floodplain were used to assess the influence of the engineered river management structures on groundwater flows and levels in the associated alluvial aquifers, and measurements of water chemistry and simple modelling were used to estimate the flux and removal of NO₃⁻ that resulted from the cycling of water through the floodplain aquifer. The significance of the nitrate loss was assessed in terms of river NO₃⁻ flux.

The floodplain studied was that of the River Thames in the vicinity of the city of Oxford in the southern United Kingdom. The many studies undertaken in the area over the period of recent decades have characterised the hydrogeology of the sediments and resulted in an extensive water level monitoring network (summarised in Macdonald et al., 2012).

2. Materials and methods

2.1. Study area

The River Thames flows along the western edge of Oxford (Fig. 1). The floodplain within the Oxford valley has an area of 20.4 km², varying in width from 410 to 2170 m. It is bordered by high ground formed from incised Quaternary river terraces and Jurassic bedrock. The floodplain is underlain by alluvial sediments; a layer of fine-grained silts and clays over very permeable sands and gravels, with a total thickness of two to six metres (Newell, 2008). Almost all of these sediments are bounded laterally and below by low permeability bedrock of Upper Jurassic Oxford Clay. The floodplain has down-valley gradient of 0.053% but locally contains shallow channels and raised interfluves, which can influence flood water distribution.

Although the local urban area mainly occupies the high ground surrounding the floodplain, approximately 14% of the floodplain is urbanised (Fig. 1). There are 20 historical licensed 'landfills' on the floodplain (Fig. 1), with a total surface area of 1.05 km^2 . These landfills are mostly mounds of waste material sitting on the floodplain surface. A large gravel quarry, now closed, is located in the north of the floodplain (Fig. 1). Land designated as ecologically sensitive, primarily lowland floodplain meadows, occupies 3.62 km^2 (18%) of the floodplain (Fig. 1). Amongst other factors such as management practices, temperature and nutrient status, and soil pH, these types of meadows are highly sensitive to soil moisture and its temporal fluctuation that, in turn, is dependent on depth to groundwater (Wheeler et al., 2004; Punalekar et al., 2016).

The River Thames source is in the Jurassic limestone hills 60 kms to the west of Oxford. Within the Oxford valley the River Thames breaks up into a series of channels before reforming into a single channel as it flows out of the valley. The length of the River Thames in the study area is 16.1 km. The main secondary channel is the Seacourt Stream, which becomes the Hinksey Stream in the south of the valley (hereafter also referred to as the Seacourt Stream). The River Cherwell flows into the Thames to the south of the city centre. The long-term mean flow in the Thames, measured upstream of Oxford, is 18.48 m³/s (Marsh et al., 2008). Since 2000 there have been five major flooding events that have affected the urbanised areas of the floodplain. Groundwater flooding is a significant component of the flooding in the city (Macdonald et al., 2012), mainly affecting subsurface infrastructure such as the inundation of house basements and the surcharging of sewers. Eighty-five per cent of the Oxford floodplain is inundated by the modelled 1-in-100 year return period flood (Environment Agency, 2009). The percentage of inundated floodplain resulting from the July 2007 flood, which was estimated to be between a 15- and 20-year return flood (Macdonald et al., 2012), was 63%.

The Thames has six locks and associated weirs within the Oxford valley (Fig. 1), with an average separation of 3.2 km; the locks furthest upstream (Eynsham) and downstream (Sandford) define the study area. All the locks, apart from Sandford, were most recently rebuilt in the first half of the 20th Century; Sandford was rebuilt in 1972. However, in all cases there have been weirs at these locations for centuries (Thacker, 1968). The difference between the mean water level at the tail of Eynsham Lock and the head of Sandford Lock is approximately 5 m. (NB, in Fig. 1 both the lock names and an alphanumeric identification are given, however in the remainder of the paper only the latter will be used when referring to the locks.)

The locks in Oxford are typical of those found on the non-tidal River Thames (354 km in length). Thirty-three locks are located over a 198 km reach of the river, with an average separation of 6.2 km. The purpose of the locks is to maintain the river upstream at navigable levels, higher than those that naturally occurred prior to their construction, enabling boats to move between the upstream and downstream levels.

2.2. Monitoring infrastructure and data

The water levels and flow regime in the floodplain aquifer system were investigated within the study area. Water chemistry from samples taken within a sub area of the floodplain (see Fig. 1) were used to examine nutrient dynamics.

There is a dense network of water level monitoring sites within the Oxford study area (Fig. 2). The study used data from 51 sites at which water levels were monitored over the previous three decades. Surface water monitoring sites were a combination of stilling wells with digital water level loggers, gaugeboards and locations, such as bridges, with known datums from which water levels were measured. Groundwater monitoring sites were drilled boreholes with diameters from 5 to 20 cm, completed in the gravel aquifer at least 1 m below the estimated minimum groundwater level. Measurements were made at the groundwater monitoring sites with a combination of digital water level loggers and manual water level meters. The monitoring network included eight paired surface water and groundwater level monitoring sites. At some locations these paired sites were combined with other groundwater sites to form water level monitoring transects. Water levels at monitoring sites not instrumented with loggers were measured manually as part of a series of floodplain-wide surveys. These surveys were undertaken on a monthly basis from May 2007 to March 2010.

River levels were also obtained from the Environment Agency. It monitors the upstream (head) and downstream (tail) river levels at five locks on the River Thames in the Oxford area (L2 to L6), as well as at four sites in the secondary streams (S1 to S4).

All monitoring sites had datums, the heights of which were surveyed relative to mean sea level. A map of groundwater level contours is presented in Section 3.1.3. This map is based on groundwater and surface water level measurements made over a two-day period in May 2007 when no rainfall occurred, converted to water levels relative to sea level, and hand-contoured. A raster dataset was created from these contours using the 'Topo to Raster' tool within ArcGIS (ESRI, 2015).



Fig. 2. Water level monitoring network and superficial geology. Sites referred to in the text are labelled. Note, some of the surface water monitoring sites are located on minor water courses not included in the figure. Contains Ordnance Survey data © Crown copyright and database right (2018).

This was combined with a digital elevation model, obtained using the Light Detection And Ranging (Lidar) surveying method by the Environment Agency (data licensed under the UK Open Government Licence v3.0), to produce a map of depth to groundwater.

Precipitation data for the area were obtained from the Radcliffe Meteorological Station in central Oxford (51° 45′40 N, 1° 15′50 W; Burt and Shahgedanova, 1998), via the archive of the Centre for Environmental Data Analysis (www.ceda.ac.uk). The mean annual precipitation and air temperature for 1986–2015, were 670 mm and 10.6 °C, respectively.

2.3. Groundwater nutrient concentrations

2.3.1. Water sampling

The focussed nutrient study was undertaken within a zone stretching across the width of the floodplain (Fig. 1) to better understand nutrient cycling in the subsurface, assess the importance of the hydrological regime in controlling this cycling, and quantify the flux of NO_3^- (note, all concentrations and fluxes are NO_3^- , rather than NO_3^- -N). The study area included: to the east of the River Thames, an area of ecologically protected land used for communal grazing (Port Meadow) and a large waste dump (Burgess Field); and to the west, an area used primarily for pasture (Fig. 2). In this area, when the River Thames is out-of-bank, made ground immediately to the west forces river water to flood across Port Meadow to the east.

In addition to water level measurements, water samples were taken from within the area from 23 boreholes completed in the alluvial sediments (Gooddy et al., 2014), as well as from the River Thames (Fig. 2). A series of samples were obtained during 14 sampling rounds over the period May 2010 to March 2015. A minimum of three borehole volumes were purged from each site, and samples were collected when stable readings for pH, electrical conductivity (EC) and dissolved oxygen (DO) were obtained. Samples for chloride (Cl⁻) and nitrogen species were filtered and collected in 30 ml plastic bottles. The samples were analysed for Cl⁻ and NO₃⁻ using ion chromatography, and ammonium (NH₄⁻) by flow colorimetery. Samples for dissolved organic carbon (DOC) analysis were collected and filtered through 0.45 µm silver filters into 10 ml glass bottles and measured by the standard technique of acidification to pH < 3, then conversion to CO₂ by 680 °C combustion catalytic oxidation (Pt catalyst), followed by high sensitivity infra-red analysis of the gas. All analyses were carried out in the British Geological Survey's laboratories that are accredited by the UK Accreditation Service. Field data, including bicarbonate (HCO₃⁻), pH, temperature, EC and DO, were all determined at site; a flow-through cell was used for unstable field parameters to ensure representative insitu values were obtained.

2.3.2. Nitrate flux modelling

The assessment of nutrient cycling within this study focussed on the processing of groundwater NO3⁻. A conceptual model of groundwatersurface water interaction was developed in the study through an analysis of the water level data collected, in combination with the threedimensional geological model of the floodplain aquifer (Newell, 2008). To examine dominant processes controlling groundwater NO3⁻ concentrations, a simple single-cell mixing model was set up for the floodplain aquifer in the focussed nutrient study area and applied separately to the zones of the aquifer to the east and to the west of the River Thames. The model was oriented perpendicular to the river, approximately along a groundwater flow path (see Section 3.1.3). The model assumed a constant thickness aquifer, with the thickness averaged from the three-dimensional geological model; this was considered reasonable given the lack of variability in the thickness of the floodplain aquifer in this area (Newell, 2008). The model boundary conditions included lateral inflow (QL), rainfall recharge (RP) and river flood water recharge (R_B). A water balance was maintained by making lateral discharge from the cell, the flow from the aquifer to the river (Q_A) , equal to the total input over the period of a year (Fig. 3; Eq. (1)).

$$R_P + R_R + Q_L = Q_A \tag{1}$$

Lateral inflow was calculated using Darcy's Law. The hydraulic conductivity used was within the range specified for the Oxford floodplain gravels by Dixon (2004) and the hydraulic gradient was based on averaged observed groundwater levels for the modelled zone. Rainfall recharge was approximated using the output of a soil moisture balance model (Mansour and Hughes, 2004). Flood water recharge was estimated based on the volume of unsaturated material and the frequency of flooding, assuming all the unsaturated material was filled and recharged the aquifer, and the remainder of the flood water was rejected. The flood water recharge calculation used: i) an averaged depth to groundwater for winter months (i.e. December, January and February) over the 7 years prior to 2015, based on measurements from water level loggers in the zones of interest; ii) porosity of the alluvium measured within the floodplain (Hodgson, 2008; Gardner, 1991); and

iii) a flood frequency based on the occurrence of major floods in the period of the study, defined by occurrences of river levels at T1 (Fig. 1) rising over 57.6 m above mean sea level (masl). The potential for the retention of flood water on the floodplain in topographical lows that allowed delayed recharge, was not accounted for. This meant the model may have slightly underestimated the flux of NO_3^- to the floodplain aquifer.

Nitrate concentrations associated with each of the inputs were based on the median of measurements obtained through the river and groundwater sampling campaign. However, the sensitivity of the model was tested using a wide range of flood water NO_3^- concentrations, recognising the non-linear relationship between river NO3⁻ concentrations and river flows that have been measured in the River Thames (Neal et al., 2006). Nitrate inputs not included were those associated with: to the east, waterfowl, grazing horses and livestock, as these import a negligible amount of nitrogen; and to the west, pasture, as no fertiliser was added to this land. The model calculated NO3concentrations assuming complete mixing within the aquifer cell; this was reasonable as the aquifer is thin, homogeneous and highly permeable and almost fully incised by the river (Newell, 2008). The model ran on a time step of one year, which was considered suitable given that the average residence time, estimated using Darcy's Law and the parameters given in Table 1, was greater than one year. The NO₃ concentration at the end of timestep i + 1 was calculated using Eq. (2)

$$C_{A_{l+1}} = \frac{C_R.\ R_{R_{l+1}} + C_L.\ Q_L + C_{A_l}.\ (A - R_P - R_{R_{l+1}} - Q_L)}{A} \tag{2}$$

where: C_R is the NO₃⁻ concentration in the river recharge water; $R_{R_{i+1}}$ is the river flood water recharge over timestep i + 1; C_L is the NO₃⁻ concentration in the lateral inflowing groundwater; C_A is the NO₃⁻ concentration in the aquifer cell; and A is the volume of the aquifer cell, equal to the width of the floodplain times the depth of alluvial sediments, multiplied by the porosity. It was assumed the NO₃⁻ concentration in the rainfall was negligible. A NO₃⁻ removal factor was used to match the average modelled NO₃⁻ concentration over one flood cycle with the average observed NO₃⁻ concentrations in all boreholes in the relevant zone for the period 2010 to 2015.

3. Results and discussion

3.1. Water levels and flows in the Oxford floodplain

3.1.1. Surface water levels

Water levels at the locks and intervening monitoring sites on the River Thames, and at sites on the Seacourt Stream were used to characterise the spatio-temporal variability of the river network within the Oxford area. The influence of the locks on river levels and the



Fig. 3. Schematic of flows within the single-cell mixing model. See Section 2.3.2 for definition of terms.

Table 1

Parameters and input variables for the mixing-cell model of groundwater and NO_3^- in the aquifers on the east and west of the River Thames (Section 2.3) and necessary denitrification to match observed concentrations. Where the model is sensitive to parameters/input variables, values are shown in brackets that produce higher and lower aquifer NO_3^- concentrations and denitrification factors.

Parameters/input variables	unit	East of R. Thames	West of R. Thames
Aquifer thickness	m	2.5 (1.5/3.5)	2.5 (1.5/3.5)
Aquifer width	m	500	500
Aquifer porosity	-	0.2 (0.1/0.3)	0.2 (0.1/0.3)
Aquifer hydraulic conductivity	m/d	200	200
Groundwater level gradient	-	0.0004	0.00125
NO ₃ ⁻ concentration of lateral inflow	mg/L	0	2.3 (4.6/1.5)
Annual rainfall recharge	m	0.1	0.1
Unsaturated zone depth prior to flood inundation	m	0.13 (0.20/0.08)	0.73 (1.00/0.50)
Alluvium porosity	-	0.4 (0.5/0.3)	0.4 (0.5/0.3)
Flood event frequency	years	2 (1.5/3)	2 (1.5/3)
Flood water NO ₃ ⁻ concentration	mg/L	25 (35/15)	25 (35/15)
Resultant modelled aquifer NO ₃ ⁻ concentration	mg/L	6.69 (28.16/0.94)	10.56 (35.25/3.04)
Average observed aquifer NO ₃ ⁻ concentration	mg/L	0.04	2.62
Denitrification factor required to match modelled and observed	-	0.98 (1.00/0.79)	0.76 (0.95/0.07)

comparison with stream water levels is shown by a series of box plots produced from monitoring data for the period 1 April 2009 to 31 March 2013 (Fig. 4; Supplementary Information Table S1). The differences between median head and tail water levels at the locks range from 0.76 to 1.75 m. The impact of these steps in river level was that river gradients between locks on the River Thames (calculated using the median lock water levels) were small (average of 0.003%) compared with the gradients between monitoring sites on the Seacourt Stream (average of 0.034%). The different characteristics of lock heads and tails are illustrated (Fig. 4): tail water levels had an asymmetric distribution with a greater interquartile range and higher peak levels, similar to that of the more naturally flowing Seacourt Stream; and head water levels had an interquartile range that is an order of magnitude smaller than the tail water levels.

If it is assumed that lock tail levels are representative of 'natural' levels and, were locks not in place, the river levels between adjacent lock locations are proportional to distance along the river reach, then the differences between the current managed river levels and the natural river levels can be estimated. For example, the difference between managed and natural median river levels at T1, T2 and T3 would be 0.95, 0.69 and 0.61 m, respectively.

3.1.2. Groundwater-surface water interaction

Relative water levels within groups of monitoring sites were examined to assess the interaction of surface water bodies and aquifers. These relationships helped with the contouring of point measurements



Fig. 4. Box plots of water levels at sites on the Seacourt Stream and the River Thames, from downstream (left) to upstream (right), for 1 April 2009 to 31 March 2013. For data used to produce the box plots see Supplementary Information Table S1. Lengths of reaches between monitoring sites (km) are shown above the arrows. See Fig. 2 for locations. In the case of the locks, levels are plotted for the downstream (d/s) and upstream (u/s) sides. Contains Environment Agency data licensed under the Open Government Licence v3.0.



Fig. 5. Groundwater and surface water levels at three locations. Black lines are the river/stream water levels and other lines are groundwater levels (see Fig. 2 for locations): a) paired groundwater/surface water sites on the Seacourt Stream; b) a transect perpendicular to the River Thames upstream of L5; c) north-east to south-west transect through the River Thames at Port Meadow.

of water levels. The contours allowed groundwater flow directions to be determined and depths to groundwater from the ground surface to be mapped (see Section 3.1.3).

For the secondary streams, data show that surface water levels and adjacent groundwater levels had a very similar fluctuation pattern (e.g. Fig. 5a), and that the network of streams were gaining groundwater for the majority of the time. The interaction of aquifer and surface water body was different for the River Thames. In general, along reaches of the Thames upstream of the locks, for example upstream of L5 (Fig. 5b), steep gradients from the river towards the adjacent aquifer occurred for the majority of time. River bed elevation profiling, undertaken as part

of the hydraulic modelling of the River Thames (Environment Agency, 2009), also suggests a saturated hydraulic connection between river and aquifer was maintained, i.e. the river did not become perched with an intervening unsaturated zone.

Along the reach of the River Thames between L3 and L4, adjacent to Port Meadow, the groundwater-river dynamics were different to those upstream of L5. To the west of the river, consistent steep gradients from river (T1) to aquifer (T1c) were again evident (Fig. 5c). To the east of the river, however, groundwater levels (T1a) were, for the majority of the time, above the river level. The groundwater gradient to the east of the river is likely to be due to a combination of two aspects: when flooding occurs it is always to the east, due to the raised bank to the west, resulting in relatively high groundwater recharge in this area; and the maintenance of a high river stage, due to the downstream lock, acts as a barrier to groundwater flowing westward towards the Seacourt Stream (see Section 3.1.3). The dynamics during periods of overbank flooding are illustrated in the winter months of 2009/10 in Fig. 5c. When the river overflowed its bank (compare the river level with the ground level at T1a, Fig. 5c), the groundwater level rose to match the river level. The groundwater levels took longer to recess than the river and, for a time, surface water on the floodplain was the result of groundwater flooding.

During the seasonal dry period, groundwater levels to the east of the river fell sufficiently to cause a temporary local reversal of the flow direction. Again, the depth of the river bed was such that the hydraulic connection between the river and aquifer was maintained during these periods.

3.1.3. Groundwater levels and flow patterns

In Fig. 6, maps of contoured groundwater levels relative to mean sea level (Fig. 6a) and depth to groundwater (Fig. 6b) are presented for May 2007. The groundwater flow lines and the relative depths of groundwater are typical for the alluvial aquifer in Oxford for all but the short periods when very high river levels occurred.

The flow lines show the complex patterns associated with the river management structures (Fig. 6a). Groundwater mounds coincide with raised river levels upstream of the locks. The steep water level gradients from river to aquifer in these reaches suggest that the river bed hydraulic conductivity is much lower than that of the floodplain aquifer. The low river bed hydraulic conductivity is expected as the water has a greater depth as a result of the locks, and therefore flows relatively slowly, allowing a greater proportion of fine sediment to be deposited. The positive river-to-groundwater gradient also enhances colmation.

The water entering the adjacent aquifer from the River Thames flows towards the lock bypass channels or nearby smaller streams, which form lines of groundwater discharge. The contours also show a groundwater mound created upstream of weir W2 on the Seacourt Stream in the southern part of the floodplain. It is notable in this area that a groundwater trough occurred between the River Thames and this mound, coincident with the narrow urbanised strip running north to south (compare Figs. 1 and 6a). It is postulated that the subsurface infrastructure associated with the urbanised area, such as the network of sewers and storm drains, provide a route for groundwater to discharge here, drawing down the groundwater level.

Another notable flow pattern is from north-east to south-west across the aquifer in the Port Meadow area, through the line of the River Thames, towards the Seacourt Stream. This flow pattern indicates that there is likely to be some recharge to the alluvial aquifer from higher terrace gravels to the north-east but also identifies discharge to the Seacourt Stream as having a strong influence on groundwater flows. The contours show that the level of the River Thames is higher than that of the Seacourt Stream by tens of centimetres.

As discussed in Section 3.1.2, the groundwater level gradient to the east of the River Thames was relatively low compared with that to the west, suggesting the maintenance of high levels of the River Thames in this area inhibits lateral groundwater flow. Data from the water level



Fig. 6. a) contoured groundwater levels and groundwater flow lines within the alluvial floodplain aquifer; and b) depth to groundwater within the alluvial aquifer. Both maps are based on water levels measured in May 2007. Refer to Fig. 1 for elements not included in the legend. Contains Ordnance Survey data © Crown copyright and database right (2018).

loggers for the period 1 March 2010 to 28 February 2015 in boreholes T1a and T1d (Fig. 7), located on the western and eastern sides, respectively, of the River Thames in the Port Meadow area, highlight that the depth to groundwater in the floodplain aquifer on the eastern side was generally shallower (median: 1.01 mbgl in T1d; 0.24 mbgl in T1a). Geological logs from these boreholes show that for the majority of the time at T1a, the unsaturated zone remained within the fine-grained alluvium, whereas at T1d it extended to the underlying gravels.

The map of contoured depth to groundwater (Fig. 6b) highlights where the gravels of the modern floodplain rise up the valley sides, and the areas of man-made ground (e.g. Burgess Field licensed landfill), both resulting in relatively large depths to groundwater.

A comparison of maps in Fig. 6 shows that the raised levels of the River Thames created by the locks are associated with areas of aquifer where groundwater levels were relatively shallow, in the case of May 2007 often within 0.5 m of the ground surface (in red). Fig. 6b highlights the co-location of floodplain meadows and areas with relatively shallow groundwater. It may be that this co-location is, in part, related to waterlogging in the area upstream of river management structures, which historically made urban development more problematic, allowing floodplain meadows to survive. However, the co-location may also be because the soil moisture conditions associated with the shallow groundwaters created by the river management structures are suited to rare floodplain meadows plant communities. These plant communities have been shown to be highly sensitive to: soil moisture conditions, with centrimetric differences in groundwater level being linked with notably different plant assemblages (Silvertown et al., 2015); and the range of groundwater level fluctuations, influenced for example by

dams and dykes (Leyer, 2005). Where river restoration schemes have been undertaken that have involved the raising of groundwater levels in the associated alluvial aquifers through the removal of deeply incised channels, studies of pre- and post-intervention have shown these are linked with substantial changes in floodplain vegetation composition (Loheide and Gorelick, 2007; Hammersmark et al., 2009).

3.2. Groundwater nitrate and associated parameters

Selected parameters measured at sites within the area, associated with nutrient cycling, are presented in Fig. 7 in the form of box plots. These box plots include all measurements from the October 2010 to February 2015 period. This dataset includes the same number of winter and summer sampling rounds (note, measurements below the detection limit are included in the box plots as half of the detection limit). The geographical grouping of sample sites is described in the caption of Fig. 7.

The data highlight the influence of the waste dump on the groundwater chemistry. High concentrations of DOC and NH_4^- in the LF group (medians 8.0 and 40 mg/L, respectively) are indicators that the waste dump is a significant pollution source. The NH_4^- concentrations in the LF group are possibly due to the reducing conditions in the alluvial sediments (median DO 0.47 mg/L). Anaerobic conditions and available OC also promote denitrification, which is consistent with the nitrous oxide (N₂O) concentrations measured by Gooddy et al. (2014), denitrification being the primary process that produces N₂O in groundwater (Jurado et al., 2017). Nitrate concentrations are above detection limit in only 38 of the 105 samples obtained from boreholes



Fig. 7. Box plots of depths to groundwater and water chemistry for a series of monitoring sites (see: Fig. 1 for extent of this area within the overall study area; Fig. 4 for box plot legend; and Supplementary Information Table S1 for data used to produce the box plots). Sample sites are categorised as: groundwater west of the River Thames (WT); River Thames surface water (SW); groundwater east of the Thames in the floodplain under the influence of the Burgess Field landfill (LF); and groundwater east of the Thames in the other areas of the floodplain (FP). Labels are included for sites referred to in the text. Contains Ordnance Survey data © Crown copyright and database right (2018).

within the LF group. The median NO_3^- concentration in the LF group is low (0.04 mg/L), as is the standard deviation (0.26 mg/L), indicating there is limited spatial and temporal variability. Given the neutral pH levels in groundwater (Gooddy et al., 2014), high concentrations of HCO₃⁻ (median 823 mg/L) are also evidence of the oxidation of OC and denitrification (Vidon et al., 2010). Groundwaters from the FP group have comparatively low concentrations of NH₄⁻ (median 0.10 mg/L), low DO (median 0.32 mg/L), and high DOC and HCO₃⁻ concentrations (medians 2.5 and 393 mg/L, respectively). Very low groundwater NO₃⁻ concentrations (median 0.03 mg/L) suggest substantial denitrification in this zone, as in LF. The standard deviation in the NO₃⁻ concentration is low (0.05 mg/L), again indicating limited spatio-temporal variability. The River Thames frequently flows out of bank onto the floodplain in the Port Meadow area to the east of the river; during the period of sampling there were two major floods. The NO_3^- concentration in the river water sampled as part of this study at T1 is high (median and interquartile range of 25 and 6.3 NO_3^- mg/L, respectively), in contrast to the groundwater. The low groundwater NO_3^- concentrations that occurred in the gravel aquifer here could be due to a range of factors: the limitation on recharge of high NO_3^- flood waters to the aquifer due to the shallow water table; dilution caused by low NO_3^- groundwater inflowing laterally to the area from the alluvial aquifer to the north; rainfall recharge; or NO_3^- removal by denitrification.

The mixing-cell model described in Section 2.3 was used to examine the likely contribution of each of these factors. Table 1 has values for the model parameters and input variables for the eastern zone of the River Thames (Port Meadow), as well as the resulting groundwater NO_3^- concentration, and the degree of denitrification required to match the observed average concentration here. Where the model is sensitive to parameters and input variables, low and high estimates of these are included to indicate the uncertainty in the denitrification factor calculated.

The model indicates that the groundwater NO_3^- concentrations are insensitive to both concentration of groundwater flowing laterally into this area of the aquifer, and the rainfall recharge. The model does show that the frequency of flooding, the NO_3^- concentration of river flood waters and the depth of groundwater prior to the flood are potentially major factors in determining the flux of NO_3^- into the floodplain aquifer. The model also shows that significant biogeochemical processing must be taking place for groundwater NO_3^- to remain as low as observed. The model requires a high NO_3^- removal factor for simulated NO_3^- concentrations to match the observed concentrations.

Nitrate concentrations in groundwater to the west of the river are higher than to the east, with a median of 0.90 mg/L, over an order of magnitude greater. The DO concentrations (median 0.40 mg/L) are similar to those to the east of the river. As presented in Section 3.1.3, water level contouring indicates groundwater flows north-east to southwest through the floodplain sediments and the line of the River Thames, towards the Seacourt Stream. A comparison of the EC of samples from a transect across the River Thames is used to estimate the proportion of the flow of groundwater in the western zone of the aquifer that is sourced from the river (assuming EC is conservative). The transect includes borehole T1e (44 m to the east of the River Thames), the river itself, and borehole T1c (42 m to the west of the river). Based on the median ECs (1773, 629 and 677 µS/cm, respectively) it is estimated that 95% of the lateral flow in the aquifer to the west of the Thames comes from the river. Although this river water had high NO₃⁻ concentration, the concentration in borehole T1c is low in comparison (median 2.3 mg/L), indicating that there is likely to be significant denitrification in the hyporheic zone of the river.

The mixing-cell model was also applied to the west of the Thames, using a lateral inflow of water with a NO_3^- concentration based on the average value from T1c. The model again identifies the frequency of flooding, the NO_3^- concentration of river flood waters and the depth of groundwater as important controls on the flux of NO_3^- into the floodplain aquifer but, in addition, that the lateral flow of water from the River Thames, with raised NO_3^- concentration that occurs in this zone of the aquifer (Table 1), with its deeper water table, is less than to the east of the river although this has a greater degree of uncertainty. It is possible that the lower degree of denitrification compared with the eastern zone of the focussed study area is due in part to the deeper groundwater, that rises into the fine-grained alluvium for a shorter proportion of the year.

It is acknowledged that the mixing model is a simplified representation of the system that makes a number of assumptions about the flows within the floodplain system. However, it does indicate that it is very likely that overall there is substantial removal of NO₃⁻ within the floodplain sediments. This is important for the chemical quality of public and private groundwater supplies sourced from localised shallow aquifers. This indicates that in settings similar to those of the study area it may not be necessary to put in place measures to control river water quality if the primary purpose is to improve groundwater quality.

In this study it was not possible to compare pre- and post-construction conditions due to the historical nature of the engineered river management structures within the floodplain. However, evidence presented in Section 3.1 does show the influence of the locks on river and groundwater levels. There is a limited amount of peer-reviewed research that has been undertaken examining the influence of changes to structures in rivers on the nutrient status of groundwater in the associated aquifer through measurements before and after the intervention (Bellmore et al., 2017). However, in a study related to the impacts of the construction of small temporary dams, Hill and Duval (2009) showed that the rise in groundwater levels that occurred in the adjacent alluvial aquifer resulted in a significant reduction in groundwater NO_3^- concentrations; raised groundwater levels in the Oxford floodplain associated with the locks may have a similar influence.

3.3. Nitrate fluxes

This study highlights that in settings where there is no lateral regional-scale inflow of high-NO₃⁻ concentration groundwater, as in the case of Oxford where the floodplain aquifer is isolated by the underlying poorly permeable bedrock, river inflow can be the primary input of NO₃⁻ to groundwater. The influence of the river management structures on NO₃⁻ fluxes in the context of Oxford is complex. On river reaches influenced by the river management structures, where river levels were raised, river water recharged the alluvial aquifer due to the positive gradient from the river to the aquifer. The groundwater chemistry sampling shows that the hyporheic zone is highly efficient at reducing NO₃⁻ concentrations. To provide an approximation of the mass of NO₃⁻ that could be removed from recharging river water flowing through the river bed, we estimated a mass-balance for a section of the River Thames in the western zone of the focussed nutrient study area. The volume of water flowing through the river bed was estimated using Darcy's Law; parameter values were chosen to maximise the annual estimate of mass of NO3⁻ removed. Parameters used were: depth of river 2 m (typical depth of the River Thames in the Oxford valley; Environment Agency, 2009); river-to-aquifer gradient 5%, based on median water levels over the period 2010 to 2015 at T1 and T1c; river bed permeability of 1 m/d (the maximum quoted for the River Thames by Younger et al., 1993); and a river NO_3^- concentration of 25 mg/L (the median concentration measured during the period of the focussed nutrient study; see Section 3.2). Assuming denitrification of 90% in river water passing through the river bed (as was seen to the west of the River Thames, based on median concentrations measured during the period of the focussed nutrient study), the mass of NO₃⁻ removed, calculated using these values, is 1.0×10^3 kg/a/km.

The implication from the groundwater chemistry in the eastern zone of the focussed nutrient study, and the mixing-cell modelling, is that the majority of NO₃⁻ that infiltrates the aquifer as a result of river overbanking onto the floodplain is removed through denitrification. Using the maximum river-water aquifer recharge calculated in the mixing-cell model, an average annual mass of NO₃⁻ that could potentially be removed in this zone is estimated as 0.8×10^3 kg/a/km.

These calculations provide an upper estimate of the NO₃⁻ removed through interaction between river and alluvial aquifer for a setting such as Oxford in which the underlying impermeable bedrock limits lateral inflow of high NO₃⁻ groundwater. If we apply the per kilometre NO₃⁻ removal to the full length of the River Thames through the Oxford valley, then the annual mass is 2.9×10^4 kg. For comparison, the annual River Thames in-channel flux of NO₃⁻ was calculated using the mean annual flow entering the study area and the river NO₃⁻ concentration quoted in Table 1. The resultant mass of 1.8×10^7 kg is three orders of magnitude greater than the flux of river water NO₃⁻ into the floodplain sediments. Even though the evidence from the study indicates most of this NO₃⁻ would be removed from the aquatic system through denitrification, its removal would only have a small impact on the downstream NO₃⁻ flux.

4. Conclusions

High spatio-temporal density of water level monitoring and a series of water quality surveys provided detailed insight into the hydrology and water chemistry of a river system and its associated alluvial aquifer. Although the Oxford floodplain setting has its own specific characteristics, it was used here to highlight general issues relating to the influence of managed rivers on groundwater flows within floodplain aquifers, and the implications for the flux of NO_3^{-1} .

Engineered river management structures are commonplace both along the length of the River Thames, and in England and Wales in general, with references indicating their widespread occurrence in other countries. The study shows the degree to which river management structures can control river levels. Raised river levels in reaches upstream of structures were shown to create zones of increased groundwater storage within adjacent alluvial aquifers. As a result other smaller water courses can become important locations for groundwater discharge. Such discharge patterns result in complex flows that cause water entering the aquifer to follow significantly longer flow paths to reach surface water discharge zones than might occur under a more natural hydrological regime.

The study highlights that in settings where there is no lateral regional-scale inflow of high NO_3^- concentration groundwater, a primary input of NO_3^- to groundwater can be from the inflow of river water to the alluvial aquifer when the river is in flood and via the river bed where there is a positive hydraulic gradient from river to aquifer.

In relation to NO₃⁻ fluxes, the influence of the river management structures can be complex, as illustrated in the Oxford study. Here, on river reaches influenced by the river management structures, the positive gradient from the river to the aquifer can increase the flux of NO₃⁻ into the floodplain aquifer, however, the hyporheic zone has been shown to be highly efficient at denitrifying the water.

Raised groundwater levels, associated with river management structures, also create conditions that help to control NO3⁻ concentrations within the alluvial aquifer: a shallow water table limits the volume of high NO_3^- water that can infiltrate through the floodplain; a shallow unsaturated zone contained within the carbon-rich, finegrained sediments promotes anaerobic conditions; and longer residence times associated with complex flowpaths mean more time for groundwater denitrification to occur. The simple mixed-cell modelling undertaken within this study indicates that a large proportion of the NO₃⁻ entering floodplain aquifers under similar conditions is likely to be removed by denitrification. However, as efficient as floodplain aquifers may be locally in removing the influx of NO3⁻ where very shallow groundwater conditions occur, as this study has shown, the amount removed may be a small proportion of the in-channel flux where river NO₃⁻ concentrations are high due to upstream inputs, such as point source discharges and high NO3⁻ groundwater baseflow associated with agricultural activities.

The study also identified that shallow groundwaters associated with river management structures were spatially correlated with the location of protected floodplain meadows. Other research is highlighted in which modifications to engineered river management structures, such as those related to river restoration schemes, that resulted in changes to groundwater levels, have had significant impacts on sensitive floodplain vegetation, as well as on the potential for the removal of nutrients from the aquatic system.

Acknowledgements

This study was funded primarily by the UK Natural Environment Research Council (NERC) National Capability resources, devolved to the British Geological Survey. Components of the monitoring infrastructure and data collection were funded by the Environment Agency of England and Hanson UK, and by the NERC-funded project "Floodplain Underground Sensors - a high-density, wireless, underground sensor network to quantify floodplain hydro-ecological interactions" [grant number NE/1006923/1]. Thomas and Sam Dixon and Roy Mogdridge of Groundwater Monitoring and Drilling Ltd are thanked for the many years of water level monitoring across the Oxford floodplain. Thanks are also due to those members of staff of the British Geological Survey (BGS) who undertook water sampling and analysis as part of this study, including Peter Williams, Dan Lapworth, Debbie White and Barry Townsend. Appreciation is given for their advice and support to Chris Jackson, Majdi Mansour and Marco Bianchi (BGS), and Andrew Wade and Anne Verhoef (University of Reading). David Macdonald and Daren Gooddy publish with the permission of the Executive Director, BGS (NERC).

Declarations of interest

None.

Appendix A. Supplementary data

Supplementary data associated with this article can be found, in the online version, at https://doi.org/10.1016/j.ecoleng.2018.09.005.

References

- Antiguedad, I., Zabaleta, A., Martinez-Santos, M., Ruiz, E., Uriarte, J., Morales, T., Comin, F.A., Carranza, F., Español, C., Navarro, E., Bodoque, J.M., 2016. A simple multicriteria approach to delimitate nitrate attenuation zones in alluvial floodplains. Four cases in south-western Europe. Ecol. Eng.
- Ascott, M.J., Lapworth, D.J., Gooddy, D.C., Sage, R.C., Karapanos, I., 2016. Impacts of extreme flooding on riverbank filtration water quality. Sci. Total Environ. 554, 89–101.
- Baillieux, A., Campisi, D., Jammet, N., Bucher, S., Hunkeler, D., 2014. Regional water quality patterns in an alluvial aquifer: direct and indirect influences of rivers. J. Contam. Hydrol. 169, 123–131.
- Battin, T.J., Sengschmitt, D., 1999. Linking sediment biofilms, hydrodynamics, and river bed clogging: evidence from a large river. Microb. Ecol. 37 (3), 185–196.
- Bellmore, J.R., Duda, J.J., Craig, L.S., Greene, S.L., Torgersen, C.E., Collins, M.J., Vittum, K., 2017. Status and trends of dam removal research in the United States. Wiley Interdiscip. Rev. Water 4 (2), 1164.
- Bernard-Jannin, L., Sun, X., Teissier, S., Sauvage, S., Sánchez-Pérez, J.M., 2016. Spatiotemporal analysis of factors controlling nitrate dynamics and potential denitrification hot spots and hot moments in groundwater of an alluvial floodplain. Ecol. Eng.
- Blasch, K.W., Constantz, J., Stonestrom, D.A., 2007. Thermal methods for investigating ground-water recharge (No. 1703-1). Geological Survey (US).
- Boulton, A.J., Findlay, S., Marmonier, P., Stanley, E.H., Valett, H.M., 1998. The functional significance of the hyporheic zone in streams and rivers. Annu. Rev. Ecol. Syst. 29, 59–81.
- Brunke, M., 1999. Colmation and depth filtration within streambeds: retention of particles in hyporheic interstices. Int. Rev. Hydrobiol. 84 (2), 99–117.
- Burt, T.P., Matchett, L.S., Goulding, K.W.T., Webster, C.P., Haycock, N.E., 1999. Denitrification in riparian buffer zones: the role of floodplain hydrology. Hydrol. Process. 13, 1451–1463.
- Burt, T.P., Pinay, G., Matheson, F.E., Haycock, N.E., Butturini, A., Clement, J.C., Danielescu, S., Dowrick, D.J., Hefting, M.M., Hillbricht-Ilkowska, A., Maitre, V., 2002. Water table fluctuations in the riparian zone: comparative results from a pan-European experiment. J. Hydrol. 265, 129–148.
- Burt, T.P., Pinay, G., Grimm, N., Harm, T., 2013. Conservation and the Riparian Zone. In: River Conservation: Challenges and Opportunities, pp. 217.
- Burt, T.P., Shahgedanova, M., 1998. An historical record of evaporation losses since 1815 calculated using long-term observations from the Radcliffe Meteorological Station, Oxford, England. J. Hydrol. 205 (1), 101–111.
- Calver, A., 2001. Riverbed permeabilities: Information from pooled data. Groundwater 39 (4), 546–553.
- Cisowska, I., Hutchins, M.G., 2016. The effect of weirs on nutrient concentrations. Sci. Total Environ. 542, 997–1003.
- Clément, J.C., Aquilina, L., Bour, O., Plaine, K., Burt, T.P., Pinay, G., 2003. Hydrological flowpaths and nitrate removal rates within a riparian floodplain along a fourth-order stream in Brittany (France). Hydrol. Process. 17 (6), 1177–1195.
- Correll, D.L., Jordan, T.E., Weller, D.E., 1997. Failure of agricultural riparian buffers to protect surface waters from groundwater nitrate contamination. Forest 22, 23.
- Davies, B.R., Walker, K.F. (Eds.), 1986. The Ecology of River Systems. W. Junk, Dordrecht, pp. 793.
- Dixon, A.J., 2004. Hydrology of the Oxford Meadows cSAC. Environment Agency Report. Doble, R.C., Crosbie, R.S., Smerdon, B.D., Peeters, L., Cook, F.J., 2012. Groundwater
- recharge from overbank floods. Water Resour. Res. 48 (9). Doppler, T., Franssen, H.J.H., Kaiser, H.P., Kuhlman, U., Stauffer, F., 2007. Field evidence
- of a dynamic leakage coefficient for modelling river-aquifer interactions. J. Hydrol. 347 (1), 177–187.
- Downs, P., Gregory, K., 2014. River Channel Management: Towards Sustainable Catchment Hydrosystems. Routledge.
- Environment Agency, 2009. Oxford Flood Risk Management Strategy: Hydraulic modelling – model build and calibration final report. Environment Agency Report, Bristol, UK.
- ESRI (Environmental Systems Research Institute), 2015. ArcGIS Desktop: Release 10.3.1, Environmental Systems Research Institute, Redlands, California.
- Forshay, K.J., Stanley, E.H., 2005. Rapid nitrate loss and denitrification in a temperate river floodplain. Biogeochemistry 75 (1), 43–64.

Gardner, C.M.K., 1991. Water regime of river meadows: Yarnton Mead case study. Wallingford: Institute of Hydrology Report to River and Coastal Engineering Group, Ministry of Agriculture Fisheries and Food.

Gilvear, D.J., Spray, C.J., Casas-Mulet, R., 2013. River rehabilitation for the delivery of multiple ecosystem services at the river network scale. J. Environ. Manage. 126, 30–43.

- Gooddy, D.C., Macdonald, D.M.J., Lapworth, D.J., Bennet, S., Griffiths, K.J., 2014. Nitrogen sources, transport and processing in peri-urban floodplains. Sci. Total Environ. 494, 28–38.
- Gregory, K.J., 2006. The human role in changing river channels. Geomorphology 79 (3), 172–191.
- Hammersmark, C.T., Rains, M.C., Wickland, A.C., Mount, J.F., 2009. Vegetation and water-table relationships in a hydrologically restored riparian meadow. Wetlands 29 (3), 785–797.
- Hatch, C.E., Fisher, A.T., Ruehl, C.R., Stemler, G., 2010. Spatial and temporal variations in streambed hydraulic conductivity quantified with time-series thermal methods. J. Hydrol. 389 (3), 276–288.
- Haycock, N.E., Pinay, G., 1993. Groundwater nitrate dynamics in grass and poplar vegetated riparian buffer strips during the winter. J. Environ. Qual. 22 (2), 273–278.
- Hill, A.R., Duval, T.P., 2009. Beaver dams along an agricultural stream in southern Ontario, Canada: their impact on riparian zone hydrology and nitrogen chemistry. Hydrol. Process. Int. J. 23 (9), 1324–1336.
- Hodgson, R., 2008. Groundwater and Flooding in Oxford: Examining Key Soil Processes. MSc Dissertation. University of Reading.
- Hoehn, E., 2002. Hydrogeological issues of riverbank filtration–a review. Riverbank filtration: understanding contaminant biogeochemistry and pathogen removal, edited by: Ray, C., NATO Sci. Ser., IV. Earth Environ. Sci 14, 17–42.
- Hucks Sawyer, A., Bayani Cardenas, M., Bomar, A., Mackey, M., 2009. Impact of dam operations on hyporheic exchange in the riparian zone of a regulated river. Hydrol. Process. 23 (15), 2129–2137.
- Irvine, D.J., Brunner, P., Franssen, H.J.H., Simmons, C.T., 2012. Heterogeneous or homogeneous? Implications of simplifying heterogeneous streambeds in models of losing streams. J. Hydrol. 424, 16–23.
- Jurado, A., Borges, A.V., Brouyère, S., 2017. Dynamics and emissions of N2O in groundwater: a review. Sci. Total Environ. 584, 207–218.
- Kellogg, D.Q., Gold, A.J., Groffman, P.M., Addy, K., Stolt, M.H., Blazejewski, G., 2005. In situ ground water denitrification in stratified, permeable soils underlying riparian wetlands. J. Environ. Qual. 34, 524–533.
- Krause, S., Bronstert, A., Zehe, E., 2007. Groundwater–surface water interactions in a North German lowland floodplain–implications for the river discharge dynamics and riparian water balance. J. Hydrol. 347 (3), 404–417.
- Krause, S., Jacobs, J., Voss, A., Bronstert, A., Zehe, E., 2008. Assessing the impact of changes in landuse and management practices on the diffuse pollution and retention of nitrate in a riparian floodplain. Sci. Total Environ. 389 (1), 149–164.
- Leyer, I., 2005. Predicting plant species' responses to river regulation: the role of water level fluctuations. J. Appl. Ecol. 42 (2), 239–250.
- Lee, H., Koo, M.H., Kim, K., Kim, Y., 2015. Spatio-temporal variations in stream-aquifer interactions following construction of weirs in Korea. Groundwater 54 (3), 448–458.
- Loheide, S.P., Gorelick, S.M., 2007. Riparian hydroecology: a coupled model of the observed interactions between groundwater flow and meadow vegetation patterning. Water Resour. Res. 43 (7).

Macdonald, D.M.J., Dixon, A., Newell, A.J., Hallaways, A., 2012. Groundwater flooding

within an urbanised flood plain. J. Flood Risk Manage. 5 (1), 68-80.

- Macklin, M.G., Jones, A.F., Lewin, J., 2010. River response to rapid Holocene environmental change: evidence and explanation in British catchments. Quat. Sci. Rev. 29 (13), 1555–1576.
- Mansour, M.M., Hughes, A.G., 2004. User manual for the distributed recharge model ZOODRM. British Geological Survey Internal Report, IR/04/150, 61.
- Marsh, T.J., Hannaford, J., 2008. UK Hydrometric Register. Hydrological data UK series. Centre for Ecology & Hydrology.
- Matula, S., Mekonnen, G.B., Báťková, K., Nešetřil, K., 2014. Simulations of groundwatersurface water interaction and particle movement due to the effect of weir construction in the sub-watershed of the river Labe in the town of Děčín. Environ. Monit. Assess. 186 (11), 7755–7770.
- McClain, M.E., Boyer, E.W., Dent, C.L., Gergel, S.E., Grimm, N.B., Groffman, P.M., Hart, S.C., Harvey, J.W., Johnston, C.A., Mayorga, E., McDowell, W.H., 2003. Biogeochemical hot spots and hot moments at the interface of terrestrial and aquatic ecosystems. Ecosystems 6 (4), 301–312.
- Naganna, S.R., Deka, P.C., Ch, S., Hansen, W.F., 2017. Factors influencing streambed hydraulic conductivity and their implications on stream-aquifer interaction: a conceptual review. Environ. Sci. Pollut. Res. 24 (32), 24765–24789.
- Neal, C., Jarvie, H.P., Neal, M., Hill, L., Wickham, H., 2006. Nitrate concentrations in river waters of the upper Thames and its tributaries. Sci. Total Environ. 365 (1–3), 15–32.
- Newell, A.J., 2008. Morphology and Quaternary geology of the Thames floodplain around Oxford. British Geological Survey Open Report, OR/08/030, p. 30.
- Pinay, G., Black, V.J., Planty-Tabacchi, A.M., Gumiero, B., Décamps, H., 2000. Geomorphic control of denitrification in large river floodplain soils. Biogeochemistry 50, 163–182.
- Pinay, G., Ruffinoni, C., Wondzell, S., Gazelle, F., 1998. Change in groundwater nitrate concentration in a large river floodplain: denitrification, uptake, or mixing? J. North Am. Benthol. Soc. 17 (2), 179–189.
- Punalekar, S., Verhoef, A., Tatarenko, I.V., van der Tol, C., Macdonald, D.M.J., Marchant, B., Gerard, F., White, K., Gowing, D., 2016. Characterization of a highly biodiverse floodplain meadow using hyperspectral remote sensing within a plant functional trait framework. Rem. Sens. 8 (2), 112.
- Rivett, M.O., Buss, S.R., Morgan, P., Smith, J.W., Bemment, C.D., 2008. Nitrate attenuation in groundwater: a review of biogeochemical controlling processes. Water Res. 42 (16), 4215–4232.
- Silvertown, J., Araya, Y., Gowing, D.J.G., 2015. Hydrological niches in terrestrial plant communities: a review. J. Ecol. 103 (1), 93–108.
- Sophocleous, M., 2002. Interactions between groundwater and surface water: the state of the science. Hydrogeol. J. 10 (1), 52–67.
- Thacker, F.S., 1968. The Thames Highway: Volume II Locks and Weirs. David & Charles. Vidon, P., Allan, C., Burns, D., Duval, T.P., Gurwick, N., Inamdar, S., Lowrance, R., Okay, J., Scott, D., Sebestyen, S., 2010. Hot spots and hot moments in riparian zones: potential for improved water quality management. JAWRA J. Am. Water Resour. Assoc. 46, 278–298.
- Wheeler, B.D., Gowing, D.J.G., Shaw, S.C., Mountford, J.O., Money, R.P., 2004. Ecohydrological Guidelines for Lowland Wetland Plant Communities. Environment Agency (Anglian Region).
- Younger, P.L., Mackay, R., Connorton, B.J., 1993. Streambed sediment as a barrier to groundwater pollution: insights from fieldwork and modelling in the River Thames Basin. Water Environ. J. 7 (6), 577–585.

SUPPLEMENTARY INFORMATION

Site	water level percentiles				
	5 th	25 th	50 th	75 th	95 th
S1	53.81	53.86	53.90	53.97	54.21
S2	54.37	54.45	54.58	54.77	55.32
S3	55.04	55.05	55.10	55.25	55.78
S4	56.20	56.29	56.38	56.48	56.86
L2 d/s	53.80	53.83	53.86	53.90	54.12
L2 u/s	54.68	54.72	54.74	54.77	54.82
L3 d/s	54.68	54.72	54.76	54.87	55.56
L3 u/s	56.45	56.48	56.50	56.52	56.56
T1	56.48	56.52	56.56	56.72	57.29
L4 d/s	56.52	56.57	56.65	56.95	57.53
L4 u/s	58.08	58.12	58.15	58.19	58.26
T2	58.07	58.13	58.17	58.24	58.42
L5 d/s	58.10	58.16	58.20	58.30	58.73
L5 u/s	58.87	58.91	58.96	59.01	59.09
Т3	58.83	58.88	58.92	58.99	59.17
L6 d/s	58.90	58.96	59.03	59.22	59.86
L6 u/s	59.72	59.76	59.80	59.84	59.99

Table S1Percentile water levels for sites on the Seacourt Stream and River Thames (see Figure 2
for site locations), for 1 April 2009 to 31 March 2013. Note, d/s and u/s refer to
downstream and upstream, respectively, water levels for locks.

Deverenter	Cito	No.	percentiles				
rarameter Sile		measurements	5 th	25 th	50 th	75 th	95 th
Depth to	T1d	1826	0.20	0.82	1.01	1.20	1.32
groundwater	T1a	1826	-0.48	0.01	0.24	0.45	0.65
	WT	20	0.01	0.42	0.90	3.3	7.9
NO	SW	11	18	21	25	27	30
NU ₃	LF	105	0.01	0.01	0.04	0.09	0.53
	FP	17	0.01	0.01	0.03	0.06	0.12
	WT	20	1.8	1.9	2.3	2.7	2.9
NDOC	SW	11	2.4	2.6	2.8	3.4	3.8
NPOC	LF	103	6.3	7.1	8.0	9.1	15
	FP	17	2.1	2.4	2.5	2.7	3.3
DO	WT	30	0.10	0.20	0.40	0.90	3.6
	SW	11	8.1	9.2	9.5	10	12
	LF	115	0.00	0.28	0.47	0.72	3.5
	FP	24	0.00	0.20	0.32	0.41	0.78
	WT	43	266	294	315	328	351
HCO₃	SW	14	178	222	251	265	271
	LF	121	583	724	823	890	969
	FP	23	298	371	393	404	433
	WT	26	0.00	0.04	0.08	0.19	0.32
	SW	20	0.02	0.05	0.12	0.25	0.98
11174	LF	93	0.32	11	40	52	59
	FP	17	0.04	0.08	0.10	0.12	0.63

Table S2 Percentiles for sites box plots of depth to groundwater and water chemistry for a series of monitoring sites shown in Figure 6 (units: depth to groundwater, m; chemical parameters, mg/l). Sample sites are categorised as: groundwater west of the River Thames (WT); River Thames surface water (SW); groundwater east of the Thames in the floodplain under the influence of the Burgess Field landfill (LF); and groundwater east of the Thames in the other areas of the floodplain (FP).

Chapter 3: Nitrogen sources, transport and processing in peri-urban floodplains

The work presented in this chapter has been published in the following journal paper: Gooddy, D.C., Macdonald, D.M.J., Lapworth, D.J., Bennett, S.A. & Griffiths, K.J. (2014). Nitrogen sources, transport and processing in peri-urban floodplains. *Science of the Total Environment*, 494, 28-38.

Statement of contribution, as David Macdonald is not the lead author:

My overall percentage contribution to this paper was 35% and my specific contribution was to:

- devise the research question and design the methodology, with Gooddy and Lapworth;
- carry out the GIS analysis to identify how representative the study area is of the Thames catchment;
- co-design and supervise the installation of the borehole monitoring network;
- undertake elements of the groundwater sampling, as part of a team of five fieldworkers;
- coordinate the measurement of particle size distribution and calculate hydraulic conductivities;
- undertake the flux calculations;
- along with Gooddy and Lapworth, interpret the results; and
- write substantial sections of the methodology, results and discussion and produce 3 of the 6 figures.

Contents lists available at ScienceDirect





journal homepage: www.elsevier.com/locate/scitotenv



Nitrogen sources, transport and processing in peri-urban floodplains



D.C. Gooddy ^{a,*}, D.M.J. Macdonald ^a, D.J. Lapworth ^a, S.A. Bennett ^b, K.J. Griffiths ^a

^a British Geological Survey, Maclean Building, Wallingford, Oxfordshire OX10 8BB, UK

^b British Geological Survey, Keyworth, Nottingham NG12 5GG, UK

HIGHLIGHTS

GRAPHICAL ABSTRACT

- Peri-urban floodplains have been found to be populated with legacy landfills.
 Very high concentrations of nitrogen as
- ammonium are found.
- Isotopic methods identify legacy landfill as the ammonium source.
- Ammonium is not attenuated and is reaching the river.
- It is estimated that 27.5 tonnes of ammonium may be delivered to the river annually.



ARTICLE INFO

Article history: Received 16 May 2014 Received in revised form 27 June 2014 Accepted 27 June 2014 Available online 13 July 2014

Editor: Damia Barcelo

Keywords: Floodplain hydrogeology Groundwater surface water interaction Urban groundwater Nitrogen isotopes Nutrient fluxes

ABSTRACT

Peri-urban floodplains are an important interface between developed land and the aquatic environment and may act as a source or sink for contaminants moving from urban areas towards surface water courses. With increasing pressure from urban development the functioning of floodplains is coming under greater scrutiny. A number of peri-urban sites have been found to be populated with legacy landfills which could potentially cause pollution of adjacent river bodies. Here, a peri-urban floodplain adjoining the city of Oxford, UK, with the River Thames has been investigated over a period of three years through repeated sampling of groundwaters from existing and specially constructed piezometers. A nearby landfill has been found to have imprinted a strong signal on the groundwater with particularly high concentrations of ammonium and generally low concentrations of nitrate and dissolved oxygen. An intensive study of nitrogen dynamics through the use of N-species chemistry, nitrogen isotopes and dissolved nitrous oxide reveals that there is little or no denitrification in the majority of the main landfill plume, and neither is the ammonium significantly retarded by sorption to the aquifer sediments. A simple model has determined the flux of total nitrogen and ammonium from the landfill, through the floodplain and into the river. Over an 8 km reach of the river, which has a number of other legacy landfills, it is estimated that 27.5 tonnes of ammonium may be delivered to the river annually. Although this is a relatively small contribution to the total river nitrogen, it may represent up to 15% of the ammonium loading at the study site and over the length of the reach could increase in-stream concentrations by nearly 40%. Catchment management plans that encompass floodplains in the peri-urban environment need to take into account the likely risk to groundwater and surface water quality that these environments pose.

Crown Copyright © 2014 Published by Elsevier B.V. All rights reserved.

* Corresponding author. E-mail address: dcg@bgs.ac.uk (D.C. Gooddy).

http://dx.doi.org/10.1016/j.scitotenv.2014.06.123 0048-9697/Crown Copyright © 2014 Published by Elsevier B.V. All rights reserved.

1. Introduction

In recent decades anthropogenic inputs of N, for food production by intensive agriculture and urbanisation, have caused increases in macronutrient fluxes, and have led to widespread N pollution of aquatic systems (Foster et al., 1982; Burt et al., 2011; Whitehead and Crossman, 2012; Lapworth et al., 2013). This is a global issue with implications for food production and security, water quality and land management/ planning (Galloway, 1999; Galloway et al., 2004). In Europe and North America there have been a series of policies and accompanying regulations which have focused on reducing point and diffuse N and P pollution and associated problems of eutrophication and ecological degradation of freshwaters. For example, the European Water Framework Directive (WFD 2000) has the aim for fresh and marine waters to reach 'good ecological status' by 2015.

The zone of transition from rural to urban land-use is often referred to as the peri-urban area. Un-lined waste sites or landfills are a common source of N and C inputs to surface and groundwaters in peri-urban floodplains (Wakida and Lerner, 2005; Corniello et al., 2007). In Europe and North America there is a legacy of historic pollution from these sites (e.g. Heaton et al., 2005; Lorah et al., 2009). As such, landfill leachate plumes may contain high dissolved organic C (DOC) and NH_4^+ concentrations as well as ferrous iron, chloride and bicarbonate relative to natural floodplain conditions (Lorah et al., 2009). These commonly develop a series of distinct redox zones (Lyngkild and Christensen, 1992). In such systems, dilution, sorption and denitrification may be effective attenuation processes within the shallow groundwater system. The attenuation of N pollution from landfills, as well as other sources within peri-urban settings, requires fluctuations in redox conditions. These may be expected in floodplain settings due to rapid changes in water levels and episodes of surface water recharge and inundation.

Nitrogen in the form of ammonium (NH_4^+) is present naturally in groundwater as a result of anaerobic degradation of organic matter, and in the form of nitrate (NO_3^-) from the microbial oxidation of NH_4^+ . However, NH_4^+ and NO_3^- also occur in groundwater from anthropogenic sources. High concentrations of NH_4^+ (10–1000 mg/L) have been found in groundwaters impacted by landfill leachates and as a result of domestic and agricultural waste water disposal practices (Gooddy et al., 1998; Lawrence et al., 2000; Christensen et al., 2001; Heaton et al., 2005). High concentrations of NO_3^- are typically association with diffuse agricultural pollution from fertilisers (Oakes et al., 1981; Addiscott et al., 1991) although oxidation of anthropogenic ammonium sources also causes high NO_3^- concentrations (Gooddy et al., 2002).

Ammonium transport in the subsurface may be retarded by sorption (Ceazan et al., 1989; Buss et al., 2004). Both NH_4^+ and $NO_3^$ can be attenuated through microbially-induced transformations (DeSimone and Howes, 1998). Ammonium oxidation commonly occurs in conjunction with oxygen reduction and is termed nitrification. This results in the production of nitrite (NO_2^-) followed by NO_3^- .

$$NH_{4}^{+} + 1.5O_{2} \rightarrow NO_{2}^{-} + 2H^{+} + H_{2}O$$
(1)

$$NO_2^- + 0.5O_2 \rightarrow NO_3^- \tag{2}$$

In addition to degrading groundwater quality, NO_3^- and NH_4^+ can both be substantial sources of N in surface waters receiving groundwater (Jackson et al., 2007). Therefore it is desirable that this nitrogen is completely removed from the aquatic system. This requires bacterially mediated denitrification to convert the nitrate through intermediate stages to nitrous oxide and ultimately to nitrogen gas (Eq. (3)), which is a process requiring anaerobic conditions. This process is controlled by the availability of soluble carbon, redox status, pH as well as soil/groundwater residence times and hydrology (Haycock and Burt, 1993; Thomas et al., 1994; Burt et al., 1999; Gooddy et al., 2002).

$$NO_3^- \rightarrow NO_2^- \rightarrow NO + N_2O \rightarrow N_2(g)$$
 (3)

Therefore attenuation of N pollution from organic wastes requires alternating redox conditions, from oxidising to reducing, such as generated during water table fluctuations. Alternatively, through the anammox process, NH_4^+ can be oxidised anaerobically to nitrogen gas through the reduction of (NO_2^- derived from NO_3^- (see Eq. (3) above).

$$NH_4^+ + NO_2^- \rightarrow N_2 + 2H_2O \tag{4}$$

Despite the environmental importance of NO_3^- and NH_4^+ , there are few studies documenting their transport and reaction processes in aquifers (Heaton et al., 2005). Isotopic fractionation studies can provide an excellent tool for understanding N transport and speciation (Wassenaar, 1995; Böhlke et al., 2006). Isotopic fractionations have been reported for NH⁺₄ sorption to clays (Karamanos and Rennie, 1978), with the remaining NH_4^+ in solution relatively depleted in ¹⁵N. By contrast, nitrification results in a substantial increase in ¹⁵N for the remaining NH_4^+ (Delwiche and Stevn, 1970). Stable isotope ratios in NO_3^- have often been used to distinguish various sources of NO_3^- in groundwater, such as synthetic fertilisers and animal wastes (Gormly and Spalding, 1979; Flipse and Bonner, 1985). Denitrification causes an isotopic enrichment in the remaining nitrate (Mariotti et al., 1988). A disadvantage of the single isotope approach to denitrification studies is that process such as ammonia volatilisation can also lead to enrichment of ¹⁵N in the residual NH₄⁺ source material and in the NO₃⁻ produced during nitrification. However, by combining the $\delta^{18}\text{O}$ of the NO₃, a more reliable indicator of the denitrification process is achieved (Böttcher et al., 1990; Wassenaar, 1995; Kendall, 1998; Fukada et al., 2004). The opposing isotope fractionation effects make it possible to distinguish between sorption, nitrification and denitrification as major processes affecting N distribution in a field setting through evaluating variations in concentration and isotopic composition.

Riparian floodplains provide an important interface between terrestrial and aquatic systems (Harms and Grimm, 2008). Floodplain aquifers can be important sources of drinking water and sustain baseflow in surface waters, with important ecological implications (Sophocleous, 2002; Murray-Hudson et al., 2006). However, globally, floodplains are increasingly being encroached and developed due to anthropogenic pressures including urbanisation and intensive agriculture (Tockner and Stanford, 2002; Pinter, 2005; Werritty, 2006). In many parts of Europe, and elsewhere, there is a historical legacy of change in land use and associated historical pollution loading to floodplain groundwaters and surface waters (Burt et al., 2011; Stuart and Lapworth, 2011). Peri-urban floodplains are therefore complex, in terms of spatial heterogeneity in land use and topography, and temporal variability in recharge and redox processes (Burt et al., 2002; Burt and Pinay, 2005; Macdonald et al., 2012a; in press).

Due to changing redox conditions, floodplains and riparian settings are considered hot-spots for nutrient attenuation (Devito et al., 1999; McClain et al., 2003; Harms and Grimm, 2008). Groundwater levels within floodplain environments are typically shallow and responsive to recharge events. Vertical and spatial soil moisture conditions are also highly variable, and together these have important implication for attenuation of oxidised N (Burt et al., 1999). Periods of inundation may stimulate denitrification due to the mobilisation of C pools at shallow depths; during periods of low water tables denitrification may be restricted due to reduced C pools, low soil moisture and more oxidising conditions.

The important ecosystem services provided by floodplains and their general proximity to potentially damaging agricultural and urban nutrient sources means there is an imperative to understand nutrient processes, fluxes and attenuation mechanisms within. Due to the complex hydrogeology estimating the residence times of groundwaters and legacy nitrogen pollutants is challenging. There is therefore uncertainty as to the timescale for changes in management practices at the surface, reducing baseflow nutrient fluxes to surface waters (Wang et al., 2012; Stuart et al., 2011; Jackson et al., 2008).

This paper examines the temporal and spatial variations in N species within a typical peri-urban setting in the floodplain of the River Thames, UK. The peri-urban hydrological setting, legacy of pollution, and dependent terrestrial ecosystems means that the study is relevant across many industrialised and rapidly industrialising regions. The aim is to understand the water movement and redox status across the case study peri-urban floodplain and how this influences nutrient fluxes to the adjoining river. The objectives are to use geochemical indicators together with δ^{15} N of NH₄⁺ and δ^{15} N and δ^{18} O of NO₃⁻ to: 1) evaluate the origin of the nitrogen in groundwater; 2) ascertain whether sorption, nitrification or denitrification was occurring to attenuate the nutrient loading; and 3) estimate the amount of nitrogen that is delivered to the river system.

2. Materials and methods

2.1. Study area

The study focuses on a section of the floodplain of the River Thames in the vicinity of the city of Oxford in the southern UK (Fig. 1). The study area, known as Port Meadow, is to the north-west of the historic city centre. Port Meadow is an area of communal grazing and is regularly flooded by the River Thames, which flows along its western border. Due to areas of persistent groundwater flooding, the meadow supports a large number of waterfowl. Port Meadow is part of the larger Oxford Meadows Special Area of Conservation (SAC). The study area (Fig. 2a) also includes: the Burgess Field Nature Reserve, a former waste dump located to the east of Port Meadow; various tracts of land to the east, sited on the current floodplain, which were formerly the location of industrial landfills; the urbanised higher ground to the east of the floodplain; and the agricultural areas to the west of Port Meadow and the River Thames. The study area is bounded to the north and the south by urban developments.

The River Thames flows in a southerly direction through the city of Oxford (Fig. 1). The mean annual flow of the Thames upstream of the study area is 18.48 m³s⁻¹ (Marsh and Hannaford, 2008). The baseflow index (the ratio of long-term baseflow to total flow) for the river at this location is 0.67, reflecting the influence of influent groundwater, sourced from the limestone aquifers located in the headwaters, and the extensive floodplain gravel aquifers. The River Thames floodplain is approximately 2 km wide in the Port Meadow area, with a low gradient of less than 0.1° (0.03%). Groundwater flow is predominantly north-east to south-west; Fig. 2a includes groundwater level contours for a low-flow period. Normally groundwater flows into the River Thames from the north-east along the full reach within the case study area. Water management structures on the Thames a few hundred metres downstream of Port Meadow are a control on the hydrology of the area, by maintaining the river at relatively high levels during low-flow periods (Macdonald et al, 2012a). In the furthest downstream reach of the River Thames in the study area this can cause a reversal in hydraulic gradient for a few weeks of the year, allowing eastward flow from the



Fig. 1. The River Thames floodplain in the vicinity of the city of Oxford.

BGS® NERC 2014. Contains Ordnance Survey data® Crown copyright and database rights 2014. Licensed landfill information was provided by the Environment Agency, UK.



Fig. 2. a) Study area with monitoring network and groundwater level contours based on measurements made between 3 and 7 October 2011. Note, symbols for sites PTM26 to PTM31 indicate the centres of the piezometer nests. b) Simplified geological cross-section showing major topographic features and direction of landfill plume. BGS© NERC 2014. Contains Ordnance Survey data© Crown copyright and database rights 2014. Licensed landfill information was provided by the Environment Agency, UK.

river into the floodplain deposits. The Seacourt Stream flows along the western edge of the floodplain. Flow in this stream comes in part from the River Thames to the north, controlled by a fixed head weir. The stream is also a major groundwater discharge line within the floodplain (Fig. 2a); groundwater flows from the direction of Thames. The Oxford Canal runs along the eastern edge of the floodplain. There are also a series of drainage ditches across the floodplain.

The floodplain deposits within the Oxford Valley are made up of a shallow surface layer of fine-grained alluvium, underlain by highly permeable sands and gravels (Fig. 2b). The alluvium thickness can be up to 4 m but is typically around 1.5 m (Macdonald et al., 2012a). The sand and gravel thickness is mostly within the range of 2 to 6 m. The floodplain sediments are underlain by mudstones of the Upper Jurassic Oxford Clay Formation with some limited connection between the current floodplain sediments and second terrace gravels forming higher ground to the east of Port Meadow.

Port Meadow and the surrounding area have a long legacy of anthropogenic contamination with documented waste sites dating back to the late 1800s (Macdonald et al., 2012b). Burgess Field was used for domestic waste from 1937 to 1980. The waste here was dumped onto the floodplain surface with no underlying or lateral impermeable barriers apparent. The dump covers an area of 0.34 km², and has an average height of ~4 m above the natural ground level. The categories of waste reported to have been dumped there are: inert materials (soil, brick, concrete, glass, clay, sand etc.); semi-inert materials (wood, paper, cardboard, plasterboard, plastic, etc.); biodegradable wastes (food, sewage sludge, household, garden etc.); difficult wastes (e.g. tyres, sludges); and special wastes (hazardous chemical wastes, asbestos). Since closing, the waste dump has been grassed and planted with trees and shrubs. To the south of Burgess Field (Fig. 2a) there are a number of other areas which received domestic, building and industrial wastes during the late 19th and early 20th centuries.

The pollutant load to the floodplain of the River Thames in Oxford is thought to be typical of a substantial proportion of the floodplain within the overall Thames catchment. To assess the proportion of the Thames floodplain that can be classed as peri-urban, within a GIS a 1 km buffer was delineated around the large urban areas defined within the Ordnance Survey Strategi dataset. The floodplain within the Thames catchment was approximated using the BGS Geological Indicators of Flooding dataset (Booth and Linley, 2010) and the interception between this and the peri-urban approximation was calculated (Fig. 3). The area of floodplain is estimated as 1619 km²; the area of peri-urban floodplain is 720 km². This analysis indicates that significant areas within the Thames catchment can be classified as peri-urban and that insights from the Oxford case study will be have wider application.

2.2. Sampling

The sampling programme included 27 groundwater and 1 surface water sites across the study area (Fig. 2a). The sites include a range of land uses and were located up flow-gradient of the landfill and floodplain, down flow-gradient of the landfill and three sites located in the landfill. The existence of the unlined Burgess Field landfill was anticipated to exert an influence on the chemistry of the area so two new transects running perpendicular from the landfill towards the River Thames were installed (PTM26-28, PTM29-31). Each transect contains three nests of four piezometers, installed to target different depth horizons and sample the landfill plume (Fig. 2b). The piezometers within the nest were drilled separately, located within an area of approximately 4 m². The depth horizons approximately correspond to the soil, the alluvium, the top of the gravels and towards the base of the gravels. The piezometers within a nest are identified by letters A (shallowest) to D (deepest). The use of transects of nested piezometers with discrete screened intervals provides a good constraint on the vertical and lateral changes in hydrogeochemistry. Additional details of the sample sites included in this study are given in Macdonald et al. (2012b). To assess the impact of seasonal fluctuations in the groundwater level on redox conditions and nutrient hydrogeochemistry, sampling was conducted at approximately quarterly intervals between May 2010 and May 2012, with a further sampling round in August 2013; in total 12 rounds were undertaken. The piezometer headworks were completed in a 100 mm diameter chamber below ground level, for aesthetic, security and safety reasons, and capped with a metal plate below a grass sod so that it can be located with a metal detector. During periods of flooding a short length of 150 mm diameter pipe is driven into the

ground around the chamber, acting as a cofferdam and allowing water to be pumped away from the headworks before sampling. The design and use of the piezometers is shown in Fig. 4.

A minimum of three borehole volumes were purged from each groundwater sampling site, and samples were not collected until stable readings for pH, specific electrical conductivity (SEC) and dissolved oxygen (DO) were obtained. Samples for chloride and nitrogen species were filtered at 0.45 μ m and collected in 30 mL plastic bottles. Samples for ¹⁵N analysis of NH₄ or NO₃ and ¹⁸O-NO₃ were collected on August 4th 2013 and filtered into plastic 1 L bottles; the samples for NH₄ analysis being acidified in the field with HCl to pH 2–4. These samples were frozen and to be defrosted just before analysis. Samples for dissolved gases were collected in sealed steel ampoules at the same time as the samples for isotopic analysis as well as at selected sites on two other occasions in July 2011 and October 2011.

2.3. Chemical analyses

The samples were collected and analysed for Cl, NO₃ and NO₂ using ion chromatography (IC), and NH₄ by flow colorimetry. Field data including bicarbonate, pH, temperature, specific electrical conductance (SEC), and dissolved oxygen (DO) were all determined at site, and a flow through cell was used for unstable field parameters to obtain representative in-situ values. Dissolved nitrous oxide was measured at the British Geological Survey's Environmental Tracer Laboratories (Wallingford, UK) by Gas Chromatography with an electron capture detector (ECD) and a 3m PorapakTM Q column held isothermally at 40 °C. N₂O gas standards at 10 ppm and 100 ppm were used for calibration (Bedfont Scientific Limited, Rochester, England) and gave a linear response with the ECD. Analytical precision (1 SD) was typically <1% (Gooddy et al., 2002).

Isotope preparation and analysis was carried out at the NERC Isotope Geosciences Laboratory (Keyworth, UK). Nitrate was separated on anion resins and prepared as silver nitrate using the method of Silva et al. (2000). Ammonium was converted to NH_4SO_4 on acidified quartz filter papers using a static ammonia diffusion technique (Sebilo et al., 2004). The filters were combusted to produce N_2 for $^{15}N/^{14}N$ analysis. Ammonium and nitrate $^{15}N/^{14}N$ ratios were analysed by combustion in a Flash EA on-line to a Delta Plus XL mass spectrometer (ThermoFinnigan,

Fig. 3. Peri-urban floodplains in the catchment of the River Thames. BGS© NERC 2014. Contains Ordnance Survey data© Crown copyright and database rights 2014.

Fig. 4. Groundwater sampling during flood events: a) exposed well-head, b) sampling equipment, and c) piezometer design.

Bremen, Germany). Isotope ratios were calculated as δ^{15} N values versus air (atmospheric N₂) by comparison with standards calibrated against IAEA N-1 and N-2 assuming these had values of + 0.4% and + 20.3%, respectively. Analytical precision (1 SD) was typically <0.8\%, from repeat analysis of a sample. 18 O/ 16 O ratios of nitrate were analysed by thermal conversion to CO gas at 1400 °C in a TC–EA on-line to a Delta Plus XL mass spectrometer (ThermoFinnigan, Bremen, Germany). Isotope ratios were calculated as δ^{18} O values versus VSMOW by comparison with IAEA-NO₃ assuming it had a value of + 25.6%. Analytical precision (1 SD) was typically <1.2‰.

2.4. Flux calculations

Darcy's Law was used to calculate the flux of a conservative tracer (in this case chloride) from the floodplain sediments into the River Thames along the reach parallel to the Burgess Field landfill. Parameters used to calculate the flux are given in Table 1. The hydraulic conductivities (K) of the alluvium and the sands and gravels were estimated based on particle-size analyses of material obtained during the drilling of piezometers in Port Meadow, using the method of Boonstra and de Ridder (1981); the method relates K to the specific surface area of the sediment. Although sufficiently permeable to allow groundwater recharge to occur, the alluvium was estimated as having K values three orders of magnitude smaller than the gravel K; the alluvium was therefore ignored in the calculation of lateral flux. Given the accuracy of the method, the average gravel K was rounded to the nearest 100 as was the range of potential K values (indicated in parentheses in Table 1).

Table 1

Parameters used to calculate flux from Burgess Field landfill to the River Thames through the floodplain sediments.

Parameters	Value
Gravel aquifer hydraulic conductivity (m/d)	400 (±200)
Groundwater river inflow cross-sectional area (m)	2.5
Groundwater level gradient $(-)$	0.0004
Landfill length, parallel to river (m)	1000
Groundwater inflow to river along affected reach (m ³ /d)	400 (±200)
Mean daily river flow (m ³ /d)	$1.60 imes 10^6$

The river bed K was assumed to be equal to the gravel K as the correlation between river level and groundwater level in PTM25, which is adjacent to the river, was very high ($r^2 = 0.99$).

Geological logs of boreholes in the vicinity of the River Thames show an average thickness of alluvium of 0.8 m and depth to the base of the gravel aquifer of 4.4 m; a river survey undertaken for the Environment Agency of England has the bed elevation at its deepest as 2.8 m below the floodplain ground level. Taking these into account, as well as a likely vertical groundwater flow component to the river, a cross-sectional area for groundwater inflow to the river from the gravel aquifer of 2.5 m was used. Groundwater and river level monitoring allowed the calculation of the head gradient towards the river; the direction of flow is approximately perpendicular to the river for the majority of the period being considered (Macdonald et al., 2012b; Fig. 2a). The gradient outside of periods of short-lived high river stage ranges from 0.0008 to -0.0003; a median value of 0.0004 was used in the flux calculation. The total daily inflow of groundwater to the river along the Burgess Field reach was compared to the mean daily river flow, in Table 1.

Combining the inflows with the concentrations of dissolved species measured in boreholes between the landfill and the River Thames enabled the flux of species into the river from the floodplain aquifer to be compared with the downstream flux in the river (see the Results section).

3. Results

Table 2 shows the range of results observed for selected determinants measured over the past three years. There is relatively little seasonal variation for any given site with the greatest temporal changes being seen when sampling has followed a recent heavy rainfall event (Macdonald et al., 2012b). Consequently, greatest temporal variation is observed in the shallow piezometers. Data for electrical conductivity shows this particularly well, with the majority of measurements made in August 2013 (square brackets) within 5% of the mean conductivity sampled over the study period. Mean pH values are generally just below 7 with the highest values (>7) found in the shallower (<1 m) piezometers and the two boreholes to the far south (PTM21 and OX14)

Table 2

Borehole construction details and on-site field measurements data for selected boreholes in the Port Meadow network.

Site	Borehole depth ^a	Geology at screen	SEC	рН	DO
	m		μS/cm	-	mg/L
GBH3	5.09	See note ^b	2027 (1944–2110)	6.77 (6.68-6.86)	0.83 (0.46-1.20)
GBH5	5.60	See note ^b	1560 (1438-1681)	6.94 (6.68-7.19)	0.14 (<0.1-0.23)
GBH9	5.24	See note ^b	1463 (1403-1522)	6.86 (6.68-7.03)	0.38 (0.26-0.50)
OX14	2.72	Gravel	824 (743–977) [798]	7.43 (7.21–7.72) [7.34]	0.64 (0.19-2.04) [0.30]
PTM11	3.38	Gravel	772 (640-860) [755]	7.21 (6.93–7.99) [6.93]	0.40 (<0.1-0.79) [0.22]
PTM21	1.40	Gravel	1071 (914–1283) [1080]	7.19 (6.87–7.67) [7.67]	2.03 (0.66-4.81) [0.69]
PTM23	1.67	Gravel	1661 (957–1957) [1531]	6.99 (6.81-7.40) [6.89]	3.61 (0.45-10.6)
PTM24	0.99	Gravel	1075 (640–1250) [1180]	6.89 (6.65-7.18) [6.65]	1.24 (0.37-1.73)
PTM25	0.93	Gravel	1336 (691–1622) [1350]	6.92 (6.61-7.45) [6.61]	3.03 (0.45-10.6) [0.63]
PTM26B	0.83	Alluvium	1277 (447–1530)	6.91 (6.59-7.26)	0.64 (<0.1-1.24)
PTM26C	1.69	Gravel	1625 (1548-1846) [1566]	6.87 (6.64-7.16) [6.69]	0.43 (<0.1-0.95) [0.28]
PTM26D	2.90	Gravel	1812 (1619–19,850) [1705]	6.86 (6.64-7.17) [6.64]	0.56 (<0.1-1.33) [0.28]
PTM27B	0.76	Alluvium	1641 (1211-2060) [1821]	7.08 (6.75–7.44) [7.09]	0.62 (<0.1-1.14)
PTM27C	1.82	Gravel	1662 (1005-2090) [1878]	6.91 (6.57-7.48) [6.61]	0.52 (<0.1-0.85) [0.41]
PTM27D	2.94	Gravel	1963 (1854–2120) [1854]	6.87 (6.62-7.18) [6.65]	0.38 (<0.1-0.77) [0.25]
PTM28B	0.81	Alluvium	2263 (2010-2662)	6.91 (6.65-7.40)	1.28 (<0.1-3.6)
PTM28C	1.76	Gravel	1990 (1835–2160) [2010]	6.85 (6.64-7.12) [6.77]	0.89 (<0.1-1.71) [0.83]
PTM28D	3.64	Gravel	1895 (1755–2160) [1783]	6.83 (6.68-7.11) [6.68]	0.72 (<0.1-3.02) [0.39]
PTM29C	1.62	Gravel	1209 (1106-1303) [1190]	6.98 (6.73-7.25) [6.76]	0.38 (<0.1-0.63) [0.26]
PTM29D	3.74	Gravel	1733 (1075–1949) [1773]	6.90 (6.64-7.26) [6.64]	0.48 (<0.1-1.00) [0.21]
PTM30B	0.76	Alluvium	1168 (955–1420)	7.23 (7.04–7.36)	3.19 (0.81-7.92)
PTM30C	1.81	Gravel	1334 (688–1942) [1379]	7.00 (6.67–7.24) [6.67]	0.56 (<0.1-1.31) [0.47]
PTM30D	2.85	Gravel	1655 (911–1898) [1673]	6.93 (6.69-7.24) [6.69]	0.49 (<0.1-1.20) [0.46]
PTM31B	0.86	Alluvium	1969 (1577-2540)	7.19 (6.91–7.35)	1.36 (<0.1-2.93)
PTM31C	1.78	Gravel	1725 (1513–2050) [1590]	6.96 (6.72-7.25) [6.72]	0.56 (<0.1-1.35) [0.32]
PTM31D	3.87	Gravel	1745 (1140–1985) [1638]	6.99 (6.76-7.41) [6.76]	0.39 (<0.1-1.10) [0.39]
R. Thames	n/a	n/a	607 (450–714) [583]	8.19 (7.42–8.67) [7.42]	9.78 (7.80–12.7) [9.46]

Mean values that are presented with the range of concentrations found over the 12 sampling periods from May 2010 to August 2013 are shown in brackets. Measurements made in August 2013 are shown in square brackets.

^a All screen sections are over the bottom 0.5 m of the borehole, apart from the GBH boreholes, drilled in the 1980s, for which no screen information is available.

^b GBH boreholes are drilled into the gravel; it is expected that they are screened in the gravel and also the overlying alluvium and landfill material.

and to the far north (PTM11) of the landfill. Highest pH values are found in the River Thames. Dissolved oxygen mean values are typically <1 mg/L and often <0.1 mg/L in the deeper piezometers, in the shallower piezometers DO is up to 10 mg/L. The River Thames typically has DO concentrations at saturation for the atmospheric temperature in a given season.

Nitrate concentrations range from a high of ~30 mg/L in the river through below detection for the majority of the piezometers (Table 3).

Table 3

Nitrogen species and isotopic data for selected boreholes in the Port Meadow network.

Site	NO3-N	δ^{15} N-NO ₃	δ^{18} O-NO ₃	N ₂ O	NO ₂	NH₄-N	δ ¹⁵ N-NH₄
	mg/L	%	‰	μġ/L	μg/Ĺ	mg/L	%
GBH3	<0.01 (<0.01-5.6)			0.3	<10	22 (9.6-36) [14]	+ 12.5
GBH5	7.06 (2.55–16.2) [13.3]	+13.3	+6.2	27.9	<10	69 (9.6–78) [53]	+8.7
GBH9	16.5 (0.82-48) [0.82]	+8.7		19.9	<10 (<10-170)	42 (9.6–53) [39]	+7.0
OX14	0.12 (<0.01-0.48)			11.3 (<0.1-13.5)	<10 (<10-90)	3.3 (2.8-4.1) [3.0]	+8.9
PTM11	0.03 (<0.01-0.21)			<0.1 (<0.1-0.3)	<10	<0.2 (<0.2-2.2) [0.2]	+2.5
PTM21	0.03 (<0.0.1-0.35)			NA	<10	2.7 (1.58-6) [2.2]	+10.7
PTM23	0.12 (<0.01-1.63)			NA	<10 (<10-30)	11.5 (0.45-35.2) [25]	+8.7
PTM24	0.02 (<0.01-0.06)			NA	<10	3.3 (1.9–7.9) [5.0]	+10.5
PTM25	0.1 (<0.01-0.53)			0.1	<10 (<10-90)	0.43 (0.15-0.58) [0.4]	+19.2
PTM26B	0.09 (<0.01-0.38)			NA	<10	4.5 (1.2-24.2)	NA
PTM26C	0.06 (<0.01-0.32)			27 (<0.1-54)	<10	19 (13–22) [14]	+8.8
PTM26D	0.05 (<0.01-0.3)			1.75 (0.88-2.6)	<10 (<10-50)	27 (22–29) [18]	+8.3
PTM27B	0.05 (<0.01-2.8)			10.7 (0.5-17.3)	<10	7.7 (2.2–31)	NA
PTM27C	0.04 (<0.01-0.36)			1.8 (0.9-2.8)	<10 (<10-50)	33 (12–47) [28]	+8.3
PTM27D	0.03 (<0.01-0.09)			0.8 (<0.1-1.5)	<10	48 (2.5–50) [31]	+8.4
PTM28B	0.08 (<0.01-1.14)			NA	<10	17 (2.9–27)	NA
PTM28C	0.09 (<0.01-0.49) [0.49]	+3.7		1.1	<10	49 (43-99) [27]	+9.2
PTM28D	0.06 (<0.01-0.2)			3.5 (<0.1-39)	<10	56 (49-56) [25]	+6.9
PTM29C	0.04 (<0.01-0.17)			9.2 (2.5-15.9)	<10	21.3 (5-28) [28]	+8.0
PTM29D	0.03 (<0.01-0.14)			0.3 (<0.1-0.3)	<10	32 (5.8–37) [22]	+9.1
PTM30B	0.02 (<0.01-25.5)			1.12	<10 (<10-2800)	1.8 (0.45-15.9)	NA
PTM30C	0.05 (<0.01-0.15)			0.4	<10	14 (4–52) [41]	+7.8
PTM30D	0.05 (<0.01-0.51)			0.2 (0.1-0.3)	<10	58 (36-59) [36]	+9.1
PTM31B	0.07 (<0.01-0.69)			NA	<10	54 (1.2-60)	NA
PTM31C	0.08 (<0.01-0.27)			102 (13.7-191)	<10	63 (44–70) [44]	+8.7
PTM31D	0.06 (<0.01-0.14)			0.6 (0.2–1)	<10	48 (11-52) [39]	+9.1
R. Thames	21 (16.6-30) [18.4]	+9.92	+4.35	NA	70 (<10-140)	0.06 (<0.2-0.14)	NA

Mean values that are presented with the range of concentrations found over the 12 sampling periods from May 2010 to August 2013 are shown in brackets. Measurements made in August 2013 are shown in square brackets.

The main exceptions are three monitoring boreholes (GBH3, 5 and 9) completed within the Burgess Field waste site where NO₃-N concentrations range from <0.01 to 48 mg/L. Within the nested transect boreholes (PTM26–28 and PTM29–31), the shallow sites (B) have low (<2 mg/L) but uniformly higher mean NO₃-N concentrations compared to deeper sites. Reflected by these generally low nitrate concentrations is the relative paucity of data for nitrate isotopes, with just 2 samples yielding enough nitrate to enable measurement and one of these being the river. δ^{15} N-NO₃ values ranged from +10 to +13‰ and δ^{18} O-NO₃ ranged between +4 and +6‰. This was compounded by the high organic carbon concentrations within the plume. These high concentrations coincided with a low redox status and negligible dissolved oxygen which was observed year round for most sites.

Nitrite concentrations generally fall below the detection limit of 10 μ g/L although there are exceptions from the river Thames, the site at OX14 and the highest concentration occurring from one of the landfill monitoring boreholes. Highest concentration of N₂O, at nearly 200 μ g/L, also occurs in one of the boreholes adjacent to landfill. Other sites range from a near 'background' concentration of 0.2 μ g/L to ~54 μ g/L; the shallower transect boreholes are generally higher in concentration than the deeper transect boreholes (Table 3).

Ammonium concentrations show considerable variation from 0.05 to 99 mg/L NH₄-N with the highest concentration occurring in the land-fill boreholes and the lowest in the River Thames (Fig. 5). Concentrations are generally higher the closer the borehole is to the waste site and the deeper the borehole, with concentrations decreasing closer to the river. Concentrations from the waste site boreholes show considerable variation, with the highest concentration detected in the same site

that also had the greatest amounts of NO₃-N and N₂O. Similar to the NH₄-N data, the δ^{15} N-NH₄ values also show considerable variation from a minimum of 2.5% to a maximum of 19.2 of % (mean of 9.1%). Values in the main transect vary much less, with a mean of 8.5% and minima and maxima of 6.9% and 9.2%. The highest value occurs to the south of the main plume, whereas the lowest value occurs to the north. Values in the waste dump boreholes vary from ~7% to 12.5%.

Fig. 6 shows filled contour plots of mean N-NH₄ concentrations for the two transects within the floodplain along the flow gradient from the landfill site to the River Thames. There is a consistent trend of higher N-NH₄ concentrations with depth in the gravels, and the lowest concentrations are found in the two nests that are closest to the Thames. The highest concentrations are found in transect B, between 100 and 200 m from the edge of the landfill rather than in the landfill and at a depth greater than 55 mAOD.

The steps in estimating the percentage of the N and NH₄-N flux in the River Thames, due to that entering the river laterally from the floodplain sediments to the east, are set out in Table 3. The total N and NH₄-N groundwater influx is compared to the concentration within the River Thames by using a mean concentration in the river measured during the period of the study, and the long-term river flow upstream. Samples from the two borehole nests at the western end of the two transects (PTM26 and PTM29) provided an average deep gravel groundwater NH₄-N concentration (3.75 m borehole; 29.9 g/m³) and total NH₄-N concentration (average of 1.75 and 3.75 m boreholes; 21.6 g/m³). Estimated fluxes of NH₄-N in groundwater into the river along the Burgess Field reach are compared with the flux within the river; groundwater NH₄-N fluxes account for up to 15% (depending on the

Fig. 5. Study area with average NH₄-N concentrations in groundwater from the piezometers. Graduated symbols indicate spatial variability in NH₄-N concentrations. Note, symbols for piezometers within the nests are offset to enable the concentrations within each to be visible; piezometer C is positioned in the centre of the nest, D to the south-west and B, where measurements are available, to the north-east.

BGS® NERC 2014. Contains Ordnance Survey data® Crown copyright and database rights 2014. Licensed landfill information was provided by the Environment Agency, UK.

Fig. 6. Filled contour plot showing changes in average NH₄-N concentrations in groundwater across both transects in Port Meadow.

input concentration) of the NH_4 -N in the river. Nitrate in the groundwater is negligible; total N in the river water is the sum of NO_3 -N and NH_4 -N. Table 3 also compares the estimated total N flux of groundwater into the river with the flux within the river which is approximately two orders of magnitude smaller than that for NH_4 -N.

4. Discussion

The spatial variability in groundwater NH_4 -N raises questions of source (i.e. does all the ammonium come from the landfill?) and attenuation mechanism within the floodplain sediments. A cross plot of NH_4 -N and Cl for all available data during the study, with sites grouped by different hydrological zones within the peri-urban floodplain, is

Fig. 7. Cross-plot of NH₄-N vs Cl in the Port Meadow area combined over the study period. Sites are grouped as: GBH3, 5 and 9 – Landfill; OX14 and PTM11 – Up gradient; PTM26–PTM31 A&B – Shallow Plume; PTM26–PTM31 C&D – Deep Plume.

shown in Fig. 7. This shows the effect of dilution of the landfill contamination within the floodplain and the variability of NH_4 -N and Cl within each zone. The spread of data in Fig. 7 can be understood in terms of ternary mixing between, i) groundwater leaching from the waste site with high NH_4 -N concentrations, ii) local shallow recharge to the floodplain gravels and iii) the River Thames. Ammonium contamination in the shallow gravels is diluted relative to the deeper gravels, increasingly so down the flow gradient towards the River Thames. The samples with lower NH_4 -N concentrations within the landfill sites are associated with low water level conditions. The shallow floodplain groundwaters have elevated Cl concentrations compared to other end members; this is perhaps evidence of evaporative enrichment of ponded surface water on the floodplain in the summer prior to recharge.

A cross plot of δ^{15} N-NH₄ against NH₄⁴-N (Fig. 8) clearly shows how the majority of samples have δ^{15} N-NH₄ values that fall between ~7

Fig. 8. δ^{15} N values and concentrations of NH₄⁺-N in Port Meadow groundwaters. All unlabelled data points are from two main transects.

and 10‰ and are fairly independent of concentration. The two major outliers from this trend are PTM11 and PTM25. PTM11 is thought to be largely up-gradient of the main contaminant plume and it would appear this is borne out from the isotopic data. The isotopic ammonium delta value coupled with the much lower concentration of ammonium here is more similar to a soil delta value rather than any influence of the landfill (Heaton, 1986; Kendall, 1998). Similarly, PTM25 is in close proximity to the River Thames, and this very high value is probably more indicative of faecal waste (Heaton, 1986; Kendall, 1998) from two possible sources: treated waste water input to the River Thames upstream of Port Meadow which can move into the floodplain sediments due to the hydraulic gradient reversal that can occur in the area of PTM25 during summer months; and waterfowl that congregate close to this area where groundwater flooding within the floodplain persists longest.

The landfill boreholes show some of the most interesting N transformation chemistry. GBH5 contains high concentrations of both nitrate and ammonia as well as significant concentrations of N₂O. The presence of some measurable dissolved oxygen at times along with nitrate isotope values (δ^{15} N-NO₃ = 13.3‰ and δ^{18} O-NO₃ = 6.2‰) is highly indicative that some nitrification of ammonia has occurred followed by denitrification of the resultant nitrate, although the process has not moved to complete removal of nitrate. Similarly, lower concentrations of ammonium and some nitrate are also found at GBH9 and along with NO₂, albeit at low concentrations. This possibly suggests a lower rate of nitrification and no denitrification, since although NO₂ is an intermediate product in both process, the nitrate has to be formed before it can be removed. GBH3 has a much higher δ^{15} N-NH₄ value than the other two landfill monitoring boreholes, but it also has much lower NH₄⁺-N concentrations. This might indicate some nitrification followed by complete conversion to N₂ via denitrification. The absence of any intermediate products might suggest that the reaction is complete so removing nitrogen from the system in a region where the water table is known to fluctuate.

OX14 and PTM23 have very similar δ^{15} N-NH₄ values to the transect boreholes. Site OX14 has nitrification/denitrification intermediates which could explain a much lower ammonium concentration. PTM23 however also has a high ammonium concentration and it would appear that this site is impacted by the plume. The shallow boreholes at PTM21 and PTM24 have some of the lowest ammonium concentrations and relative to the other transect boreholes have slightly elevated δ^{15} N-NH₄ values which may be indicative of some nitrification/denitrification reactions. All of these sites have relatively high DO concentrations at some times of the year as well as higher pH values (>7) which are more conducive to denitrifying bacteria (Thomas et al., 1994). Again, the absence of any intermediate products suggests that the reaction is complete so removing nitrogen from the system in a region where the water table is known to fluctuate.

One might speculate that the relatively consistent δ^{15} N-NH₄ values observed could be a result of nitrification, which would increase the delta value, and sorption which would decrease the delta value. However, given that the oxygen availability is generally low and that the sorption and nitrification processes would have to occur in fairly equal amounts (independent of concentration – see Fig. 8) this does seem a somewhat unlikely scenario. Therefore, since there is no relationship between ammonium concentration and δ^{15} N-NH₄, sorption on to clays can most probably be ruled out as an attenuation mechanism. Away from the main plume, and on the direct edge of the landfill, there is strong evidence for both nitrification and denitrification. However, this appears limited and restricted to pockets of groundwater table fluctuation, and the majority of ammonium in the plume is not effectively retarded by sorption, with concentrations only tempered by dilution. Where the aquifer remains saturated all year round (particularly at piezometer depths C and D) the ammonium can be considered as conservative. For the floodplain to be a buffer zone for nitrogen, as would generally be the case in more rural environments (Burt et al., 1999), then it is clear that the site conditions prevent this from occurring since in the landfill plume, there is insufficient dissolved oxygen to facilitate nitrification at depths greater than 1 m.

The area of landfills/waste dumps on the River Thames floodplain in the Oxford area is significant (Fig. 1); 1.05 km² of the 15.87 km² of floodplain. Making the assumption that the influx of NH₄-N to the river via the gravel aquifer from the Burgess Field waste dump is representative of the remainder on the floodplain, behaving in a similarly conservative manner, it is possible to estimate the total influx due to all the landfills/waste dumps. This takes into account the volume of the landfill and the dilution where the landfill is not located by the river bank. The calculated flux along an approximately 8 km reach of the River Thames is estimated as 75 kg ammonium (NH₄) per day or over 27.5 tonnes per year. Assuming no additional in-stream processing, this influx would lead to a river concentration of close to 0.1 mg/L at the end of the reach compared with 0.06 mg/L at Port Meadow – or roughly 40% of total ammonia in the river coming from legacy waste dumps.

5. Conclusions

A combination of nitrogen isotopes, nitrogen speciation and dissolved nitrous oxide has been used to understand the sources of nitrogen in a complex and heterogeneous peri-urban floodplain. Based on other nitrogen isotope studies, the dominant source of nitrogen is in the form of ammonium and this has been shown to have originated from a former domestic landfill which continues to act as a source of elevated nitrogen into the environment. Despite some evidence for enitrification in areas with water table fluctuations near to the landfill or on the fringes of the landfill plume, the prevalence of year round reducing conditions in the deeper (>1 m) floodplain has resulted in the transport of ammonium, with minimal biogeochemical processing or sorption to sediments, directly into a major river. Although the calculated flux is not a large contribution to the overall nitrogen in the river, it does represent the addition of a significant proportion (~10% but up to 15%, Table 4) of the total ammonium as the river goes past each landfill. Collectively this contribution to overall ammonium concentrations could be very high. This is a hitherto unconsidered source of river pollution from the peri-urban fringe. Given the large number of urban developments along the edge of floodplains and the associated historical and at times ad hoc waste dumps, this is likely to be a scenario reflected in many other parts of the developed world. Catchment management

Table 4

Determination of ammonium and total nitrogen flux from landfill site to the river at Port Meadow.

	Deep sample	Average of shallow and deep
$ \begin{array}{l} \text{NH}_4\text{-N groundwater concentration (g/m^3)} \\ \text{NH}_4\text{-N influx to river (g/d)} \\ \text{NH}_4\text{-N river concentration (g/m^3)} \\ \text{NH}_4\text{-N flux within river (g/d)} \\ \text{N river concentration - NH}_4\text{-N + NO}_3\text{-N (g/m^3)} \\ \text{N flux within river (g/d)} \\ \text{NH}_4\text{-N groundwater influx as percentage of river flux (-)} \end{array} $	29.9 10000 (\pm 5000) 0.064 1.02 × 10 ⁵ 5.45 8.7 × 10 ⁶ 11.7 (\pm 4.9)	21.6 8600 (±4300) 8.5 (±4.2)
N groundwater influx as percentage of river flux $(-)$	0.14 (±0.07)	0.10 (±0.05)

plans that encompass floodplains in the peri-urban environment need to take into account the likely risk to groundwater and surface water quality from these legacy landfills.

Acknowledgments

This work greatly benefitted from the diligent field work of Debbie Allen and Peter Williams (BGS) who, along with the authors, undertook field sampling under a range of challenging conditions. The authors would like to thank James Sorensen (BGS) and five annonymous reviewers whose comments have helped to improve this manuscript. This paper is published with the permission of the Executive Director, British Geological Survey (NERC).

References

- Addiscott TM, Whitmore AP, Powlson DS. Farming, fertilizers and the nitrate problem. Wallingford: CAB International; 1991. p. 170.
- Böhlke JK, Smith RL, Miller DN. Ammonium transport and reaction in contaminated groundwater: application of isotope tracers and isotope fractionation studies. Water Resour Res 2006;42(W05411):1–19.
- Boonstra J, de Ridder NA. Numerical modelling of groundwater basins. ILRI publ, 29. Wageningen: International Institute for Land Reclamation and Improvement; 1981.
- Booth KA, Linley KA. Geological indicators of flooding: user guidance notes. British Geological Survey Openfile Report, OR/10/064; 2010 [16 pp.].
- Böttcher J, Strbel O, Voerkelius S, Schimidt HL. Using isotope fractionation of nitratenitrogen and nitrate-oxygen for evaluation of microbial denitrification in a sandy aquifer. J Hydrol 1990;114:413–24.
- Burt TP, Pinay G. Linking hydrology and biogeochemistry in complex landscapes. Prog Phys Geogr 2005;29(3):297–316.
- Burt TP, Matchett LS, Goulding KWT, Webster CP, Haycock NE. Denitrification in riparian buffer zones: the role of floodplain hydrology. Hydrol Process 1999;13(10):1451–63.
- Burt TP, Bates PD, Stewart MD, Claxton AJ, Anderson MG, Price DA. Water table fluctuations within the floodplain of the River Severn, England. J Hydrol 2002;262(1–4): 1–20.
- Burt TP, Howden NJK, Worrall F, Whelan MJ, Bieroza M. Nitrate in United Kingdom rivers: policy and its outcomes since 1970. Environ Sci Technol 2011;45(1):175–81.
- Buss SR, Herbert AW, Morgan P, Thornton SF, Smith JWN. A review of ammonium attenuation in soil and groundwater. Q J Eng Geol Hydrogeol 2004;37:347–59.
- Ceazan ML, Thurman EM, Smith RL. Retardation of ammonium and potassium transport through a contaminated sand and gravel aquifer: the role of cation exchange. Environ Sci Technol 1989;23:1402–8.
- Christensen TH, Kjeldsen P, Bjerg PL, Jensen DL, Christensen JB, Baun A, Albrechtsen HJ, Heron G. Biogeochemistry of landfill leachate plumes. Appl Geochem 2001;16: 659–718.
- Corniello A, Ducci D, Ruggieri G. Areal identification of groundwater nitrate contamination sources in periurban areas. J Soils Sediments 2007;7:159–66.
- Delwiche CC, Steyn PL. Nitrogen isotope fractionation in soils and microbial reactions. Environ Sci Technol 1970;4:929–35.
- DeSimone LA, Howes BL. Nitrogen transport and transformations in a shallow aquifer receiving wastewater discharge: a mass balance approach. Water Resour Res 1998;34: 271–85.
- Devito KJ, Fitzgerald D, Hill AR, Arevena R. Nitrate dynamics in relation to lithology and hydrologic flow path in a river riparian zone. J Environ Qual 1999;29:1075–84.
- Flipse Jr WJ, Bonner FT. Nitrogen-isotope ratios of nitrate in ground water under fertilized fields, Long Island, New York. Ground Water 1985;23:59–67.
- Foster SSD, Cripps AC, Smith-Carington A. Nitrate leaching to groundwater. Philos Trans R Soc Lond Ser B Biol Sci 1982;296(1082):477–89.
- Fukada T, Hiscock KM, Dennis PF. A dual-isotope approach to the nitrogen hydrochemistry of an urban aquifer. Appl Geochem 2004;19:709–19.
- Galloway DF. Addressing the global nitrogen problem. Abstr Paper Am Chem Soc 1999; 218:U591.
- Galloway JN, Dentener FJ, Capone DG, Boyer EW, Howarth RW, Seitzinger SP, Asner GP, Cleveland CC, Green PA, Holland EA, Karl DM, Michaels AF, Porter JH, Townsend AR, Vorosmarty CJ. Nitrogen cycles: past, present and future. Biogeochemistry 2004; 70(2):153–226.
- Gooddy DC, Withers PJA, McDonald HG, Chilton PJ. Behaviour and impact of cow slurry beneath a storage lagoon: II. Chemical composition of chalk porewater after 18 years. Water Air Soil Pollut 1998;107(1/4):51–72.
- Gooddy DC, Clay JW, Bottrell SH. Redox-driven changes in pore-water chemistry of the Chalk unsaturated zone beneath unlined cattle slurry lagoons. Appl Geochem 2002; 17:903–21.
- Gormly JR, Spalding R. Sources and concentrations of nitrate-nitrogen in ground water of the Central Platte Region, Nebraska. Ground Water 1979;17:291–301.
- Harms TK, Grimm NB. Hot spots and hot moments of carbon and nitrogen dynamics in a semi-arid riparian zone. J Geophys Res Biogeosci 2008:113.

- Haycock NE, Burt TP. Role of floodplain sediments in reducing nitrate concentration of subsurface run-off: a case study from the Cotswolds, UK. Hydrol Process 1993;7: 287–95.
- Heaton THE. Isotopic studies of nitrogen pollution in the hydrosphere and atmosphere a review. Chem Geol 1986;59:87–102.
- Heaton THE, Trick JK, Williams GM. Isotope and dissolved gas evidence for nitrogen attenuation in landfill leachate dispersing into a chalk aquifer. Appl Geochem 2005;20: 933–45.
- Jackson BM, Wheater HS, Wade AJ, Butterfield D, Mathias SA, Ireson AM, Butler AP, McIntyre NR, Whitehead R. Catchment-scale modelling of flow and nutrient transport in the Chalk unsaturated zone. Ecol Model 2007;209(1):41–52.
- Jackson BM, Browne CA, Butler AP, Peach DW, Wade AJ, Wheater HS. Nitrate transport in Chalk catchments: monitoring, modelling and policy implications. Environ Sci Policy 2008;11(2):125–35.
- Karamanos RE, Rennie DA. Nitrogen isotope fractionation during ammonium exchange reactions with soil clay. Can J Soil Sci 1978;58:53–60.
- Kendall C. Tracing nitrogen sources and cycling in catchments. In: Kendall C, McDonnell JJ, editors. Isotope tracers in catchment hydrology. Amsterdam: Elsevier; 1998. p. 519–76.
- Lapworth DJ, Gooddy DC, Kent F, Heaton THE, Cole SJ, Allen D. A combined geochemical and hydrological approach to understanding macronutrient sources. J Hydrol 2013; 500:226–42.
- Lawrence AR, Gooddy DC, Kanatharana P, Ramnarong V, Meesilp W. Groundwater evolution beneath a rapidly developing city in south east Asia. Hydrogeol J 2000;8:564–75.
- Lorah MM, Cozzarelli IM, Böhlke JK. Biogeochemistry at a wetland sediment—alluvial aquifer interface in a landfill leachate plume. J Contam Hydrol 2009;105(3–4):99–117.
- Lyngkild J, Christensen TH. Redox zones of a landfill leachate pollution plume (Vejen, Denmark). J Contam Hydrol 1992;10(4):273–89.
- Macdonald DMJ, Dixon A, Newell AJ, Hallaways A. Groundwater flooding within an urbanised flood plain. J Flood Risk Manag 2012a;5(1):68–80.
- Macdonald DMJ, Griffiths KJ, Lapworth DJ, Williams PJ, Stuart ME, Gooddy DC. Hydrogeochemical characterisation of a peri-urban floodplain: initial findings. British Geological Survey Internal Report, IR/12/049; 2012b. [97 pp.].
- MacDonald AM, Lapworth DJ, Hughes AG, Auton CA, Maurice L, Finlayson A, Gooddy DC. Groundwater, flooding and hydrological functioning in the Findhorn floodplain, Scotland. Hydrol Res 2014. http://dx.doi.org/10.2166/nh.2014.185. [in press].
 Mariotti A, Landreau A, Simon B. ¹⁵N isotope biogeochemistry and natural attenuation
- Mariotti A, Landreau A, Simon B. ¹⁵N isotope biogeochemistry and natural attenuation processes in groundwater: application to the chalk aquifer of northern-France. Geochim Cosmochim Acta 1988;52:1869–78.
- UK hydrometric register. In: Marsh TJ, Hannaford J, editors. Hydrological data UK series. Centre for Ecology & Hydrology; 2008. [210 pp.].
- McClain ME, Boyer EW, Dent CL, Gergel SE, Grimm NB, Groffman PM, Hart SC, Harvey JW, Johnston CA, Mayorga E, McDowell WH, Pinay G. Biogeochemical hot spots and hot moments at the interface of terrestrial and aquatic ecosystems. Ecosystems 2003;6: 301–12.
- Murray-Hudson M, Wolski P, Ringrose S. Scenarios of the impact of local and upstream changes in climate and water use on hydro-ecology in the Okavango Delta, Botswana. J Hydrol 2006;331(1-2):73-84.
- Oakes DB, Young CP, Foster SSD. The effects of farming practices on groundwater quality in the United Kingdom. Sci Total Environ 1981;21:17–30.
- Pinter N. Environment one step forward, two steps back on US floodplains. Science 2005;308(5719):207–8.
- Sebilo M, Mayer B, Grably M, Billou D, Maritotti A. The use of 'ammonium diffusion' method for δ¹⁵N-NH4⁺ and δ¹⁵N-NO₃⁻ measurements: comparison with other techniques. Environ Chem 2004;1:99–103.
- Silva SŘ, Kendall C, Wilkison DH, Ziegler AC, Chang CCY, Avanzino RJ. A new method for collection of nitrate from fresh water and the analysis of nitrogen and oxygen isotope ratios. J Hydrol 2000;228:22–36.
- Sophocleous M. Interactions between groundwater and surface water: the state of the science. Hydrogeol J 2002;10(1):52-67.
- Stuart ME, Lapworth DJ. A review of processes important in the floodplain setting. British Geological Survey; 2011 [32 pp. (OR/11/030)].
- Stuart ME, Gooddy DC, Bloomfield JP, Williams AT. A review of the impact of climate change on future nitrate concentrations in groundwater of the UK. Sci Total Environ 2011;409(15):2859–73.
- Thomas KL, Lloyd D, Boddy L. Effects of oxygen, pH and nitrate concentration on denitrification by Pseudomonas species. FEMS Microbiol Lett 1994;118:181–6.
- Tockner K, Stanford JA. Riverine flood plains: present state and future trends. Environ Conserv 2002;29(3):308–30.
- Wakida FT, Lerner DN. Non-agricultural sources of groundwater nitrate: a review and case study. Water Res 2005;39:3–16.
- Wang L, Stuart ME, Bloomfield JP, Butcher AS, Gooddy DC, McKenzie AA, Lewis MA, Williams AT. Prediction of the arrival of peak nitrate concentrations at the water table at the regional scale in Great Britain. Hydrol Process 2012;26:226–39.
- Wassenaar LI. Evaluation of the origin and fate of nitrate in the Abbotsford Aquifer using the isotopes of ¹⁵N and ¹⁸O in NO₃⁻. Appl Geochem 1995;10:391–405.
- Werritty A. Sustainable flood management: oxymoron or new paradigm? Area 2006; 38(1):16–23.
- Whitehead PG, Crossman J. Macronutrient cycles and climate change: key science areas and an international perspective. Sci Total Environ 2012;434:13–7.

Chapter 4: Groundwater flooding within an urbanised flood plain

The work presented in this chapter has been published in the following journal paper: Macdonald, D.M.J., Dixon, A., Newell, A. & Hallaways, A. (2012). Groundwater flooding within an urbanised flood plain. *Journal of Flood Risk Management*, 5(1), 68-80.

Groundwater flooding within an urbanised flood plain

D. Macdonald¹, A. Dixon², A. Newell¹ and A. Hallaways³

1 British Geological Survey, Wallingford, UK

2 Groundwater Monitoring and Drilling Ltd, Chalgrove, UK

3 NCPMS, Environment Agency, Frimley, UK

Correspondence

David Macdonald, British Geological Survey, Maclean Building, Wallingford OX10 8BB, UK Email: dmjm@bgs.ac.uk

DOI: 10.1111/j.1753-318X.2011.01127.x

Key words

Flood mitigation; flood plain; groundwater; groundwater flooding; Oxford; River Thames; urban flooding.

Abstract

In Europe in recent years, there has been recognition of the need to better understand the risk from groundwater flooding. This recognition has been due both to the occurrence of major flooding events clearly attributable to groundwater and the inclusion of groundwater flooding in European and national legislation. The case study of the city of Oxford on the River Thames flood plain in UK is used to examine the mechanisms for groundwater flooding in urbanised flood plain settings. Reference is made to an extensive data set gathered during a major flood event in 2007. Groundwater flooding of a significant number of properties is shown to occur in areas isolated from fluvial flooding because of high ground created historically to protect property and the transport network from flood inundation. The options for mitigating this form of flooding are discussed; measures to increase the rate of conveyance of flood waters through Oxford, designed to reduce fluvial flood risk, have also been recognised as a means for reducing groundwater flood risk within the city.

Introduction

Groundwater flooding is the emergence of groundwater at the ground surface away from perennial river channels and can also include the rising of groundwater into man-made ground, including basements and other subsurface infrastructure (Macdonald et al., 2008). The impact of groundwater flooding can be severe under conditions where the 'normal' ranges of groundwater level and groundwater flow are exceeded. In Europe, the risk from groundwater flooding has received more attention since its inclusion in the EU Floods Directive (2007/60/EC). The Directive, which came into force in November 2007, includes provisions for assessing the risk from groundwater flooding (Cobby et al., 2009), producing groundwater 'flood hazard maps' and introducing measures to address any significant risk. The inclusion of groundwater within the Directive follows serious groundwater flooding events in the past decade. The impact of groundwater flooding has been most severe in areas of Chalk outcrop and in the flood plains of major rivers. Groundwater flooding in Chalk catchments occurs where antecedent conditions of high groundwater levels and high unsaturated zone moisture content combine with intense rainfall. Resulting groundwater level rises of up to tens of metres can cause significantly increased stream base flow and spring flow, and the reactivation of dormant springs in dry valleys remote from perennial stream channels. Flooding is often prolonged because of high groundwater levels maintained by extended periods of drainage from the unsaturated zone (Pinault *et al.*, 2005). Examples include the flooding in 2000 and 2003 in southern England (Finch *et al.*, 2004) and the Somme Valley (Negrel and Petelet-Giraud, 2005).

The focus of this paper is groundwater flooding in the flood plains of major rivers. Where the deposits associated with flood plains are permeable, these are generally saturated to levels close to the ground surface and hydraulically well connected to the associated rivers. Here, groundwater can contribute significantly to river flow during summer months and extended dry periods; in wetter periods, rivers can be effluent, recharging the flood plain sediments. During periods of increased flow, and before the banks are overtopped, the naturally high ground of river levees can contain river water while low-lying ground beyond can be flooded because of rising groundwater. In the case of the River Danube flood plain in 2008, damage occurred to buildings because of water pressure on basements, resulting from abnormally high groundwater levels associated with high river levels (Kreibich and Thieken, 2008). The onset and recession of groundwater flood events in this type of setting is typically much shorter than that in Chalk catchments.

Flood plains in the past have been attractive locations for urban development for reasons such as their coincidence

Figure 1 River Thames and its tributaries in the Oxford area. BGS © NERC 2011. Ordnance Survey topographic material reproduced with the permission of Ordnance Survey on behalf of The Controller of Her Majesty's Stationery Office, © Crown Copyright. Licence number 100017897/2011.

with sources of water and food, their potential for cultivation, and their proximity to transportation routes (Montz, 2000). Where settlements were originally located on high ground within or close to river valleys, these have inevitably spread onto the flood plain. Historically, this development has often continued even with the threat of frequent flooding. In urban areas on the flood plain, a cycle can develop in which defences to protect property and infrastructure from flooding increases the general risk from flooding, for example by removing flood storage, and hence raising flood levels, which in turn requires more substantial defences. In recent times, the folly of continuing this cycle has been recognised and a move has been made towards working with natural processes to manage the risk of flooding rather than trying to remove it (Fleming, 2002).

The topographical changes to flood plains resulting from urbanisation can have a significant influence on the location, timing and extent of groundwater flooding. This can make the assessment of risk highly complex. Identifying measures to reduce the risk of groundwater flooding as part of overall flood risk management schemes is challenging. A case study is reported here of the city of Oxford in the UK, which is located on the flood plain of the upper reaches of the River Thames. Groundwater flooding has been recognised in the city as a component of the overall flooding story (Macdonald *et al.*, 2007). The Oxford case study is used as a means to examine the risks from groundwater flooding and the options for mitigating these risks in an urbanised flood plain.

Background

The city of Oxford is situated within a relatively narrow valley of the upper reaches of the River Thames (Figure 1). The flood plain is on average 2 km wide; however, it narrows downstream to only 0.5 km. Although most of the city is located on older river terraces and bedrock above the current flood plain, from the late 19th century, pressure for housing near the city centre resulted in significant urban development on the flood plain; it is estimated that, currently, approximately 3400 primarily residential properties and a large number of commercial properties are located within the 100-year return period flood event envelope as defined by the Environment Agency. The urban areas of Oxford have

Figure 2 Superficial geology in the Oxford valley. BGS © NERC 2011. Some features of this map are based on digital spatial data licensed from the Centre for Ecology and Hydrology (Moore *et al.*, 1994). This map includes NEXTMap Britain elevation data from Intermap Technologies.

historically suffered from serious flooding. There were relatively few major flood events in the second half of the 20th century; however, in recent years, there have been four notable floods over the period of a decade, in April 1998, December 2000, January 2003 and most recently in July 2007. The Oxford flood plain is underlain by permeable shallow sands and gravels (Figure 2), and a significant number of properties were affected by flooding from rising groundwater, which was either the sole cause or the initial cause prior to inundation from fluvial waters.

In February 2002, the Environment Agency of England and Wales commenced the Oxford Flood Risk Management Study (OFRMS) to identify options to reduce the flood risk in Oxford within the 100-year return period flood plain of the River Thames and its tributary, the River Cherwell (Ball *et al.*, 2008). During the early stages of the OFRMS, it was recognised that groundwater flooding and the links to fluvial flooding were important considerations. Together with the British Geological Survey, a jointly funded research project was initiated in 2005 on the controls, location and timing of groundwater flooding in the city. The insights from this project have informed the choice of proposed flood mitigation measures.

In July 2007, serious flooding took place within the flood plain of the River Thames in Oxford. This was the result of a moist, subtropical air mass moving slowly north over central England, resulting in extreme rainfall on July 19 and 20. Some of the highest total rainfalls occurred in the headwaters of the River Thames, for example 140.1 mm in 24 h in Chastleton, Oxfordshire (Marsh and Hannaford, 2008). This rainfall event followed what was the wettest summer in England and Wales since 1912. Resulting flow in the River Thames to the south of Oxford at Sandford peaked at 224.8 m³/s; this was the maximum flow at this location in 2007 and the fifth highest annual maximum flow since 1894. The peak river flow in Oxford occurred approximately 5 days after the rainfall event. The total rainfall in Oxford itself for July 19 and 20 was measured at 60.6 mm. The monitoring in place for the groundwater flooding element of the OFRMS was able to capture the impacts on water levels of the July 2007 event and provides the primary reference for this paper.

Methodology

The approach taken within the Oxford study was to develop a baseline understanding of the river and groundwater system, and use any flood events that occurred during the study period to explore the potential role of groundwater in flood events covering a range of return periods. The components of the study are described here.

Topographic data

In 2005, the Environment Agency undertook a topographic survey of the flood plain using LIDAR (Light Detection and Ranging), an optical remote sensing technology that measures ground elevation to centimetric accuracy (Cracknell and Hayes, 2007). Among other applications within this study, LIDAR data provided a means to assess the urban build-up on the flood plain and its impact on flood pathways, and the potential for groundwater flooding.

Three-dimensional (3D) modelling

A 3D geological model of the flood plain superficial deposits in the study area was built to aid the conceptualisation of the shallow groundwater system and to enable the potential storage capacity of the aquifer to be assessed. The model comprises three layers, including made ground, alluvium, and the underlying sands and gravels. The base of the model over the majority of the study area is the Oxford Clay, a Jurassic argillaceous sedimentary rock. ArcGIS, in conjunction with the 3D visualisation packages, geological surveying and investigation in 3 dimensions (Kessler et al., 2009) and geological object computer aided design, were used to construct the model. Borehole logs were used to create a series of cross-sections with surfaces produced by interpolation between the sections. The layers within the geological model were attributed with estimates of specific yield made during previous studies within the Oxford flood plain (Institute of Hydrology, 1987; Dixon et al., 1990; Gardner, 1991). The geological model was then combined with groundwater level surfaces contoured using measured groundwater and surface water levels.

Monitoring network

An extensive network of groundwater and surface water level monitoring points exists within the Oxford flood plain resulting from this and previous studies. At some sites, monitoring has been undertaken since the early 1980s (Institute of Hydrology, 1987). Manual monitoring of water levels was undertaken on a monthly basis between 2005 and 2009; the total number of monitoring points at the end of 2009 was 235. Digital water level recorders were installed in 51 of these monitoring points. At a number of sites, there are combinations of two or more of the following: piezometers completed in the gravel aquifer, piezometers completed in the overlying alluvium, surface water stilling wells and flood water recorders. Data were made available from recorders installed in the upstream and downstream waters at four of the Environment Agency locks in the study area. During the July 2007 flood, a further six temporary manual flood water monitoring points were established, which provided additional data in urban areas for the rise and recession of flood waters during the event. As well as water levels, river flows are routinely measured within the study area by the Environment Agency on the River Thames and its tributaries, the Rivers Evenlode and Cherwell. Rainfall is measured on a 15-min interval at two of the locks. There is also a long rainfall record from the Radcliffe Meteorological Station at Oxford.

Flood observation and impacts

During the period of the study, one major flood event (July 2007) and a number of minor floods occurred. Observation during floods is crucial in understanding flood mechanisms. Flood sources and pathways can vary over the period of a flood, and a monitoring network cannot feasibly capture all of this for a study area as large as the Oxford flood plain. A questionnaire sent out by the Environment Agency following the December 2000 flood event captured some information on the occurrence of groundwater flooding, but generally it is hard to obtain such information; in addition to the reluctance of owners to provide information on the impact of flooding on their property, it is often difficult to identify with confidence that the source of flooding is groundwater. Information was obtained on possible locations of groundwater flooding in 2003 and 2007 based on residents' observations.

Flood mapping

A national assessment of groundwater flood susceptibility undertaken by the British Geological Survey (McKenzie et al., 2010) includes the River Thames flood plain in the Oxford area within its 'very high' category, that is, it has mapped the underlying superficial geology as permeable and the groundwater levels within 2 m of ground level. An attempt was made to improve on this broad-scale assessment in Oxford and to quantify risk rather than just susceptibility. Although separate river (1D and 2D) and groundwater models have been developed for Oxford, attempts to link these to enable simulation of the river-aquifer response during flood events is still in the development stage. Preliminary mapping of groundwater flood-prone urban areas was attempted based on the understanding of flood mechanisms gained during the study. LIDAR data were used to identify those low-lying urban areas that are isolated from fluvial flooding, at least for low return period events or in the early stages of higher return period events. For these areas, the minimum level at which groundwater flooding could potentially occur and the level at which fluvial waters would inundate a location were assessed. Based on the understanding of flood water pathways and levels during an event and the likely response of groundwater levels, the potential for groundwater flooding at the locations identified earlier was assessed. Where Environment Agency property threshold data were available, the groundwater flood levels identified were used to assess the likelihood of groundwater flooding affecting individual properties. Anecdotal information on the location of groundwater flooding in the three recent major floods was used to validate the outcome from this step. Where property threshold data were not available, the reported incidents of groundwater flooding were used to improve confidence. The output from an Environment


Figure 3 Thickness of made ground above the natural flood plain. BGS © NERC 2011.

Agency river flood model (Environment Agency, 2009), which gives the flood elevation for flood return periods of 2, 5, 10, 20, 50, 100, 200 and 500 years, was used to give a preliminary indication of the range of fluvial flood return periods for which groundwater flooding on its own might occur.

Results

Baseline conditions

Analysis of the topographic data for the flood plain, involving the picking of locations that are at the natural flood plain level and interpolating, allowed the thickness of the made ground above the natural flood plain to be mapped (Figure 3). This highlights the built-up residential and commercial areas within the city, the road and rail networks, and those sites previously used for dumping waste. The role of this high ground in constraining the movement of flood waters down the flood plain is important.

The geological logs used to create the 3D geological modelling of the superficial deposits underlying the Oxford flood plain all include some alluvium below the ground surface. The model produced has a continuous layer of alluvium of up to 4 m in thickness, but more typically 1.5 m, which covers a layer of sands and gravels beneath the flood plain, typically within a range of 2–6 m in thickness.

Water level data show that the flow of groundwater within the flood plain sediments is complex, due largely to the influence of locks and weirs on the River Thames, and associated bypass streams. These have created numerous zones of recharge from, and discharge to, the river network. Groundwater levels generally fluctuate within the upper few metres of the flood plain sediments with a greater range of fluctua-



Figure 4 Typical river and groundwater hydrographs from the Oxford flood plain monitoring network. BGS © NERC 2011.

tion occurring at the flood plain margins. In the vicinity of rivers and streams, groundwater levels generally correlate well with surface water levels, indicating good hydraulic connection (Figure 4). Dixon (2004) reports hydraulic conductivity of the sands and gravels in the River Thames flood plain in Oxford as very high, ranging between 100 and 1000 m/d. This compares with an estimate of hydraulic conductivity of the alluvium of 0.3 m/d made by both Gardner (1991) and Hodgson (2008).

There is no evidence of buildings on the flood plain with basements constructed below the base of the alluvium, and building foundations are not thought to have a major influence on groundwater flow within the flood plain as these are generally less than a metre in depth for domestic properties, and therefore unlikely to affect lateral groundwater movement within the sands and gravels aquifer. The few large buildings on the flood plain were constructed on raised ground, and the foundations will not have penetrated the gravel aquifer sufficiently to significantly change groundwater flow patterns. There is only one location within the urbanised flood plain where a relatively small volume of gravel has been extracted and the void created has not been infilled.

Combining groundwater levels with the LIDAR data allows the depth to groundwater to be contoured; this is

shown in Figure 5 for May 2007, a period when the groundwater levels were relatively low. This figure (in combination with Figure 3) highlights that the depth to groundwater is generally greater under the man-made ground and at the valley edges; groundwater is shallower in areas close to the River Thames upstream of locks, where raised heads cause enhanced aquifer recharge from the river, and also at the southern end of the valley where the narrowing of the flood plain causes restricted flow of groundwater down-valley.

The available storage within the unsaturated zone of the flood plain is very small compared with flood water volumes. Estimating this storage is highly problematic because of the variability of the lithology of the flood plain deposits, the challenge in estimating storage parameters and the influence of the capillary zone. However, using an approximation of less than 10% for specific yield of the flood plain deposits (Institute of Hydrology, 1987; Dixon et al., 1990; Gardner, 1991), the volume of available subsurface storage within the flood plain (as defined by the outcrops of alluvium and flood plain gravels, Figure 2) for May 2007 is calculated to have been 2.6×10^6 m³. For comparison, this is equivalent to just over 3 h of the peak flow in the River Thames downstream of Oxford during the July 2007 flood event. The available subsurface storage is significantly smaller during typical winter periods.



Figure 5 Depth to groundwater within the flood plain superficial deposits in the Oxford valley based on groundwater levels measured in May 2007. BGS © NERC 2011.

Flood conditions

Surface flood water pathways

Surface flood waters were observed following generally similar pathways in July 2007 to the floods in 2000 and 2003, which were also primarily River Thames floods (in comparison, the flood in 1998 was associated more with the River Cherwell catchment). Fluvial flood waters flowed south, parallel with the line of the River Thames. Major structures caused barriers to this flow, including the embankment of a dual carriageway (A34), the Botley Road and associated properties, and the Oxford southern bypass road and historic waste dumps in the south (Figure 3). In the south of Oxford, the main Birmingham to London railway line and the urban areas of Grandpont and New Hinksey (Figure 3), both of which run north to south, separated the flood waters in the west and the east of the valley. The result of all of these areas of high ground was the creation of a series of flood cells, which gradually filled as flood waters continued to enter the Oxford section of the Thames valley from upstream. The flood waters within these cells eventually overtopped, causing flooding of property in the Botley Road

area and in the New Hinksey area. Approximately 160 properties were flooded internally in the flood of 2000, a similar number in 2003 and over 200 in 2007. The 2000 and 2003 floods were classified by the Environment Agency as 10- to 15-year return period events, and the 2007 flood event as a 15- to 20-year return period event. In New Hinksey, at a few locations, water flowed from the flood cell associated with the Hinksey Stream, east towards the River Thames which had flooded to a lower level. Figure 6 shows a map of approximate peak flood water elevations during the July 2007 flood, which identifies the flood cells.

Prior to flood waters overtopping structural barriers to flood, subsurface pipework was also seen to be a key pathway for flow. For example, water from flooded areas was flowing out of storm drains on the downstream side of topographic barriers. The ballast fill that surrounds underground pipes can also provide a high permeability pathway for groundwaters during flood events (J. Packman, pers. comm., Centre for Ecology and Hydrology).

Topographic controls on the location of groundwater flooding

River level, flood level and groundwater level data collected during the July 2007 event have helped understand the mechanism by which groundwater flooding occurs within the city of Oxford. Anecdotal information on the location and nature of flooding in 2000 and 2003 suggests the same mechanism controlled groundwater flooding in these events.

In the July 2007 event, responses in groundwater levels were seen both as a result of the rain falling directly on Oxford and the high river levels that occurred in the following days. In the 17 boreholes with automatic water level recorders completed in the shallow gravel aquifer (which are well distributed across the study area), increases in groundwater level seen in the day following the event ranged from 0.28 m to 1.23 m, with an average of 0.59 m. Groundwater levels did not rise above ground level at any of these sites at this time, although in all cases, groundwater levels were within the alluvium (nota bene the limited areas of standing water that occurred within the city immediately following the rainfall event were due to the drainage system being overwhelmed by the volume of rain water). Following the initial peak in groundwater levels, there was a period of recession of up to a few days. These groundwaters then responded to the rises in river levels caused by flood waters reaching Oxford from higher in the River Thames catchment. In the majority of the 17 groundwater monitoring sites, groundwater levels rose above ground level at their peak. Artesian conditions were also measured at 12 additional sites that were manually dipped approximately a day before the flood peak.



Figure 6 Estimated peak flood water elevation in the Oxford area during the July 2007 flood event. This map includes NEXTMap Britain elevation data from Intermap Technologies. BGS © NERC 2011.

Groundwater response to rainfall and fluvial flooding at sites in the south Oxford area during July 2007 is shown in Figure 7, along with river flood levels. These include (Figure 1) water levels associated with the Hinksey Stream, the River Thames upstream of Iffley lock and the Weirs Mill Stream, and the groundwater level in a borehole, NH1, completed in the sands and gravels in the New Hinksey area (Figure 6). Water levels prior to 19 July 2007 show that the Hinksey Stream and River Thames upstream of the Iffley Lock are at a similar elevation. The Weirs Mill Stream is at a much lower level, similar to the downstream elevation of Iffley Lock. A borehole located next to the Weirs Mill Stream provides evidence that the stream normally acts as a line of discharge from the gravel aquifer. This discharge influences the groundwater level in borehole NH1 (Figure 7). The response in these river and groundwater levels occurred within 4-8 h of the start of the rainfall event. The groundwater level in NH1 rose by 0.53 m, peaking within approximately 9 h of the start of the event. After this point, groundwater levels began to recess. The River Thames levels were partially controlled by the raising of the weir boards in anticipation of high flows to follow. The Hinksey Stream has limited management structures on it and continued to rise over the following days, overbanking and flooding into the surrounding area, primarily farm land. Eventually, flows in the River Thames were also too great and it too overbanked. After the initial rain-dependent peak, the groundwater level at NH1 recessed for approximately 2 days but then started to rise again. A double peak was seen in the flood waters associated with the Hinksey Stream, caused by the lag in flood waters moving from headwaters of a number of tributaries of the Thames. This double peak is reflected in the shape of the groundwater hydrograph for NH1, demonstrating that the fluvial flood waters have some control on the groundwater levels. The peaks allow a good estimate to be made of the lag between the fluvial floods and the groundwater at NH1; the lag between the first peaks was approximately 20 h and between the second peaks was approximately 22 h. Following the second peak, the groundwater levels again recessed, with the gravel aquifer drainage in the locality being controlled by the lower fluvial flood levels associated with downstream of Iffley Lock.



Figure 7 Groundwater levels in borehole NH1 in the New Hinksey area of Oxford during the July 2007 flood event, along with water levels nearby in the River Thames, the Weirs Mill Stream and the Hinksey Stream. Also shown, 15-min rainfall data, ground level at NH1 and the level of the threshold point above which the River Thames would flood the ground at NH1. BGS © NERC 2011. Rainfall and water level data for the River Thames were provided by the Environment Agency.

In the area of borehole NH1, there was significant flooding of gardens, a small number of low-lying properties flooded above the ground floor and many more flooded under ground-floor floorboards. This low-lying area has been protected from fluvial flooding in the past three major floods as it is surrounded by high ground created for the main railway line, roads and housing to raise them above the flood plain. The comparison of groundwater level in NH1 and flood water level in the vicinity, as well as observations of artesian flow, confirmed that the flooding was caused by the emergence of groundwater at surface. At peak flooding, the gravel water level was only 1 cm higher than the flood water level, indicating that the relatively low permeability alluvium, which is 1.1 m thick at this location, did not significantly inhibit the vertical movement of groundwater.

The depth of groundwater flooding at NH1 was, at its peak, 0.25 m. The peak flood waters had an elevation of 55.54 maOD; the threshold of the nearest flooded property to NH1 is 55.42 maOD. The level beyond which fluvial flooding from the River Thames to the east would breach the high ground and flow into the low-lying area in the vicinity of NH1 is approximately 0.2 m above the highest level to which the River Thames rose in this area.

The situation that occurred in the New Hinksey area was seen elsewhere in the urban flood plain areas of Oxford where built-up ground has isolated low-lying areas, protecting them from low return period fluvial flood events but also creating the conditions for groundwater flooding during these events. Often the groundwater flooding only impacts areas of relatively low importance, such as gardens or outhouses; however, there is a significant number of properties on low-lying ground that are vulnerable. Figure 8 illustrates the conditions during the July 2007 flood at a location in the Botley Road area of Oxford, where properties downgradient of the road were initially protected from fluvial flooding by high ground, but where flooding was reported which it was thought was due to rising groundwater. In the latter period of the event, the properties suffered fluvial flooding as surface flood waters overtopped the Botley Road. Figure 8 shows an upstream fluvial flood level and an interpolated groundwater level for the area of the properties based on monitoring in boreholes a few hundreds of metres from the location.

The greatest number of potentially vulnerable properties in Oxford, however, are those older properties in which the void created by raising the ground floor up from the flood plain when they were first built was converted to living space during the period of relatively infrequent high return period flood events in the second half of the 20th century. These rooms below the ground floor are relatively low-lying, at a level close to that of the natural flood plain. Tens of properties that have basement conversions have been identified in south Oxford, and there is evidence of some of these being flooded in the recent events. Some property owners have added waterproof membranes that have been successful in stopping groundwater ingress during flood events.



Figure 8 Fluvial flood water levels upstream of the Botley Road and interpolated groundwater levels at an urban location down-gradient of the Botley Road, Oxford during the July 2007 flood event. Also shown, 15-min rainfall data, the threshold of the lowest property in the urban area down-gradient of the Botley Road and the level above which the flood waters upstream of the Botley Road would overtop. BGS © NERC 2011. 15-min rainfall data were provided by the Environment Agency.

Recession of water levels after flooding events

A typical characteristic of groundwater floods is its relatively slow onset and recession in comparison with fluvial floods. For example, in the Chalk aquifer of northern Europe, flooding in some locations has been seen to last for months after the fluvial flood waters associated with the same events have recessed (Pinault et al., 2005). In permeable flood plain deposits, the recession of groundwater levels will be significantly faster. The strong hydraulic connection that sees groundwater levels respond quickly to rises in river level also means that river channels are effective at draining aquifers once the fluvial flood event has passed. In July 2007, it took groundwater levels in the gravel aquifer between only 2 and 15 days to recede back to below ground level from first becoming artesian. However, it was observed that in some isolated low-lying areas, even though groundwater levels in the gravels recessed, flood waters sitting on top of the relatively low permeable alluvium sediments took a longer period to drain.

Mapping of groundwater flooding

The mapping of groundwater flooding was undertaken to help the Environment Agency gauge the scale of the vulnerability in Oxford from this form of flooding. The approach, described in the methodology section earlier, was a first-pass mapping exercise that made a number of significant assumptions. Any restriction on the vertical movement of groundwater from the gravel aquifer to above ground level due to the alluvium layer was not taken into account. The mapping exercise made the assumption that groundwater flooding could occur wherever the groundwater head was thought likely to be above ground level. In areas where there were few groundwater level data, an assumption was made, based on the strong hydraulic connection between river and aquifer, that groundwater levels could be estimated by interpolating between adjacent flood cells.

The mapping exercise showed that there are large areas of urban Oxford that could potentially be affected by groundwater flooding that for certain flood return periods would not be prone to fluvial flooding. However, the majority of these areas are gardens and not internal to property. There are estimated to be only tens of properties that may be affected by groundwater flooding at ground floor level where fluvial flooding had not already occurred. However, this assessment was limited by the availability of flood threshold data as there are over 200 properties in areas that could be vulnerable which have not had their ground-floor flood threshold level surveyed. The exercise identified a significant number of properties that are potentially vulnerable to groundwater flooding of rooms below ground floor level. Walking surveys identified over 80 properties with basements that may be in use as living areas or for storage.

With reference to the output of an Environment Agency river model, the flood events from which properties are most vulnerable to groundwater flooding alone were identified as those with return periods of 10–25 years. Lower return period events would appear to result in flood levels that are not sufficiently high to cause serious groundwater flooding; events with higher return periods would likely cause fluvial flooding to mask the initial groundwater flooding. For these events, groundwater would again be an issue if flood defence measures were introduced that held back fluvial flood waters but did not reduce the heads driving water into the gravel aquifer.

Discussion

The experience of flooding in Oxford highlights the complexities created by an urbanised area located on a major river flood plain. Topographical variations, resulting in many cases from construction to protect properties and the transport network from flooding, create complex flood pathways and areas of high vulnerability. Data collected during flood events in Oxford have shown that groundwater flood-prone areas can be created by the isolation of lowlying areas from fluvial flooding as a result of surrounding man-made ground. Groundwater levels rise in response to direct rainfall, and to elevated river levels and associated fluvial flood zones. Waters from the fluvial flood zones make their way to these isolated low-lying areas by passing through the permeable sediments underlying the flood plain. Although water level data indicate that the low permeability, near-surface alluvium inhibits the flow of water into and out of the underlying sands and gravels aquifer, there is a substantial amount of evidence that the alluvium is sufficiently permeable to allow groundwater flooding to occur. However, data on the alluvium is limited and is not currently sufficient to identify how vertical flow during periods of groundwater flooding is spatially distributed, and whether it is dominated by windows of higher permeability material. Where buildings have been constructed, the removal of alluvium also provides preferential pathways for upward groundwater movement as do man-made drainage channels.

Where artesian conditions do occur, these can result in flooded gardens, cause drainage problems (e.g. inundation of sewerage systems) and create dampness beneath floorboards. Groundwater can also flood properties in these low-lying areas where the ground floor or the inhabited basement is close to the level of the natural flood plain.

Quantifying the risk of groundwater flooding in these urban environments is difficult. The combination of flood mechanisms means that the degree of groundwater flooding will vary according to the nature of the overall flood event. The height to which groundwater levels reach during a flood event will depend on a number of flood event-specific factors: the amount of rainfall directly on the city as opposed to the upstream catchment, the rate at which river levels rise and the period for which fluvial flooding persists, as well as the antecedent soil and groundwater conditions. A methodology based on detailed topography, property thresholds, groundwater level data and the output from a river flood model has been used here to provide a preliminary assessment of groundwater flood risk that fits well with recent observed and reported flood events.

Options for mitigating groundwater flood risk are limited. Where basements are prone to flooding, these can be waterproofed or pumps can be installed, although the anecdotal evidence available from the 2007 flood event showed the latter to have variable effects. Impermeable barriers to the base of the shallow aquifer to stop groundwater flow into an area of housing is an option, although the change in groundwater flows patterns could have detrimental environmental impacts outside of flooding periods. The main approach being proposed by the Environment Agency for fluvial flood risk reduction in the Oxford area is to increase the conveyance of flood waters through the Thames flood plain. This would be achieved by the removal or widening of structures that restrict flow, maximising the use of existing channels, and the limited introduction of new channels. Baffles would be installed to maintain flows during low flow periods and to avoid the gravel aquifer being overdrained such that lowered groundwater levels have detrimental impacts on dependent ecosystems.

It is thought that this approach to fluvial flood risk reduction is an appropriate means to reduce groundwater flood risk in Oxford and in similar settings as it reduces groundwater recharge from fluvial flood zones. It is recognised that there remains the potential for groundwater levels to rise above the ground surface as a result of direct rainfall recharge. This will depend again on antecedent soil and groundwater levels. In the Oxford case, using the attributed 3D geological model and a water table surface contoured from monitored water level data collected prior to the July 2007 event, a very approximate value for the averaged capacity of the flood plain aquifer to accept recharge was calculated as equivalent to a rainfall event of the order of 150 mm. However, locally, including areas with groundwater floodprone properties where the water table is relatively shallow, a substantially smaller rainfall event could result in groundwater levels rising above ground level due to direct rainfall recharge alone. In winter periods, the risk of this occurring is significantly higher due to the degree of saturation of the shallow flood plain sediments. This insight has implications for the application of sustainable urban drainage systems in similar settings, suggesting that techniques that delay recharge are worth considering.

Conclusions

- In recent years in Europe, there has been recognition that there is a requirement to better understand the risk from flooding as the result of abnormally high groundwater levels. This has been due both to the occurrence of major flooding events clearly attributable to groundwater and the related inclusion of groundwater flooding in European and national legislation.
- 2. In flood plains underlain by highly permeable deposits, groundwater rise leading to groundwater flooding can be due to direct rainfall recharge as well as flow into the sediments from rivers with high water levels and areas inundated with fluvial flooding. However, the good hydraulic connection between river and aquifer means that the aquifer can drain quickly as fluvial flood waters recess. Groundwater flooding in these settings is relatively short-lived compared with other groundwater flood settings, for example Chalk catchments.
- 3. The Oxford case study has shown that man-made ground built up from the natural flood plain to reduce the risk of fluvial flooding of property and the transport network can create adjacent isolated low-lying areas that are prone to groundwater flooding. In Oxford, there are a limited number of properties that are flooded above ground level in these areas but potentially tens of properties with inhabited basements that could flood as a result of abnormally high groundwater levels. It is estimated that these properties are affected by groundwater flooding alone during relatively low return period flood events of 10–25 years; during higher return period events, groundwater flooding will precede fluvial flood inundation.
- 4. Appropriate options for mitigating groundwater flooding in these urban flood plain settings are limited particularly where it is the result of direct rainfall recharge. As is proposed through the Oxford Flood Risk Management Study, an effective means to reduce the risk of groundwater flooding at the city scale is to increase the rate of conveyance of flood waters, reducing the heads within the flood cells that can drive water into the underlying permeable sediments. The potential for groundwater flooding due to rainfall alone has implications for the implementation of sustainable urban drainage systems.
- 5. Generally, in urban flood plain settings, fluvial flood assessments will underestimate the extent of flooding if aquifer pathways are not taken into account. Further work is required to link river and groundwater models to simulate flood events, to enable better quantification of the risk of groundwater flooding and to assess the potential for using river level monitoring in early warning systems for groundwater flooding.

Acknowledgements

The authors would like to thank past and present colleagues who have contributed to the Oxford groundwater flooding study, in particular Nigel Bray and Rob Hall of the Environment Agency, David Carden, Neil Blazey, Ross Kettle and David Wilson of Black & Veatch, and Sam and Tom Dixon of Groundwater Monitoring & Drilling Ltd. Thanks also goes to Hanson Aggregates for allowing the use of their water level data. We would also like to thank those residents of Oxford who have helped in the collection of monitoring data and in the sharing of knowledge on flooding events in the city.

References

- Ball L.G.A., Clegg M.J., Lewis L. & Bell G. Finding a long term solution to flooding in Oxford: the challenges faced. Proceedings of the European Conference on Flood Risk Management: Research into Practice. 30 September–2 October 2008 Keble College, Oxford, UK, 2008.
- Cobby D., Morris S., Parkes A. & Robinson V. Groundwater flood risk management: advances towards meeting the requirements of the EU floods directive. *J Flood Risk Manag* 2009, **2**, (2), 111–119.
- Cracknell A.P. & Hayes L. *Introduction to remote sensing*, 2nd edn. London: Taylor and Francis, 2007.
- Dixon A.J. Hydrology of the Oxford Meadows cSAC. Environment Agency Report, 2004.
- Dixon A.J., Bradford R.B., Cooper D.M., Reeve C.R. & Tucker D.K. A hydrological assessment of the impact of proposed sand and gravel extraction on an adjacent ecologically sensitive floodplain environment – a case study. In: J.W. Gaskarth & A.C. Lumsden, eds. *Proc. of 6th Extractive Industry Geol. Conf.* Birmingham: Birmingham University, 1990, 70–81.
- Environment Agency. Oxford flood risk management strategy hydraulic modelling – model build and calibration final report. Bristol, Environment Agency, 2009.
- Finch J.W., Bradford R.B. & Hudson J.A. The spatial distribution of groundwater flooding in a chalk catchment in southern England. *Hydrol Process* 2004, **18**, (5), 959–971.
- Fleming G. *Flood risk management*. London: Thomas Telford Ltd, 2002.
- Gardner C.M.K. *Water regime of river meadows: Yarnton Mead case study*. Wallingford: Institute of Hydrology Report to River and Coastal Engineering Group, Ministry of Agriculture Fisheries and Food, 1991.
- Hodgson R. Groundwater and flooding in Oxford: examining key soil processes. MSc Dissertation, University of Reading, 2008.
- Institute of Hydrology. Worton Rectory Farm (Oxon) feasibility study on the likely effects of proposed gravel scheme on Yarnton and Pixey Mead SSSI. Institute of Hydrology Report, 1987.

- Kessler H., Mathers S.J. & Sobisch H.G. The capture and dissemination of integrated 3D geospatial knowledge at the British Geological Survey using GSI3D software and methodology. *Comput Geosci* 2009, **35**, 1311–1321.
- Kreibich H. & Thieken A.H. Assessment of damage caused by high groundwater inundation. *Water Resour Res* 2008, 44, W09409, doi:10.1029/2007WR006621.
- Macdonald D.M.J., Bloomfield J.P., Hughes A.G., MacDonald A.M., Adams B. & McKenzie A.A. Improving the understanding of the risk from groundwater flooding in the UK. Proceedings of the European Conference on Flood Risk Management, Oxford, 30 September–2 October, 2008.
- Macdonald D.M.J., Hall R., Carden D., Dixon A., Cheetham M., Cornick S. & Clegg M. Investigating the inter-dependencies between surface and groundwater in the Oxford area to help predict the timing and location of ground-water flooding and to optimise flood mitigation measures. Proceedings of 42nd Defra Flood and Coastal Management Conference, 2007.
- Marsh T.J. & Hannaford J. The summer 2007 floods in England & Wales – a hydrological appraisal. Centre for Ecology & Hydrology Report, 2008.

- McKenzie A.A., Rutter H.K. & Hulbert A.G. The use of elevation models to predict areas at risk of groundwater flooding. In: C. Fleming, S.H. Marsh & J.R.A. Giles, eds. *Elevation models for geoscience*. London: Geological Society of London, Geological Society Special Publications, 345, 2010, 75–79.
- Montz B.E. The generation of flood hazards and disasters by urban development of floodplains, In: D.J. Parker, ed. *Floods*, Vol. 1. London & New York: Routledge, 2000, 116– 127.
- Moore R.V., Morris D.G. & Flavin R.W. Sub-set of UK digital 1:50,000 scale river centre-line network. Wallingford: NERC Institute of Hydrology, 1994.
- Negrel P. & Petelet-Giraud E. Strontium isotopes as tracers of groundwater-induced floods: the Somme case study (France). *J Hydrol* 2005, **305**, (1–4), 99–119.
- Pinault J.-L., Amraoui N. & Golaz C. Groundwater-induced flooding in macropore-dominated hydrological system in the context of climate changes. *Water Resour Res* 2005, **41**, 1029– 1036.

Chapter 5: Persistence of flooding on permeable lowland floodplains: the role of shallow groundwater and surface infiltration

The work presented in this chapter is to a standard and in a format that enables it to be submitted to an international peer-reviewed journal.

Persistence of flooding on permeable lowland floodplains: the role of shallow groundwater and surface infiltration

Macdonald DMJ^{a,*}, Jackson CR^b, Wade AJ^c

- ^a British Geological Survey, Maclean Building, Wallingford, Oxfordshire, OX10 8BB, UK (* corresponding author, <u>dmjm@bgs.ac.uk</u>)
- ^b British Geological Survey, Environmental Science Centre, Keyworth, Nottinghamshire, NG12 5GG, UK
- ^c School of Archaeology, Geography and Environmental Science, University of Reading, Reading, RG6 6AB, UK

Abstract

Natural levees and modified topography can retain water on urbanised floodplains for long periods after the peak of fluvial flood events. The duration that flood waters remain on the floodplain can be reduced by drainage infrastructure but where this is absent, shallow groundwater and low hydraulic conductivity surface geology are key controls. Few flood inundation models sufficiently incorporate groundwater and infiltration processes. A model system was developed coupling a commonly used flood inundation model with a groundwater flow model which allows the vertical exchange of water through a surface layer. The model system was applied to an intensely monitored lowland floodplain in Oxford, UK. The model was used to examine water exchange during a large flood event and the duration of flooding was examined through a sensitivity analysis of the surface layer hydraulic conductivity. This modelling study shows that by taking into account low-depth flood waters, flooding can last up to weeks after the main fluvial flood event has passed, with the length of the period being very sensitive to the surface layer hydraulic conductivity. This persistent flooding can have a significant socio-economic impact, for example by restricting vehicle movement through road closures. There is an imperative for flood risk managers to collect evidence to assess the scale of the current and future impact of this type of flooding and to use modelling tools, such as developed in this study, to examine measures to mitigate where there is significant risk.

1. Introduction

The global socio-economic cost of flooding is substantial and increasing rapidly as a result of a higher frequency and severity of flood events, land subsidence and increasing density of vulnerable populations in flood-prone locations (Jongman et al., 2012; Ward et al., 2017; Willner et al., 2017). The characteristics of flooding that are most strongly correlated with impact are frequency, depth, flow and duration (Thieken et al., 2005; Kreibich et al., 2009). In fluvial flood events, much of the water that flows out of river channels onto the associated floodplain will find its way back to the river following the flood peak, as the river level recedes. However, natural levees and anthropogenic changes to topography, particularly in urban areas, may mean that shallow water is retained on the floodplain (Lewin & Ashworth, 2014), extending the period of flooding (McMillan and Brasington, 2007). In this context, Moftakhari et al. (2018) define 'nuisance flooding' as shallow depths of flood water of 3-10 cm, that do not pose significant threat to public safety or cause major property damage but do impact transport and public health. This impact may be caused by direct contact with flood waters but may also be indirect, for example by blocking the road network (Hammond et al., 2015). Moftakhari et al. (2017) showed that the cumulative exposure to frequent, relatively small flood events could cause a greater economic impact than exposure to less frequent extreme events.

In urban areas, engineered drainage may help remove flood waters (Hibbs & Sharp, 2012). Where drainage is absent, the recession of the flood waters will be influenced by the hydraulic conductivity of the underlying floodplain sediments. Land development that has occurred in recent human history, such as the stripping of natural vegetation to allow agriculture, has contributed to higher loads of fine sediments within rivers globally (Walling & Fang, 2003; Macklin et al., 2010). The deposition of this material downstream has produced floodplain sediments that commonly have a low hydraulic conductivity surface layer (alluvium), often contrasting with the substantially more permeable

underlying deposits (Bricker & Bloomfield, 2014). Alluvium hydraulic conductivity can vary greatly depending on the grain size distribution, the degree of compaction and the amount of organic matter (Bridge, 2009).

In addition to the hydraulic conductivity of the shallow floodplain sediments, the recession of water trapped there by the topography may be limited by saturated ground associated with high groundwater levels. In highly permeable floodplains, within the relatively short timescale of a flood event, a significant rise in groundwater level can occur as a pressure response to raised river levels. This pressure response is determined by the diffusivity of the aquifer (Pinder et al., 1969) and hence is greater where confined conditions occur. Although river recession after flood events will cause groundwater levels to decline, these levels are always relatively shallow in floodplain environments.

Relatively few surface-groundwater interaction studies have examined the dynamics between inundated floodplains and shallow groundwater. As a consequence the physics of the hydrological processes in areas with shallow groundwater is not well understood, and typically it is not included in river or groundwater models (Doble et al., 2012). Intensive monitoring and modelling of short sections of rivers (Bates et al., 2000; Burt et al., 2002; Claxton et al., 2003) has shown that in such settings floodplain hydrology is predominately a two-dimensional (lateral) process and that down reach flow effects are only significant at the beginning and end of flood events. Where permeable sediments underlie the floodplain, the river stage is likely to be the principal driver of water table fluctuations (e.g. Jung et al., 2004; Lewandowski et al., 2009). Doble et al. (2012) showed that the hydraulic conductivity of the river bed and the surface layer of the floodplain is important by undertaking a sensitivity analysis on a 2D vertical slice model of a conceptualised floodplain-aquifer system perpendicular to a river. They also showed that irregularities in the floodplain elevation were found to have a large impact on the infiltration volume, but given the structure of the model, did not represent flow across the floodplain in 2D. Further work is required to investigate the spatio-temporal variability of vertical floodplain-aquifer fluxes in more complex settings, for example, with irregular floodplain topography in two dimensions and adjacent to non-linear channels.

Complex, distributed models with detailed physics-based process representations, such as Mike-SHE, have typically been used to simulate variations in groundwater levels in two dimensions across a floodplain, and their interaction with channels and the land surface (Bernard-Jannin et al., 2016; Clilverd et al., 2016; House et al., 2016; Thompson et al., 2017). However, such codes are not generally used by the flood forecasting community (Teng et al., 2017; Jain et al., 2018), given their highly parameterised nature and relatively long run-times, which limits their use in probabilistic ensemble forecasting frameworks. For example, Bernard-Jannin et al. (2016) undertook a comprehensive study of surface water-groundwater exchanges during overbank flood events by coupling 2D modelling of surface flow with the Saint-Venant equation, and 3D modelling of unsaturated groundwater flow with the Richard's equation. One finding from the modelling of this unconfined system was that significant differences in exchanges occurred in response to floods of different magnitude. However, the distributed model was computationally expensive and could only be applied at the reach scale for a short time. By aggregating 1 m LIDAR topographic data onto a 25 m grid they reduced model run-times but found that they incorrectly defined levees and introduced errors into the simulation of the flood extent. Other small structures, such as dykes and ditches, would also need to be represented at a higher resolution to improve the accuracy of the simulations of flood extent and floodplain-aquifer interaction.

Conversely, groundwater fluxes are often neglected, or poorly represented, in reduced-complexity models, using kinematic or diffusive wave approximations, which are widely-used to simulate inundated floodplains (see Teng et al., 2017 for a review), such as FloodMap (Yu & Lane, 2006), JFLOW (Bradbrook, 2006), and LISFLOOD-FP (Bates & De Roo, 2000). For example, in modelling groundwater flooding caused by the groundwater table rising to the land surface in response to rainfall recharge, Morris et al. (2018) calculated groundwater discharge rates along a chalk valley using a simple Darcian calculation. These estimated flows drove their JFLOW flood inundation model but were calibrated

during the modelling process by comparing simulated floodplain flows against targeted spot flow measurements at observed points of groundwater emergence and at a point near the perennial head of the river downstream, which were taken during the flood event. Such an approach does not allow for predictive simulation, and for this reason, one objective of this study was to integrate a groundwater model with a cellular flood inundation model to be able to simulate floodplain aquifer interaction.

The overall aim of this research was to investigate the dynamics of interacting groundwater and flood water within a case study floodplain, to determine the influence that a low hydraulic conductivity surface layer and shallow groundwater have on flood duration. The study used the peri-urban floodplain of the River Thames within the city of Oxford, UK, an area within which the hydrogeology is well-characterised and there is a comprehensive hydrological monitoring network (Macdonald et al., 2018a). The study focussed on a major flood event on the River Thames, which occurred in July 2007. Observed river, floodplain and groundwater level data and aerial photography of the extent of flooding, provided a means to advance understanding of the response of the system during the flood event. Furthermore, these data were used to assess the performance of a new linked groundwater and flood inundation modelling code, developed by integrating the popular cellular, or 'storage cell', flood inundation model, LISFLOOD-FP, with a finite-difference groundwater model. The coupled model was then used to estimate vertical fluxes across the floodplain and to test the effect of different model parameterisations on flood persistence. The study was used to provide generic insights into the potential importance of the subsurface in increasing the impact of flooding.

2. Materials and methods

2.1 Study area

The urban and peri-urban floodplain of the River Thames at Oxford is a setting where the modification of the land surface and construction of infrastructure has produced complex flooding patterns and altered surface water-groundwater interactions (Macdonald et al., 2018a). The floodplain has an area of approximately 20.4 km², ranging in width from 410 to 2170 m (Figure 1). The valley of the Thames in the Oxford area is bounded by high ground to the west and east. Floodplain sediments in the incised river valley are almost entirely contained within impermeable clay bedrock (Newell, 2008).

The long-term mean flow in the River Thames, measured at the upstream end of the Oxford valley, is 18 m³s⁻¹ (Marsh & Hannaford, 2008). Within the valley the Thames splits into a series of channels which then rejoin at the downstream end. The confluence of the River Cherwell and the River Thames is within the Oxford valley. The water levels in the main River Thames channel are managed by a series of locks installed to increase river depths for navigation (Figure 1). These locks influence groundwater levels, creating complex flow patterns (Macdonald et al., 2018)

In Oxford, key transport routes and approximately 3400 properties are located on the Thames floodplain. These have had serious impacts from recent major floods in April 1998, December 2000, January 2003, July 2007, November 2012 and January 2014, involving both fluvial and groundwater flooding (Macdonald et al., 2012). Areas of flood water persist within parts of the city, days after river levels have receded.

This study focussed on Port Meadow (Figures 1 and 2), a flood-prone area of ground owned by the local authority. The meadow is covered primarily by short grass and is used for grazing of horses and cattle. It has environmental protection due to a rare plant species, and is designated as a historical monument due to archaeological features. Port Meadow was chosen due to its topography, as water is retained on the meadow for extended periods after flooding, and it has a good water level monitoring network.

The River Thames runs along the western boundary of Port Meadow (Figure 2a). A stream, which branches off the Thames towards the east, borders the south of the meadow. A series of shallow

channels, of less than a metre in depth, cross the meadow in the northern part and also run along the eastern edges (Figure 2a). High ground immediately to the west of the River Thames stops fluvial flood waters flowing in that direction. To the east of Port Meadow, a railway line and a closed, licensed waste dump form high ground. Within Port Meadow in the south is a further historical waste dump (indicated in Figure 2b by high ground along the southern boundary of the model domain). Together these features act to contain the majority of flood waters. A levee on the eastern bank of the Thames acts to retain flood waters on the floodplain after river levels have returned to within-bank. The ground across the main area of the meadow slopes from north to south with an average gradient of 0.053%. A digital elevation model (DEM) was available for the study area derived from a 2 m LIDAR survey (Cracknell & Hayes, 2007).



Figure 1 Superficial geology of the valley of the River Thames in the vicinity of the city of Oxford, UK. Areas in white are higher ground underlain by bedrock. The modelling study area is indicated, along with the full domain of the floodplain gravels groundwater model and surface water monitoring sites used in this model. Contains Ordnance Survey data © Crown copyright and database right (2018).



Figure 2 Port Meadow modelling study area showing a) basemap, linked model extent, relevant monitoring sites, waste dumps, and the areas of the northern part of Port Meadow that were not flooded, based on aerial photography taken by the British Geological Survey at 12:40 on 24 July 2007; b) elevation; and c) thickness of model surface layer (n.b. this layer includes made ground, hence the large thickness coincident with the waste dumps). Contains Ordnance Survey data © Crown copyright and database right (2018). Contains Environment Agency data licensed under the Open Government Licence v3.0.

The floodplain is underlain by alluvial sediments, a shallow layer of silty clay alluvium above sands and calcareous limestone gravels. In the three-dimensional (3D) geological model for the Oxford valley developed by Newell (2008), the thicknesses in the Port Meadow area range from 0.3 to 2.7 m for the alluvium (Figure 2c; n.b. this map also includes made ground) and 2.0 to 8.1 m for the sands and gravels. Hydraulic conductivities for the sands and gravels layer in the northern zone of the Oxford valley, including Port Meadow, were previously estimated from pumping and packer tests, and found to be in the range 100 to 1000 m d⁻¹ (Dixon, 2004), whilst estimates based on grain size distribution measurements from sediments collected during the drilling of boreholes within Port Meadow gave a range of 200 to 600 m d⁻¹ (Gooddy et al., 2014). In contrast, hydraulic conductivities for the alluvium estimated in studies undertaken in the Oxford valley are substantially lower, ranging from 0.03 to 0.3 m d⁻¹ (Gardner, 1991; Gowing & Youngs, 2005; Hodgson, 2008).

Studies within the Oxford floodplain that have addressed the status of terrestrial ecosystems (Institute of Hydrology, 1987; Gowing & Youngs, 2005), flood risk management (Macdonald et al., 2012) and

pollution from historical waste dumps (Gooddy et al., 2014), have resulted in an extensive water level monitoring network, both for surface and groundwaters (Figure 2a). Surface water levels are monitored in the River Thames at the downstream (S1) and upstream (S2) ends of the reach adjacent to Port Meadow, as well as an additional 24 sites across the Oxford floodplain (Figure 1). Nineteen monitoring boreholes (Figure 2a) located within Port Meadow, drilled over the period 2005 to 2010, were previously used to construct groundwater level contours for the area (Macdonald et al., 2018a). Of the 19 boreholes, 14 were equipped with digital water level recorders and, of these, seven, mostly located in the south of Port Meadow, were available for the full period used in the model calibration and the July 2007 flood event. The three boreholes that were used in this study for comparison with simulated groundwater levels produced from the linked model, P1, P2 and P3, are shown in Figure 2a; this subset was chosen to provide a more evenly distributed coverage of boreholes.

The boreholes were drilled at least 1 m into the sands and gravels beneath the alluvium and screened over the bottom 0.5 m. In addition, a hole augered to a few tens of centimetres into the alluvium, A1, was paired with gravel borehole P1. It was considered that when the water levels measured in A1 were above ground level these would be the same as the flood levels on the floodplain at this location. Within the study area, datums for the boreholes and augered sites and S1 were measured using a GPS system accurate to 3 cm. The ground levels at P1, P2 and P3 are 56.82, 57.05 and 57.59 m asl, respectively. Water level data were measured every 15 minutes at monitoring sites P1, P2, P3, A1, S1 and S2 using digital pressure loggers and compensated for atmospheric pressure using a barometric pressure logger located close to S2; both had an accuracy of 0.5 cm.

Data from the gravel monitoring sites show that for the majority of time the groundwater head fluctuates within the alluvium. Calculations of storage coefficient reported by Dixon (2004) for the Oxford valley indicate the gravel aquifer is semi-confined by the alluvium. Contours of groundwater levels across Port Meadow show an overall gradient, outside of periods of heavy rainfall, of 0.064% north-east to south-west, towards the River Thames (Macdonald et al., 2018a), implying that there is some recharge to the floodplain aquifer from higher terrace gravels to the east. To the west of the River Thames the water level gradient continues from north-east to south-west towards the Seacourt Stream, which gains groundwater in this area (Macdonald et al., 2018a).

2.2 The July 2007 flood event

Many parts of England and Wales experienced severe flooding during June and July 2007 due to extreme rainfall. Between May and July most of southern Britain registered more than twice the 1961-1990 average rainfall as a series of deep anticyclonic weather systems moved across the country (Marsh, 2008). Extreme rain fell on July 19 and 20 in the headwaters of the River Thames (e.g. 140.1 mm in 24 h in Chastleton, Oxfordshire; Marsh & Hannaford, 2007). Peak river flow in Oxford occurred approximately 5 days after the rainfall event (225 m³s⁻¹ at the lock furthest downstream within the Oxford reach of the Thames). Heavy rainfall in Oxford was 60.6 mm in total and limited to July 20. The start of the flood event in Oxford, for the purposes of this study, has been defined as 03:00 on 20 July 2007, the approximate time when the high intensity rainfall began. River levels monitored at S1 (Figure 3) and S2 began to rise within hours of the start of the rainfall at a rate of ~4 cm hr⁻¹. There were three peaks in the river level during the main event, which were due to differences in rainfall across sub-catchments of the River Thames and the timing of their contributions to flow at Oxford. The highest river level at S1, which occurred at 07:00 on 25 July 2007, was 1.22 m above the pre-event level. It took 40 days to return to the pre-event level, however there was a period of six days of rainfall in mid-August when 37.6 mm fell and river levels rose again by approximately 0.2 m.

Groundwater levels started to rise within a few hours of the start of the rainfall event (Figure 3). The groundwater levels at P1, P2 and P3 were 0.13, 0.36 and 0.76 m below ground level, respectively, prior to the event, and took 9, 6 and 38 hours to reach the ground surface. The drilling log for P3 shows the depth of the alluvium as 0.5 m, so the aquifer prior to the flood event was not confined (Figure 3); it took 8 hours from the start of the flood event to become confined. After the intense rainfall ended at

14:00 on 20 July 2007, groundwater levels quickly stabilised, or in the case of P3 decreased, but subsequently began to rise again following a similar pattern to that of the river. The groundwater levels at P1, P2 and P3 peaked at 0.85, 0.78 and 0.22 m, respectively, above ground level, 11, 5 and 7 hours after the time of the river level maximum at S1. The groundwater levels at P1, P2 and P3 declined to below ground on the 28 July, 12 August and 31 July 2007, respectively, 39, 64 and 11 days after the start of the event. Of note in the groundwater hydrographs were: the fast recession at P3 following rainfall, likely indicating a local groundwater discharge zone; and the diurnal fluctuations during the recession at P2 and P3, once groundwater fell below ground level and grass was able to transpire. The smaller diurnal fluctuations that can be seen in the surface water hydrographs are thought to be due to temperature effects on the barometric pressure loggers, as have been seen in other types of environmental sensors (Verhoef et al., 2006).

The monitoring data from A1 showed water appearing on the ground surface at the same time as the groundwater level at P1 rose above ground level (Figure 3a). The surface water level was above the groundwater level for almost all of the event, apart from an early period during the main event recession when the levels were coincident. The maximum depth of surface water at A1 during the event was 0.98 m. Surface water did not disappear at A1 until 22 September, 64 days after the start of the event and 25 days after the groundwater level at P1 had dropped below ground level.

In addition to the observational measurements, aerial photographs of the flooding of the northern half of Port Meadow were taken by the British Geological Survey at 12:40 on 24 July 2007. The areas that were not flooded at this time are shown by the grey polygons in Figure 2a.

2.3 Model codes and linkage

To simulate groundwater flow, floodplain inundation, and the interaction between groundwater and surface water we integrated a finite difference groundwater flow code with a 'storage cell' surface water model that solves for floodplain flows and levels.

2.3.1 Groundwater model

To simulate groundwater flow we use the ZOOMQ3D code (Jackson & Spink, 2004). This uses an implicit finite-difference scheme to solve the governing equation of groundwater flow on a threedimensional Cartesian grid, which can be locally refined horizontally to increase resolution. The code simulates confined and unconfined, heterogeneous and anisotropic aquifers. River-aquifer interaction is simulated using a Darcian head-dependent flux based on the difference between groundwater head and river stage and is calculated as:

$$Q_z = \frac{Kr}{B} W L \left(h_a - h_r \right) \tag{1}$$

where Kr is the vertical hydraulic conductivity of the river bed (m day⁻¹), *B* is the thickness of the river bed (m), *W* is width of the river (m), *L* is length of the river reach (m), h_a is the groundwater head (m) and h_r is the river stage (m). Equation 1 is modified for 'free drainage' when the groundwater level falls below the base of the river bed. In this case the driving head is the difference between the river stage and the base of the river bed. The river stage can either be calculated from the simulated flow through the specification of a stage-discharge rating equation or rating table, or it can be specified over time.



Figure 3 Observed groundwater and surface water levels for the Port Meadow area, as well as 15minute rainfall data from Eynsham Lock in the north west of the Oxford floodplain, for the period 19 July to 26 September 2007. River level at S1 is plotted in each chart for comparison with the groundwater levels. Contains Environment Agency data licensed under the Open Government Licence v3.0.

2.3.2 Flood inundation model

To simulate flow over the floodplain, we use the method implemented in the cellular, or 'storage cell', model of Bates et al. (2010). This is one of the set of numerical solutions used in the widely-applied LISFLOOD-FP floodplain inundation modelling code (Bates & De Roo, 2000). Storage cell models have been developed as an alternative to models that solve the full shallow water equations in two dimensions, which can involve a significant computational overhead. This reduced complexity approach discretises the floodplain into cells and calculates the flux of water in each Cartesian direction analytically, potentially reducing the computational overhead to lower than that of equivalent numerical solutions of the full shallow water equations. They take advantage of the fact that flows over floodplains are typically slow and shallow, and gradients of the local free surface are very small. Consequently, the inertial terms of the governing equations of de Saint-Venant (Chow, 1988) can be neglected. However, Bates et al. (2010) incorporated a simple inertial term into the formulation of inter-cell fluxes to improve the stability of such schemes and their range of applicability. LISFLOOD-FP has been used to simulate flood inundation at high resolutions within complex urban settings (Neal et al., 2011; Sampson et al., 2012), across large areas (Schumann et al., 2013; Dottori et al., 2016), and over decadal to centennial time-scales (Coulthard et al., 2013; Barkwith et al., 2015; Liu & Coulthard, 2017). Considering that groundwater flood events can last from weeks to months (Habets et al., 2010; Hughes et al., 2011; Macdonald et al., 2012; Gotkowitz et al., 2014; Ascott et al., 2017), alongside the functionality, simplicity, and applicability of the storage-cell model of Bates et al. (2010), we selected it as the approach to adopt in this study; the numerical solution was coded into ZOOMQ3D in C++.

2.3.3 Flood water-groundwater interaction

Flood water-groundwater interaction is modelled using the conductance concept, in which there is an exchange flux of water through an interface connecting the two domains (Kollet & Maxwell, 2006; Anderson et al., 2015). This is simulated using a similar but modified version of Equation 1, again limiting the maximum flux to represent free drainage. Typically the width of floodplain model cells is less than 10 m, and as groundwater model cells are one to two orders of magnitude greater, multiple DEM cells will be associated with one groundwater solution point. Consequently, the total flow between the floodplain and aquifer is calculated by:

$$Q_{z} = \sum_{i=1}^{m} \sum_{j=1}^{n} \frac{K_{z}^{ij}}{B^{ij}} \Delta l^{2} \left(h_{a} - h_{s}^{ij} \right)$$
(2)

where Δl is the width of the square DEM cells, m and n are the number of DEM cells in the x and y directions covered by the square or rectangular groundwater model grid cell, and h_s^{ij} is the surface water elevation at DEM cell i, j. The B and K_z terms are considered to be the thickness and hydraulic conductivity of the surface layer controlling the vertical exchange, data for which are specified at the same resolution as that of the DEM, hence their ij superscripts. Rather than incorporate this equation into the implicit finite-difference groundwater solution, to reduce computational cost, this flux is calculated explicitly at the start of a groundwater model time-step, t^0 ; flood infiltration or groundwater head at the end of the time-step, t^1 , is calculated. The surface water level at time t^0 is adjusted, and then the surface water model is run to t^1 , using its multiple, much shorter adaptive time-steps.

2.4 Model structure

2.4.1 Groundwater model structure

The ZOOMQ3D groundwater flow model of Oxford, the boundary of which is shown in Figure 1, was developed by Macdonald et al. (2018b). The extent of the model is limited to the floodplain alluvial sands and gravels and does not include the alluvium above, as its very low hydraulic conductivity contributes little to lateral groundwater flow, or higher terrace gravels, as these are not generally in

direct hydraulic contact with the floodplain sediments. The floodplain sand and gravel aquifer was modelled as a single, homogeneous layer as there was insufficient spatial information that could be used to sub-divide it. The model mesh is regular horizontally and composed of 25 m square cells. The top and base of the sand and gravel layer were obtained from the 3D geological model developed by Newell (2008).

Rainfall recharge to the model was calculated on a daily basis using a soil moisture balance model (Mansour & Hughes, 2004). Boundary conditions at the edge of the model are no-flow, except at three locations where the flows are specified. This includes two sections across narrow widths of the floodplain, perpendicular to the course of the Rivers Thames and Cherwell where they enter the model domain, with modelled boundary groundwater inflows of 420 and 90 m³ d⁻¹, respectively. These are calculated using Darcy's Law with groundwater hydraulic gradients parallel to the river taken from hand-contoured groundwater levels from May 2007, aquifer thicknesses from the 3D geological model, and a hydraulic conductivity of 1000 m d⁻¹. The third location is a section adjacent to the higher terrace gravels underlying the city centre (see Figure 1; Macdonald et al., 2018b), with the total boundary flow equal to the modelled long-term average groundwater recharge over the higher terrace gravels in this area, distributed evenly over the relevant boundary nodes in space and time.

All rivers shown in Figure 1 are included in the model. The river node widths were averages for each river and stream within the model domain estimated from topographic maps, and the river bed thickness was a constant 1 m for all nodes. A single river bed hydraulic conductivity was assigned to each of two categories of water course, the main River Thames and other water courses; these were calibration parameters. These two categories were created as the River Thames is effluent over much of the model domain (Macdonald et al., 2018a), , due to the influence of locks, which is likely to result in lower river bed conductivities (Naganna et al., 2017), while the other water courses are mainly gaining. Observed river levels were imposed on model river nodes, using data from digital loggers at 22 monitoring stations within the Oxford floodplain (Figure 1); river levels between these locations were linearly interpolated.

2.4.2 Flood inundation model structure

The extent of the flood inundation model constructed for the Port Meadow area is shown in Figure 2. The features that form the boundaries are: to the west, the eastern edge of the channel of the River Thames; to the north, an urban area and road; to the east, a railway embankment; and to the south, the high ground associated with a historical waste dump. The ZOOMQ3D model domain within the linked model extends well beyond the boundary of the LISFLOOD-FP model, to areas of the floodplain that are also inundated with flood waters. The assumption is made that the exclusion of the interaction between fluvial flood waters and the floodplain aquifer in these areas does not influence fluxes within the LISFLOOD-FP model boundary.

The ground surface was defined using a 5 m DEM derived by averaging the 2 m DEM. This aggregation of the DEM was necessary to reduce run times to practicable lengths for the model sensitivity testing. Time-variant water levels were imposed on the River Thames channel cells in the west for the period 17 July to 30 September 2007, and updated every 15 minutes throughout the simulated period. These water levels were consistent with those specified at the corresponding river nodes of the groundwater model, and calculated using linear interpolation between measurements from monitoring stations S1 and S2 (Figure 2a). The River Thames was the only significant source and sink of water, to and from the flood inundation model domain.

The thickness of the floodplain alluvium was taken from the 3D geological model (Newell, 2008), interpolated onto a 5 m grid (Figure 2c), and used to define the surface layer. Flood water-groundwater exchange fluxes were calculated at the beginning of each 15 minute groundwater model time-step, and applied uniformly over the multiple, shorter adaptive time-steps of the flood inundation model. Spatial variability in the vertical hydraulic conductivity is anticipated (Hester et al., 2016), however it was not possible to map this within the study, and a single value was used across

the model domain. It is anticipated that zones of relatively high surface layer hydraulic conductivity may exist and these could be significant for flood water recession; the sensitivity analysis on this parameter helped assess the significance.

2.5 Model application

The simulation of the 76-day period from 17 July 2007 using the linked model took 13 hours to run on a PC; the code does not yet take advantage of parallel computing. This limited possible approaches to exploring the model parameter space. Consequently, a two-stage approach was taken to calibrating the linked model. In stage one, the unlinked groundwater model was calibrated with a Monte Carlo process using monthly data over a period of eight years. In stage two, the parameters identified through the unlinked groundwater model calibration provided the starting point for further exploration of the linked model parameter space using the July 2007 flood event, and a sensitivity analysis focussing on the surface layer hydraulic conductivity.

2.5.1 Groundwater model calibration

The groundwater flow model of the Oxford floodplain sands and gravels was originally developed to provide a tool to investigate groundwater flow patterns and groundwater level responses to extreme river flows, in the context of groundwater flooding at a location 2 km downstream of the Port Meadow study area (Macdonald et al., 2018b). As a result, less attention had been paid to the fit of the model to the observations across Port Meadow. The calibration of the unlinked groundwater model was performed using an ensemble of 1000 simulations, varying the sand and gravel aquifer hydraulic conductivity and specific storage, and the river bed hydraulic conductivity. Each run simulated the period January 2004 to December 2012, and was evaluated against observations from January 2006 onwards to remove the effect of potentially biased initial conditions. This period was chosen to be long enough to cover a range of wet and dry years and to maximise the availability of groundwater level time-series data. The time-varying stage of the River Thames and other water courses were imposed on the model river nodes by interpolating monthly median values of observed levels at monitoring points along the channels. The model was run using mean monthly recharge rates and four time-steps per month. Parameter values were sampled uniformly from the ranges specified in Table 1, which also lists the justifications for these ranges; these include references to field studies within the Oxford floodplain (see Section 2.1) and the River Thames Basin, as well as relevant peer-reviewed literature. The calibration focussed on piezometers P1, P2 and P3 for which the root mean square errors (RMSE) between the simulated and observed monthly median groundwater levels were calculated. The groundwater model took approximately 15 minutes to run.

2.5.2 Model application to the July 2007 flood event

The linked groundwater and flood inundation model system was used to investigate river-floodplainaquifer interaction and the implications for flood duration. The model runs simulated the period 17 July to 30 September 2007 using a 15 minute time-step for the groundwater model; the flood inundation model uses much shorter adaptive time-steps, as determined by the stability condition of its explicit scheme (see Bates et al., 2010).

The linked model parameter space was explored based on a combination of the results of the unlinked groundwater model calibration (Section 2.5.1), and the parameter ranges specified in Table 1 for the Manning's roughness coefficient and surface layer hydraulic conductivity. The calibrated groundwater model parameters were varied as follows: 50% higher and lower aquifer hydraulic conductivity; specific storage an order of magnitude higher and lower; and 50% higher and lower river bed hydraulic conductivity. The performance of the linked model was assessed against: the aerial photography of the flooding; groundwater levels recorded in piezometers P1-3; and the surface water level recorded at A1.

Two sets of model parameters were then used to examine the persistence of flooding on the floodplain, and the related dynamics of flood infiltration and groundwater discharge through the

alluvium surface layer. In addition, simulations were performed in which the groundwater level was artificially held at a low level (10 cm above the base of the gravel aquifer) to assess flood persistence under conditions where shallow groundwater has no influence, as infiltration of flood water through the alluvium could occur at the free-drainage rate.

Model component	Model parameter	Range used in calibration	Units	Justification
Groundwater	Aquifer hydraulic conductivity	50 - 1500	m d ⁻¹	Encompasses the range stated by Dixon (2004) for the Oxford
	Aquifer specific storage	10 ⁻⁵ - 10 ⁻²	m ⁻¹	floodplain gravels (see Section 2.1)
	Bed hydraulic conductivity - River Thames	0.05 – 20	m d ⁻¹	Encompasses values reported for River Thames in Younger et al. (1993) and Calver (2001), and in the review by Naganna et al. (2017)
	Bed hydraulic conductivity - other water courses	0.5 – 200	m d ⁻¹	An order of magnitude greater than River Thames as these water courses are generally gaining groundwater
Flood inundation	Manning's roughness coefficient	0.025 – 0.05	m ^{1/3} s ⁻¹	Based on plausible range for grassland/pasture from Werner et al. (2005).
Surface layer	Surface layer vertical hydraulic conductivity	0.01 – 0.5	m d ⁻¹	Encompasses the range of values measured in Oxford floodplain (see Section 2.1) and consistent with Doble et al. (2012)

 Table 1
 Parameters of the linked model system and the range within which they were varied in the calibration process and sensitivity analysis

3. Results

In the description of the observations and simulations of the July 2007 event, the period from the start of the event up to 5 August, at the end of the steep recession, is referred to as the main flood event; the subsequent period is referred to as the long recession (Figure 3).

3.1 Groundwater model calibration

The RMSE calculated between simulated and observed monthly median groundwater levels from boreholes P1, P2 and P3 for the 1000 Monte Carlo runs of the groundwater model are plotted in Figure 4. This shows that the only parameters within this model structure that have an identifiable control on model performance at the three piezometers are the hydraulic conductivities of the gravel aquifer and the bed of the River Thames. The average of the RMSEs for P1, P2 and P3 reduces as the aquifer hydraulic conductivity increases. Based on this, the value was set to 1000 m d⁻¹ as this was the highest value within the range identified by Dixon (2004). The bed hydraulic conductivity of the River Thames was set to 10 m d⁻¹ for the same reason, as this was the highest value identified for the River Thames in studies referenced by Calver (2001). Specific storage was not identifiable because the driving river level boundary conditions, which propagate rapidly through the very high-diffusivity aquifer, are the dominant control on the groundwater levels. A value of specific storage of 0.001 was initially chosen based on the results for south Oxford in Macdonald et al. (2018b). As the simulated groundwater levels at P1, P2 and P3 were not dependent on the hydraulic conductivity of the bed of the other rivers, this was set to the same value as that of the bed of the River Thames.

Using these parameters, the RMSE values over the 2006 to 2012 period for P1, P2 and P3 were 12, 15 and 18 cm respectively. Empirical cumulative distribution functions of simulated and observed mean monthly groundwater levels for the three piezometers are plotted in Figure 5. The simulated maxima over the period were 3 cm low at P1, and 16 and 19 cm high at P2 and P3, respectively, which was

deemed acceptable considering that the unlinked groundwater model does not simulate fluxes between the aquifer and the floodplain. The lowest groundwater levels were simulated to be too high. We believe this is due to the inability of the single-layer groundwater model to represent groundwater flow west from Port Meadow under the River Thames to the Seacourt Stream, when levels are very low (Macdonald et al., 2018a). This could potentially occur if the thickness and hydraulic conductivity of the river bed sediments means the hydraulic connection between the river and aquifer is relatively poor at low river water levels, as has been reported in other studies (Doppler et al., 2007). However, groundwater levels observed in P1, P2 and P3 at the end of September 2007 are above 56.6 m asl, and as such are captured acceptably by the model.



Figure 4 Individual and averaged root mean square error between simulated and observed monthly median groundwater levels from boreholes P1, P2 and P3 for the 1000 Monte Carlo runs of the ZOOMQ3D groundwater flow model for the Oxford gravel aquifer model over the period 2006 to 2012.



Figure 5 Cumulative distribution functions of median monthly groundwater levels over the period 2006 to 2012, from boreholes P1, P2 and P3: observed, and simulated using the ZOOMQ3D groundwater flow model for the Oxford gravel aquifer model with selected parameters identified through model calibration.

3.2 Simulation of the 2007 flood event

3.2.1 Comparison of observed and simulated water levels

Given the long run times of the linked model system, a Monte Carlo approach to assessing model performance was not possible. The linked model started with the groundwater model parameters identified from the model calibration in Section 3.1, and varied these parameters, along with Manning's roughness coefficient, n, and surface layer hydraulic conductivity, K_z , with the approach set out in Section 2.5.2. The model performance was assessed by varying the values of one parameter at a time and calculating the RMSE at P1, P2, P3 and A1. The results of this exploration of the parameter space are tabulated in Table S1 in Supplementary Information (SI).

Very little change was seen in RMSEs as a result of varying K_z (with $n \ 0.025 \ m^{1/3} \ s^{-1}$). The RMSEs were in the ranges 7-8 cm, 7-10 cm, 15-19 cm and 5-6 cm for P1, P2, P3 and A1 respectively, with a K_z of 0.3 m d⁻¹ giving the best average fit for the four monitoring sites, 8 cm (Figure 6a). A similar lack of variation in results occurred when the specific storage, *Ss*, was varied; here, two values of K_z were used, 0.03 and 0.3 m d⁻¹. RMSEs for P1, P2, P3 and A1 were in the ranges 6-7 cm, 8-10 cm, 15-19 cm and 5-6 cm, and 6-7 cm, 7-10 cm, 15-17 cm and 5 cm, for the K_z values of 0.03 and 0.3 m d⁻¹, respectively. The best average fit, 8 cm, was achieved with *Ss* of 0.001 m⁻¹ and K_z of 0.3 m d⁻¹ (Figure 6b).

The largest RMSE was obtained with the variation in the aquifer hydraulic conductivity, K. The model was run with values of K of 500, 1000 and 1350 m d⁻¹, with Ss 0.001 m⁻¹, K_z 0.3 m d⁻¹ and n 0.025 m^{1/3} s⁻¹

¹. RMSEs were in the ranges 7-16 cm, 7-23 cm, 13-39 cm and 5-8 cm for P1, P2, P3 and A1 respectively, however the RMSEs for the two higher values of K were very similar and at the lower end of the range of RMSEs for all sites (Figure 6c). The variations in river bed hydraulic conductivity and *n* made imperceptible differences to the RMSEs calculated for each monitoring site, and were not plotted, although are presented in Table S1.



Figure 6 Average RMSEs for sites P1, P2, P3 and A1 for selected linked model parameter sets: a) K_z varied - $K = 1000 \text{ m} \text{ d}^{-1}$, $Ss = 0.001 \text{ m}^{-1}$, $Kr = 10 \text{ m} \text{ d}^{-1}$, $n = 0.025 \text{ m}^{1/3} \text{ s}^{-1}$; b) Ss varied, with two values of K_z , 0.03 and 0.3 m d⁻¹ - $K = 1000 \text{ m} \text{ d}^{-1}$, $Kr = 10 \text{ m} \text{ d}^{-1}$, $n = 0.025 \text{ m}^{1/3} \text{ s}^{-1}$; c) K varied - $K_z = 0.3 \text{ m} \text{ d}^{-1}$, Ss = 0.001 m⁻¹, $Kr = 10 \text{ m} \text{ d}^{-1}$, $n = 0.025 \text{ m}^{1/3} \text{ s}^{-1}$; c) K varied - $K_z = 0.3 \text{ m} \text{ d}^{-1}$, Ss = 0.001 m⁻¹, $Kr = 10 \text{ m} \text{ d}^{-1}$, $n = 0.025 \text{ m}^{1/3} \text{ s}^{-1}$.



Figure 7 Simulated flood depth (m) on Port Meadow for specified days from the start of the flood event, from model runs using baseline parameters with: a) to d) K_z 0.3 m d⁻¹; and e) K_z 0.03 m d⁻¹. For b) includes areas in northern Port Meadow that were mapped as non-flooded from BGS aerial photographs.

3.2.2 Flood water extent

Based on the model parameter exploration presented in Section 3.2.1, the aquifer parameters identified in the groundwater model calibration in Section 3.1 were again used, along with K_z 0.3 m d⁻¹ and *n* 0.025, to assess the model performance in simulating flood water extent. The output was compared with a model run using K_z 0.03 m d⁻¹ to assess the influence of K_z on flood water persistence and vertical fluxes between aquifer and floodplain. Using the former of the two sets of parameters, the peak simulated flood water levels occurred at 14:30 on 25 July 2007, Day 6 of the flood event, when the observed depth at P1 was 1.05 m. The extent and depth of flooding across Port Meadow is shown in Figure 7a-d, for four times: 1, 5, 17 and 43 days after the start of the flood event. The model on Day 5 (Figure 7b) compares well to the aerial photography at this time, and reproduces the areas not inundated in the north, and the small mound in the centre of the meadow.

Figure 7a shows that at the end of 20 July, the first day of flooding, most of the southern end of the meadow is simulated to be inundated, with a maximum depth of water of 0.63 m. The simulated depth of water at P1 at this time is 0.37 m. By 5 August, Day 15 (Figure 7c), at the end of the steep recession from the main flood event, the southern end of the meadow is still flooded and there are other isolated areas of flooding. By 31 August, Day 43 (Figure 7d), only the drainage network and the southern end of the meadow are simulated to be under water. At this time, groundwater is below ground level at all but a few low-lying locations associated with drains (Figure 8). The effect of varying K_z on the results is only seen in the latter stages of the flood simulation. Decreasing the hydraulic conductivity of the alluvium ten-fold to 0.03 m d⁻¹ increases the area of the floodplain inundated on Day 43 from 1.5%, or 5.78 ha, to 5.5%, or 21.86 ha (Figure 7e).



Figure 8 Depth to groundwater on 31 August 2007, Day 43 of the flood event, simulated by the baseline model using $K_z = 0.03$ m d⁻¹





3.2.2 Groundwater-floodplain fluxes

The simulated total vertical flux of water between the aquifer and the floodplain is shown in Figure 9. In the simulations with values of K_z of 0.03 and 0.3 m d⁻¹, the total discharge from the aquifer to the floodplain over the period of the model run in both cases is greater than the total infiltration; 5.38 x 10^4 m³ compared to 4.52 x 10^4 m³ for K_z of 0.03 m d⁻¹, and 8.48 x 10^4 m³ compared to 6.28 x 10^4 m³ for K_z of 0.3 m d⁻¹. The majority of the discharge occurs within the main flood event in both runs. However, the pattern is different for infiltration: with the lower K_z it occurs almost totally within the long recession; while it occurs both during the main event and the long recession with the higher K_z . Maps of the total flux (Figure 9b, d) show that discharge occurs over most of the floodplain, although higher amounts occur in relatively low-lying areas away from the river channel. Infiltration is focused in the drains and depressions that may be associated with old stream channels, and also in the south of the floodplain where water that drains from the floodplain is trapped.

The linked model estimated flows onto the Port Meadow floodplain during the flood peak of over 200 m³s⁻¹. This is similar to the peak flows in the River Thames in Oxford during the event, at the lock furthest downstream. In comparison, the total net groundwater flow through the river nodes of the groundwater model for the full 76-day period of the simulation, which is predominantly from the aquifer back into the river, was only 820 and 950 m³, respectively for K_z of 0.03 and 0.3 m d⁻¹.

3.2.3 Response at monitoring sites

The sensitivity to variation in K_z of simulated groundwater and surface water levels and vertical fluxes at the three monitoring locations, P1/A1, P2 and P3, was investigated. The observed and simulated water levels are shown in Figure 10. The time-series of the vertical fluxes at these points are presented in Figure 11. First, results are described for each site in turn, and then for the simulations with the groundwater level held artificially low (the 'free-drainage' runs).

At P1, as is the case for the observations, the simulated surface water levels were above the groundwater levels for the majority of the overall event, for every value of K_z . The simulated surface water levels were the same for every K_z until the end of the main event, as the dominant control during this period is overbank flow. The shape of the simulated surface water hydrograph matches that of the observed hydrograph well, although during the main flood event simulated levels are approximately 5 cm too high. The hydraulic conductivity of the surface layer has a substantial effect on the length of the long recession. With $K_z = 0.01 \text{ m d}^{-1}$ water remained on the floodplain at P1 until the end of the simulation. The difference in the duration of flooding for K_z values of 0.3 and 0.03 m d⁻¹ was 18 days, although with the lower 3 cm threshold of nuisance flooding defined by Moftakhari et al. (2018) the difference reduced to 10 days.

Differences between the groundwater level hydrographs during the main event, simulated using the range of K_z values, are small and of the order of 4 cm. The effect of the persistence of floodplain inundation on the groundwater hydrograph is apparent in the long recession. With more permeable alluvium ($K_z = 0.3 \text{ m d}^{-1}$), the groundwater level is initially higher, due to increased infiltration, but then falls below the hydrograph for the lower hydraulic conductivity case ($K_z = 0.03 \text{ m d}^{-1}$).

Figures 11a and b show the time-series of vertical fluxes at P1 from the simulations using K_z of 0.03 and 0.3 m d⁻¹, respectively. As with the observations, in both cases the simulated groundwater level is initially above the surface water level. This results in a short period of groundwater discharge to the floodplain. Subsequently, higher surface water levels drive infiltration until the floodplain dries. Infiltration increases with time as the groundwater level declines at a faster rate than the surface water level.

There are no surface water level observations at P2 but simulated groundwater levels compare well with the observations. With K_z set to 0.03 and 0.3 m d⁻¹, simulated peaks levels are only 9 and 6 cm higher than those observed, respectively. The gradient of the simulated recession is lower than the observed. However, the pattern of groundwater levels rising above and then falling below the ground surface in response to the mid-August rainfall was reproduced.

Groundwater levels are above surface water levels for the full period of each simulation at P2. The simulated groundwater level rises in response to the changing river stage. The rapid transfer of the pressure head through the high diffusivity aquifer resulted in a time lag between observed peak river level and simulated peak groundwater level at P2 of 1.5 and 1 hours, for K_z of 0.03 and 0.3 m d⁻¹ respectively. The groundwater level rises higher than the predicted floodplain surface water level, thus driving groundwater discharge (Figures 11c and d). The driving head difference with K_z 0.3 m d⁻¹ is less than 1 cm, compared to 8 cm with K_z 0.03 m d⁻¹, but results in discharge rates that are four times higher. The simulated surface water levels are the same for each simulation, as at P1, as these are controlled by overbank flows until the river returns to within its banks. There is a separate period

of flooding on the floodplain at P2 associated with the rise of the groundwater level above the ground surface in mid-August (Figures 10b and 11d); during this period the model showed no fluvial flood waters reaching this location.



Figure 10 Observed and simulated water levels at a) P1, b) P2 and c) P3. Simulated water levels from baseline model and from free-draining version of baseline model, with $K_z = 0.03$ m d⁻¹ and 0.3 m d⁻¹

Simulated groundwater levels at P3 do not compare as well as with observations as at other sites (Figure 10c), and the levels are different by approximately 20 cm. This difference is consistent over much of the hydrograph and it is possible, therefore, that at this site there was a mistake with the measurement of the datum. However, the relatively poor match of simulated and observed levels is consistent with the results of the calibrations runs of the groundwater model at this location. The recession of the simulated levels is shallower than that of the observations, and the groundwater level simulations do not replicate the flashy response to the rainfall in mid-August. As for the observations, simulated groundwater levels do not rise above ground surface in response to the mid-August rainfall.

Again, the simulated surface water levels are very similar for all the values of K_z applied, and in all cases are lower than the groundwater levels for the full period of the run. As at P2, all fluxes at this location are from the aquifer to the floodplain (Figures 11e and 11f) and with higher K_z values, the fluxes are substantially higher.

The results of simulations in which the groundwater level was held artificially just above the base of the aquifer, so that any infiltration occurred at the free-drainage rate, are also shown in Figure 10 for values of K_z of 0.3 and 0.03 m d⁻¹. At all three sites, and in both simulations, the surface water levels with and without fluctuating groundwater levels are the same. The influence of groundwater was seen once the river levels recess to within bank. The free draining model at P1 with K_z of 0.03 m d⁻¹ has the surface water disappearing quickly after the main flood event, on 10 August 2007 (Figure 10a), 37 days before water disappears in the model with dynamic groundwater. For this site, with K_z of 0.03 m d⁻¹, the free-draining simulation results in the floodplain drying 5 days earlier than with K_z of 0.03 m d⁻¹.

At P2 and P3, the period that the simulated surface waters remain on the floodplain is not long when compared with P1. With the free-drainage model and K_z of 0.03 m d⁻¹, surface water disappears at P2 and P3 by 5 August and 31 July; that is 13 and 3 days, respectively, before that simulated by the model with dynamic groundwater. With K_z of 0.3 m d⁻¹, the difference in the timing of the disappearance of the surface water between the two models is only 1 day at P2, with no difference at P3.

4. Discussion

4.1 Limitations of the model system and transferability

In assessing the ability of the linked model system to simulate water levels in a flooded floodplain and its transferability to other settings, the model limitations need to be considered.

Groundwater recharge is calculated by the ZOODRM model (Mansour & Hughes, 2004) prior to running the ZOOMQ3D model. This approach introduces inconsistencies where soil water balances are used to calculate recharge in areas that are inundated with flood water. In addition, within the linked model system there are a number of time-variant processes that are not simulated during run time that may be particularly significant when investigating the persistence of flood waters on the floodplain. Rainfall input to areas already flooded is not simulated, nor is evaporation. In very shallow groundwater conditions, evapotranspiration directly from the saturated zone in areas where flood waters have abated may also be significant for the water balance, especially for summer flood events. In the application of the linked model system reported here, ignoring the rainfall input is not considered significant due to the low amounts during the period that flood waters occur on the floodplain. However, it is acknowledged that direct evaporation from flood waters could have resulted in a faster recession of the flood water level, potentially allowing a better fit between modelled and observed water levels with lower surface layer hydraulic conductivities.

The linked model system does not simulate flow within the unsaturated zone. The assumption is that in floodplain environments the total saturation of sediments occurs within a comparable timescale to flood inundation. In the case study floodplain this is particularly so due to the shallow range within which groundwater fluctuates, and the high diffusivities that result in rapid propagation of the flood

wave from the river. In floodplain settings where the sediments are unconfined and less permeable, with a deeper unsaturated zone, the lack of explicit modelling of the unsaturated zone may be problematic, and further examination of the significance of the assumptions of the model in this context is required.



Figure 11 Simulations of water levels and vertical fluxes for P1 (a, b), P2 (c, d) and P3 (e, f) from the baseline model using K_z 0.03 m d⁻¹ and 0.3 m d⁻¹. Vertical fluxes in m³ d⁻¹ over the 5 x 5 m cell containing the observation point.

Although time-varying river flows and levels can be simulated by ZOOMQ3D, in the current version of the code the translation of these onto the linked model DEM has to be prescribed *a priori* as a boundary condition. This has been possible in this study due to the high density of river level monitoring stations. However, for application to other areas where such data are not available, the code needs to be developed to transfer the river level simulated by ZOOMQ3D to LISFLOOD-FP during run-time.

The incorporation of groundwater-floodplain interaction in the finite-difference solution of the groundwater model using an explicit difference term (Wang and Anderson, 1982), where the flux depends on the groundwater and surface water levels at the start of the groundwater time-step, and the differing lengths of the ZOOMQ3D and LISFLOOD-FP time-steps, may introduce some oscillations into these flux time-series. This makes it possible for there to be circumstances when groundwater discharge and floodplain infiltration occur during alternate time steps, as can be seen in Figure 11d, though here the floodplain recharge fluxes are negligible.

The linked model system incorporates the assumptions of the component codes. In particular for LISFLOOD-FP there is the requirement that flows are slow and shallow, and gradients of the local free water level surface are small (Bates et al., 2010).

Crucially, a limitation in the linked model system is that it does not allow the effects of subsurface drainage infrastructure to be simulated. This would likely increase the rate at which retained water is removed from the floodplain. This infrastructure also provides pathways for the movement of flood waters during the onset of flooding. The area of the Oxford floodplain used in this case study is without subsurface drainage and has provided a good test case for the newly developed model system, however, further development of the model system is required to enable this subsurface route for water flow to be incorporated prior to the transfer of the model to urban settings. Other challenges in urban areas include: the mapping of surface layer thickness, for example due to the impact of the construction of building foundations; and the hydraulic conductivity attribution of the surface layer given the mix of spatially complex geology and anthropogenic changes to the land surface. There would also be benefit in using parallel processing to substantially reduce run times, especially in urban areas with complex topography where high resolution DEMs are needed.

However, as the case study described here has shown, the model can be successfully applied to periurban floodplains where there is no subsurface infrastructure but there has been some anthropogenic modification of the floodplain topography. In addition, the model system could be used to explore issues such as: groundwater recharge from flood inundation, a mechanism that has been identified in some settings as a key component of aquifer water balances; spatiotemporal variation in the ingress of polluted surface waters to floodplain aquifers; and the efficacy of some natural flood management measures.

4.2 Fit of model simulations to observations

The linked model system has enabled the dynamics of the July 2007 flood event to be reproduced and provided a good fit to the observed groundwater and surface water level time-series. RMSEs of approximately 7, 7, 15 and 5 cm were achieved at sites P1, P2, P3 and A1, respectively. Within the model parameter space explored, the range of the RMSEs at these sites was limited to a few cm, apart from when the lowest value of aquifer hydraulic conductivity was applied. The combination of the general confined nature of the gravel aquifer, and associated low storage coefficient, with its high hydraulic conductivity produces very high aquifer diffusivity. As a result, groundwater levels across the aquifer, both those observed and modelled, closely followed the rapid rise in levels in the River Thames during the July 2007 flood event. The overriding control that river levels have on groundwater levels was the reason for the limited range of variability in modelled water levels, leading to the equifinality within the model parameter space.

The spatial extent of the flooding simulated by the linked model matched well that derived from aerial photography. However, it is recognised that the confidence in the results produced by the model would have been greater if more observations of surface water level on the floodplain were available. It is also recognised that on flat floodplains, relatively small differences in level can be highly significant. This is particularly so when quantifying the duration of flooding at shallow depths of less than a few tens of centimetres. This highlights the need for accurately measured datums and water levels. Great efforts were made in this study to produce accurate measurements, but the potential for the accumulation of errors in the observed data and the DEM must be recognised when assessing the fits between simulated and observed water levels.

Floodplain sediments can be highly heterogeneous, with associated variation in aquifer parameters. Due to a lack of data at the necessary spatial resolution, in developing the Oxford floodplain model, significant simplications had to be made both for the aquifer parameters of the sands and gravels aquifer and the surface layer alluvium. However, given these uncertainties, the study still provides important insights into the functioning of floodplains during flood events.

4.3 Floodplain water fluxes

As the modelled groundwater levels were generally higher than the levels of the overbanked river water on the meadow, although only by a few cm (Figure 11), groundwater discharge was simulated across much of the floodplain during the main flood event (Figure 9). It is thought that this occurs because groundwater levels are controlled by distal river levels, and overbanked flood waters, once beyond the river levee, are lower than these over much of the floodplain. The simulated groundwater discharge was not substantial; the total net volume discharging over the main flood event was equivalent to average depths of water of 35 and 55 mm, for values of K_z of 0.03 and 0.3 m d⁻¹, respectively. However, the uncertainty in modelled groundwater and surface water levels needs to be taken into account in this context and could mean the reversal of the direction of vertical fluxes during the main flood event, although these fluxes are still likely to be small.

With K_z of 0.03 m d⁻¹, the majority of flood infiltration occurred during the long recession and, if averaged over the whole of the Port Meadow floodplain, was equivalent to a depth of water of 29 mm. This is approximately 25% of the long-term average annual recharge estimated using a soil moisture balance model (Macdonald et al., 2018a). The temporal distribution of flood infiltration from the simulations with K_z of 0.3 m d⁻¹ was different to that with K_z of 0.03 m d⁻¹, with the more permeable alluvial layer allowing infiltration to occur during the main event. Net infiltration during the long recession with K_z of 0.3 m d⁻¹ was equivalent to a depth across the floodplain of 41 mm. However, in comparison to the modelled groundwater discharge that occurred during the flood event, modelled flood infiltration was not widespread, occurring only in areas that are relatively low-lying. The larger water level gradients in these low-lying areas mean uncertainties in modelled water levels may not have been as significant for flux calculations as they were during the main flood event. Macdonald et al. (2018a) identified the infiltrating flood water as providing a significant input of nitrate to the floodplain aquifer; the results of the current study indicate that these high nitrate waters do not infiltrate uniformly across the floodplain but are spatially focussed.

The reconnection of rivers with their floodplains and the construction of levees to retain water upstream of vulnerable urban centres, known as offline storage, is an approach that is used to manage downstream flood risk by lowering flood peaks (Lane, 2017). The consequence of the high groundwater levels during the main flood event at Port Meadow was to reduce the storage height of the floodplain but only by a few centimetres. This compares to the equivalent average storage height of the floodplain at the peak of the July 2007 flood, which was 48 cm. Therefore in this case study, groundwater discharge to the floodplain does not compromise the storage of the floodplain significantly but under other circumstances, groundwater conditions may need to be taken into account in the design of offline storage schemes.
4.4 Flood duration

The study highlights the importance of modifications of the surface topography as a factor in flood duration. In areas to which water drains and is trapped by surrounding high ground, modelling has shown that the groundwater level and the hydraulic conductivity of the underlying sediments are key in controlling the duration that flood waters remain on the ground surface. The slow decline in water level lengthened the period of flooding by days or weeks. Although depths of flood water may be very shallow, other research (Moftakhari et al., 2018) has shown that in urban areas as little as 3 cm of water can cause significant nuisance. The modelling here showed that an order of magnitude difference in the hydraulic conductivity of the floodplain surface sediments resulted in a difference in the period of flooding to the 3 cm threshold of 10 days. In the theoretical case when shallow groundwater was artificially held at a low level, with K_z of 0.03 m d⁻¹, the period of flooding reduced by 32 days compared to the simulation in which groundwater levels were not fixed.

Given the heterogeneity of floodplain sediments, both vertically and laterally, challenges exist in identifying areas where flood waters are likely to persist due to underlying ground conditions. However, with the potential for a greater frequency of flooding, and groundwater levels that are maintained at shallower depths in lowland floodplains by raised levels in the associated rivers, nuisance flooding may become more prevalent. The economic drivers may be strong enough to require this form of flooding to be considered routinely in the modelling of flood hazards. If so, it will be necessary to include processes in standard flood routing models that allow the vertical fluxes at the floodplain to be incorporated.

5. Conclusions

Through this study a groundwater flow model has been linked to a widely-applied floodplain inundation model to investigate the dynamics of interacting groundwater and flood water. This model system was developed recognising the lack of flood inundation models that have a physics-based representation of groundwater, that are able to simulate over periods of weeks to months with sufficiently fast run times to undertake sensitivity analyses. While recognising the limitations of the linked model system and the uncertainties in the simulations, the model application to a lowland periurban floodplain was considered successful. The model set-up, and the hypotheses that this represents in terms of hydrological processes, worked well for the study area when the model was constrained using observed hydraulic conductivities and tested against observed flood water and groundwater levels.

The modelling has shown that the fluxes of water between aquifer and floodplain during a flood event can be highly spatially and temporally variable and that the effective infiltration of flood waters to the aquifer occurred in the long period of flood recession and was focussed in relatively low-lying areas. In locations where natural levees or modifications to the floodplain topography retain flood waters on the floodplain, the modelling highlighted that the duration of flooding is highly sensitive to the hydraulic conductivity of surface geology and the presence of shallow groundwater. This can result in flood waters remaining for weeks at depths that can have significant socio-economic impact.

This research highlights that there is an imperative for flood risk managers to collect evidence to assess the scale of the current and future impact of this type of nuisance flooding and to use modelling tools, such as developed in this study, to examine measures to mitigate where there is significant risk.

Acknowledgements

This study was funded by the UK Natural Environment Research Council (NERC) National Capability resources, devolved to the British Geological Survey. Thanks are due to those staff members of the British Geological Survey (BGS) who contributed to this study: Majdi Mansour, Clive Cartwright, Debbie White, Barry Townsend, Peter Williams, Andy Newell, Dan Lapworth and Daren Gooddy. Thanks are also due to the BGS staff who took aerial photographs of the July 2007 floods.

The authors are grateful to Andy Dixon for his advice on the hydrology of the Oxford floodplain, and to him and the staff of Groundwater Monitoring and Drilling Ltd. for the long-term collection of water level data in the Oxford area. The Environment Agency is thanked for its funding of monitoring infrastructure and data collection in the study area. Topographic data derived from LIDAR were licensed from the Environment Agency under the UK Open Government Licence v3.0.

Macdonald and Jackson publish with the permission of the Executive Director, BGS.

References

- Anderson, M. P., Woessner, W. W., & Hunt, R. J. (2015). Applied groundwater modeling: simulation of flow and advective transport. Academic press.
- Ascott, M. J., Marchant, B. P., Macdonald, D., McKenzie, A. A., & Bloomfield, J. P. (2017). Improved understanding of spatio-temporal controls on regional scale groundwater flooding using hydrograph analysis and impulse response functions. *Hydrological Processes*, *31*(25), 4586-4599. doi:10.1002/hyp.11380
- Barkwith, A., Hurst, M. D., Jackson, C. R., Wang, L., Ellis, M. A., & Coulthard, T. J. (2015). Simulating the influences of groundwater on regional geomorphology using a distributed, dynamic, landscape evolution modelling platform. *Environmental Modelling & Software, 74*, 1-20. doi:10.1016/j.envsoft.2015.09.001
- Bates, P. D., & De Roo, A. P. J. (2000). A simple raster-based model for flood inundation simulation. *Journal of Hydrology, 236*(1-2), 54-77. doi:10.1016/s0022-1694(00)00278-x
- Bates, P. D., Horritt, M. S., & Fewtrell, T. J. (2010). A simple inertial formulation of the shallow water equations for efficient two-dimensional flood inundation modelling. *Journal of Hydrology*, *387*(1-2), 33-45. doi:10.1016/j.jhydrol.2010.03.027
- Bates, P. D., Stewart, M. D., Desitter, A., Anderson, M. G., Renaud, J. P., & Smith, J. A. (2000). Numerical simulation of floodplain hydrology. *Water Resources Research*, 36(9), 2517-2529. doi:10.1029/2000wr900102
- Bernard-Jannin, L., D. Brito, X. L. Sun, E. Jauch, R. Neves, S. Sauvage, and J. M. Sanchez-Perez (2016), Spatially distributed modelling of surface water-groundwater exchanges during overbank flood events - a case study at the Garonne River, *Advances in Water Resources*, 94, 146-159. doi: 10.1016/j.advwatres.2016.05.008
- Bradbrook, K. (2006). JFLOW: a multiscale two-dimensional dynamic flood model. *Water and Environment Journal, 20*(2), 79-86. doi:10.1111/j.1747-6593.2005.00011.x
- Bradbrook, K. F., Lane, S. N., Waller, S. G., & Bates, P. D. (2004). Two dimensional diffusion wave modelling of flood inundation using a simplified channel representation. *International Journal of River Basin Management*, 2(3), 211-223. doi:10.1080/15715124.2004.9635233
- Bricker, S. H., & Bloomfield, J. P. (2014). Controls on the basin-scale distribution of hydraulic conductivity of superficial deposits: a case study from the Thames Basin, UK. *Quarterly Journal of Engineering Geology and Hydrogeology*, *47*(3), 223-236. doi:10.1144/qjegh2013-072
- Bridge, J. S. (2009). *Rivers and floodplains: forms, processes, and sedimentary record*: John Wiley & Sons.
- Burt, T. P., Bates, P. D., Stewart, M. D., Claxton, A. J., Anderson, M. G., & Price, D. A. (2002). Water table fluctuations within the floodplain of the River Severn, England. *Journal of Hydrology*, 262(1-4), 1-20. doi:10.1016/s0022-1694(01)00567-4
- Calver, A. (2001). Riverbed Permeabilities: Information from Pooled Data. *Ground Water, 39*(4), 546-553. doi:10.1111/j.1745-6584.2001.tb02343.x
- Chow, V. T. (1998). *Applied Hydrology*. McGraw-Hill, Inc., New York.
- Claxton, A. J., Bates, P. D., & Cloke, H. L. (2003). Mixing of Hillslope, River, and Alluvial Ground Waters in Lowland Floodplains. *Ground Water*, *41*(7), 926-936. doi:10.1111/j.1745-6584.2003.tb02435.x
- Clilverd, H. M., Thompson, J. R., Heppell, C. M., Sayer, C. D., & Axmacher, J. C. (2016). Coupled Hydrological/Hydraulic Modelling of River Restoration Impacts and Floodplain

Hydrodynamics. *River Research and Applications*, 32(9), 1927-1948. doi:10.1002/rra.3036

- Coulthard, T. J., Neal, J. C., Bates, P. D., Ramirez, J., de Almeida, G. A. M., & Hancock, G. R. (2013). Integrating the LISFLOOD-FP 2D hydrodynamic model with the CAESAR model: implications for modelling landscape evolution. *Earth Surface Processes and Landforms, 38*(15), 1897-1906. doi:10.1002/esp.3478
- Cracknell, A. P. (2007). Introduction to remote sensing: CRC press.
- Dixon, A. (2004). Hydrology of the Oxford Meadows cSAC. Environment Agency Report.
- Doble, R. C., Crosbie, R. S., Smerdon, B. D., Peeters, L., & Cook, F. J. (2012). Groundwater recharge from overbank floods. *Water Resources Research, 48*. doi:10.1029/2011wr011441
- Doppler, T., Franssen, H.-J. H., Kaiser, H.-P., Kuhlman, U., & Stauffer, F. (2007). Field evidence of a dynamic leakage coefficient for modelling river–aquifer interactions. *Journal of Hydrology*, 347(1), 177-187. https://doi.org/10.1016/j.jhydrol.2007.09.017
- Dottori, F., Salamon, P., Bianchi, A., Alfieri, L., Hirpa, F. A., & Feyen, L. (2016). Development and evaluation of a framework for global flood hazard mapping. *Advances in Water Resources, 94*, 87-102. doi:10.1016/j.advwatres.2016.05.002
- Gardner, C. (1991). Water regime of river meadows: Yarnton Mead case study. *Report to MAFF, Institute of Hydrology, Wallingford, UK*.
- Gooddy, D. C., Macdonald, D. M. J., Lapworth, D. J., Bennett, S. A., & Griffiths, K. J. (2014). Nitrogen sources, transport and processing in peri-urban floodplains. *Science of the Total Environment*, 494-495, 28-38. https://doi.org/10.1016/j.scitotenv.2014.06.123
- Gotkowitz, M. B., J. W. Attig, and T. McDermott (2014), Groundwater flood of a river terrace in southwest Wisconsin, USA, *Hydrogeology Journal*, 22(6), 1421-1432. doi: 10.1007/s10040-014-1129-x
- Gowing, D., & Youngs, E. (2005). The requirements of Apium repens–an ecohydrological assessment. English Nature Final Report February.
- Habets, F., Gascoin, S., Korkmaz, S., Thiery, D., Zribi, M., Amraoui, N., . . . Viennot, P. (2010). Multimodel comparison of a major flood in the groundwater-fed basin of the Somme River (France). *Hydrology and Earth System Sciences*, 14(1), 99-117. doi:10.5194/hess-14-99-2010
- Hammond, M. J., Chen, A. S., Djordjević, S., Butler, D., & Mark, O. (2015). Urban flood impact assessment: A state-of-the-art review. *Urban Water Journal, 12*(1), 14-29. doi:10.1080/1573062X.2013.857421
- Hibbs, B. J., & Sharp, J. J. M. (2012). Hydrogeological Impacts of UrbanizationHydrogeological Impacts of Urbanization. *Environmental and Engineering Geoscience*, 18(1), 3-24. doi:10.2113/gseegeosci.18.1.3
- Hodgson, R. (2008). Groundwater and flooding in Oxford: examining key soil processes. University of Reading MSc dissertation.
- House, A. R., Thompson, J. R., Sorensen, J. P. R., Roberts, C., & Acreman, M. C. (2016). Modelling groundwater/surface water interaction in a managed riparian chalk valley wetland. *Hydrological Processes*, 30(3), 447-462. doi:10.1002/hyp.10625
- Hughes, A. G., Vounaki, T., Peach, D. W., Ireson, A. M., Jackson, C. R., Butler, A. P., . . . Wheater, H. S. (2011). Flood risk from groundwater: examples from a Chalk catchment in southern England. *Journal of Flood Risk Management*, 4(3), 143-155. doi:10.1111/j.1753-318X.2011.01095.x
- Institute of Hydrology. (1987). Worton Rectory Farm (Oxon) feasibility study on the likely effects of proposed gravel scheme on Yarnton and Pixey Mead SSSI. *Institute of Hydrology Report*.
- Jackson, C., & Spink, A. (2004). User's manual for the groundwater flow model ZOOMQ3D. British Geological Survey Internal Report, IR/04/140.
- Jain, S. K., Mani, P., Jain, S. K., Prakash, P., Singh, V. P., Tullos, D., . . . Dimri, A. P. (2018). A Brief review of flood forecasting techniques and their applications. *International Journal of River Basin Management*, *16*(3), 329-344. doi:10.1080/15715124.2017.1411920
- Jongman, B., Ward, P. J., & Aerts, J. C. J. H. (2012). Global exposure to river and coastal flooding: Long term trends and changes. *Global Environmental Change*, 22(4), 823-835.

https://doi.org/10.1016/j.gloenvcha.2012.07.004

- Jung, M., Burt, T. P., & Bates, P. D. (2004). Toward a conceptual model of floodplain water table response. *Water Resources Research*, 40(12). doi:10.1029/2003wr002619
- Kollet, S. J., & Maxwell, R. M. (2006). Integrated surface–groundwater flow modeling: A free-surface overland flow boundary condition in a parallel groundwater flow model. Advances in Water Resources, 29(7), 945-958.
- Lane, S. N. (2017). Natural flood management. *Wiley Interdisciplinary Reviews: Water, 4*(3), e1211. doi:10.1002/wat2.1211
- Lewandowski, J., G. Lischeid, and G. Nuetzmann (2009), Drivers of water level fluctuations and hydrological exchange between groundwater and surface water at the lowland River Spree (Germany): field study and statistical analyses, *Hydrological Processes*, 23(15), 2117-2128. doi: 10.1002/hyp.7277
- Lewin, J., & Ashworth, P. J. (2014). The negative relief of large river floodplains. *Earth-Science Reviews, 129*, 1-23. https://doi.org/10.1016/j.earscirev.2013.10.014
- Liu, B., & Coulthard, T. J. (2017). Modelling the interaction of aeolian and fluvial processes with a combined cellular model of sand dunes and river systems. *Computers & Geosciences, 106*, 1-9. doi:10.1016/j.cageo.2017.05.003
- Macdonald, D.M.J., Bloomfield, J., Hughes, A., MacDonald, A., Adams, B., & McKenzie, A. (2008). Improving the understanding of the risk from groundwater flooding in the UK. *Proceedings of the European Conference on Flood Risk Management: Research into Practice*. 30 September-2 October 2008, Keble College, Oxford, UK. CRC Press, The Netherlands.
- Macdonald, D., Dixon, A., Newell, A., & Hallaways, A. (2012). Groundwater flooding within an urbanised flood plain. *Journal of Flood Risk Management, 5*(1), 68-80. doi:10.1111/j.1753-318X.2011.01127.x
- Macdonald, D. M. J., Dixon, A. J., & Gooddy, D. C. (2018a). Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications. *Ecological Engineering*, *123*, 226-237. https://doi.org/10.1016/j.ecoleng.2018.09.005
- Macdonald, D. M. J., Jackson, C. R., & Mansour, M. M. (2018b). A groundwater flow model of the alluvial sands and gravels aquifer of the River Thames floodplain in the vicinity of Oxford, UK. *British Geological Survey Open Report*, OR/18/067.
- Macklin, M. G., Jones, A. F., & Lewin, J. (2010). River response to rapid Holocene environmental change: evidence and explanation in British catchments. *Quaternary Science Reviews, 29*(13), 1555-1576. https://doi.org/10.1016/j.quascirev.2009.06.010
- Mansour, M., & Hughes, A. (2004). User's manual for the distributed recharge model ZOODRM. *British Geological Survey Internal Report*, OR/04/150.
- Marsh, T. (2008). A hydrological overview of the summer 2007 floods in England and Wales. *Weather*, 63(9), 274-279. doi:10.1002/wea.305
- Marsh, T., & Hannaford, J. (2007). The summer 2007 floods in England and Wales–a hydrological appraisal. *Centre for Ecology & Hydrology Report*.
- Marsh, T., & Hannaford, J. (2008). UK hydrometric register. *Hydrological data UK series, Centre for Ecology & Hydrology, Wallingford, UK*.
- McMillan, H. K., & Brasington, J. (2007). Reduced complexity strategies for modelling urban floodplain inundation. *Geomorphology*, *90*(3), 226-243. https://doi.org/10.1016/j.geomorph.2006.10.031
- Moftakhari, H. R., AghaKouchak, A., Sanders, B. F., Allaire, M., & Matthew, R. A. (2018). What Is Nuisance Flooding? Defining and Monitoring an Emerging Challenge. *Water Resources Research*, 54(7), 4218-4227. doi:10.1029/2018wr022828
- Moftakhari, H. R., AghaKouchak, A., Sanders, B. F., & Matthew, R. A. (2017). Cumulative hazard: The case of nuisance flooding. *Earth's Future*, *5*(2), 214-223. doi:10.1002/2016EF000494
- Morris, S. E., Cobby, D., Zaidman, M., & Fisher, K. (2018). Modelling and mapping groundwater

flooding at the ground surface in Chalk catchments. *Journal of Flood Risk Management, 11*, S251-S268. doi:10.1111/jfr3.12201

- Naganna, S. R., Deka, P. C., Ch, S., & Hansen, W. F. (2017). Factors influencing streambed hydraulic conductivity and their implications on stream–aquifer interaction: a conceptual review. *Environmental Science and Pollution Research*, 24(32), 24765-24789. doi:10.1007/s11356-017-0393-4
- Neal, J., Schumann, G., Fewtrell, T., Budimir, M., Bates, P., & Mason, D. (2011). Evaluating a new LISFLOOD-FP formulation with data from the summer 2007 floods in Tewkesbury, UK. *Journal of Flood Risk Management*, 4(2), 88-95. doi:10.1111/j.1753-318X.2011.01093.x
- Newell, A. J. (2008). Morphology and Quaternary geology of the Thames floodplain around Oxford. *British Geological Survey Open Report*, OR/08/030.
- Pinder, G. F., Bredehoeft, J. D., & Cooper Jr., H. H. (1969). Determination of Aquifer Diffusivity from Aquifer Response to Fluctuations in River Stage. *Water Resources Research*, *5*(4), 850-855. doi:10.1029/WR005i004p00850
- Sampson, C. C., Fewtrell, T. J., Duncan, A., Shaad, K., Horritt, M. S., & Bates, P. D. (2012). Use of terrestrial laser scanning data to drive decimetric resolution urban inundation models. *Advances in Water Resources*, 41, 1-17. doi:10.1016/j.advwatres.2012.02.010
- Schumann, G. J. P., Neal, J. C., Voisin, N., Andreadis, K. M., Pappenberger, F., Phanthuwongpakdee, N., ... Bates, P. D. (2013). A first large-scale flood inundation forecasting model. *Water Resources Research*, 49(10), 6248-6257. doi:10.1002/wrcr.20521
- Teng, J., Jakeman, A. J., Vaze, J., Croke, B. F. W., Dutta, D., & Kim, S. (2017). Flood inundation modelling: A review of methods, recent advances and uncertainty analysis. *Environmental Modelling & Software, 90*, 201-216. doi:10.1016/j.envsoft.2017.01.006
- Thieken, A. H., Müller, M., Kreibich, H., & Merz, B. (2005). Flood damage and influencing factors: New insights from the August 2002 flood in Germany. *Water Resources Research*, *41*(12). doi:10.1029/2005WR004177
- Thompson, J. R., Iravani, H., Clilverd, H. M., Sayer, C. D., Heppell, C. M., & Axmacher, J. C. (2017). Simulation of the hydrological impacts of climate change on a restored floodplain. *Hydrological Sciences Journal-Journal Des Sciences Hydrologiques, 62*(15), 2482-2510. doi:10.1080/02626667.2017.1390316
- Verhoef, A., Fernandez-Galvez, J., Diaz-Espejo, A., Main, B. E., & El-Bishti, M. (2006). The diurnal course of soil moisture as measured by various dielectric sensors: Effects of soil temperature and the implications for evaporation estimates. Journal of Hydrology, 321(1-4), 147-162.
- Walling, D. E., & Fang, D. (2003). Recent trends in the suspended sediment loads of the world's rivers. *Global and Planetary Change, 39*(1), 111-126. https://doi.org/10.1016/S0921-8181(03)00020
- Wang, H. F. & Anderson, M. P. (1982). Introduction to Groundwater Modeling: Finite Difference and Finite Element Methods. W. H. Freeman, San Francisco. 237.
- Ward, P. J., Jongman, B., Aerts, J. C. J. H., Bates, P. D., Botzen, W. J. W., Diaz Loaiza, A., . . . Winsemius, H. C. (2017). A global framework for future costs and benefits of river-flood protection in urban areas. *Nature Climate Change*, *7*, 642. doi:10.1038/nclimate3350
- Werner, M. G. F., Hunter, N. M., & Bates, P. D. (2005). Identifiability of distributed floodplain roughness values in flood extent estimation. *Journal of Hydrology*, 314(1), 139-157. doi:https://doi.org/10.1016/j.jhydrol.2005.03.012
- Willner, S. N., Otto, C., & Levermann, A. (2018). Global economic response to river floods. *Nature Climate Change*, 8(7), 594-598. doi:10.1038/s41558-018-0173-2
- Younger, P. L., Mackay, R., & Connorton, B. J. (1993). Streambed Sediment as a Barrier to Groundwater Pollution: Insights from Fieldwork and Modelling in the River Thames Basin. *Water and Environment Journal, 7*(6), 577-585. doi:10.1111/j.1747-6593.1993.tb00890.x
- Yu, D., & Lane, S. N. (2006), Urban fluvial flood modelling using a two-dimensional diffusion-wave treatment, part 1: mesh resolution effects, *Hydrological Processes*, 20(7), 1541-1565. doi: 10.1002/hyp.5935

SUPPLEMENTARY INFORMATION

Aquifer	Aguifer	River	Surface	Manning's	P1	P2	P3	A1	Average
hydraulic	specific	Thames bed	layer	roughness	RMSE	RMSE	RMSE	RMSE	RMSE for
conductivity	storage	hydraulic	hydraulic	coefficient	(cm)	(cm)	(cm)	(cm)	P1, P2, P3,
(m d ⁻¹)	(m ⁻¹)	conductivity	conductivity	(m ^{1/3} s ⁻¹)					A1 (cm)
		(m d ⁻¹)	(m d ⁻¹)						
1000	0.001	10	0.01	0.025	8	10	19	6	11
1000	0.001	10	0.03	0.025	7	9	17	5	9
1000	0.001	10	0.1	0.025	7	7	16	5	9
1000	0.001	10	0.3	0.025	7	7	15	5	8
1000	0.001	10	0.5	0.025	7	7	15	5	8
1000	0.00005	10	0.03	0.025	7	10	15	5	10
1000	0.0001	10	0.03	0.025	7	8	16	6	10
1000	0.001	10	0.03	0.025	7	9	17	5	9
1000	0.01	10	0.03	0.025	6	10	19	6	10
1000	0.0005	10	0.3	0.025	7	8	15	5	9
1000	0.001	10	0.3	0.025	7	7	15	5	8
1000	0.01	10	0.3	0.025	6	7	17	5	9
500	0.001	10	0.3	0.025	16	23	39	8	21
1000	0.001	10	0.3	0.025	7	7	15	5	8
1350	0.001	10	0.3	0.025	9	7	13	5	8
1000	0.001	5	0.3	0.025	7	7	17	5	9
1000	0.001	10	0.3	0.025	7	7	15	5	8
1000	0.001	20	0.3	0.025	7	7	15	5	8
1000	0.001	10	0.3	0.025	7	7	15	5	8
1000	0.001	10	0.3	0.05	7	7	15	5	8

Table S1Linked model simulations for the July 2007 flood event: model parameters and RMSE
between observed and simulated water levels at P1, P2, P3 and A1. Note, some model
run results are repeated for ease of comparison.

Chapter 6: Conclusions

6 Conclusions

The broad aim of this thesis has been to improve the understanding of the interaction between surface waters and groundwaters in permeable lowland floodplains, with a particular focus on water and nutrient fluxes, and flooding. The conclusions from the research are presented here in relation to the four specific aims set out in Section 1.2.1. Although the research in each of Chapters 2 to 5 map on to individual aims, in assessing how the results of each chapter have addressed these aims, insights are drawn in from across the full range of research undertaken. Section 6.1 includes the implications for environmental management; areas for further work are presented in Section 6.2.

6.1 Addressing aims, and implications for environmental management

1) Improved understanding of the impact of river management structures on water and nitrate exchange in urbanised floodplain aquifers

Evidence was provided that highlighted that engineered river management structures are common on the river networks of England and Wales, and in other countries globally. Evidence from the study area showed that these structures can have a substantial effect on both river levels, and groundwater levels in adjacent permeable floodplain aquifers. Raised groundwater levels are associated with increased aquifer storage, which may be important where these are used as sources of groundwater supplies. The study also showed that in areas with dense networks of water courses, the combination of rivers with raised water levels and natural or constructed bypass channels that are generally receiving groundwater, can result in complex groundwater flow patterns. It was proposed that these complex flow patterns can result in water residing in the subsurface for longer than might have been the case with a more natural systems, with possible implications for groundwater quality.

An observation from the Oxford floodplain was that sites upstream of locks, where groundwater levels are particularly shallow, are often the locations of floodplain meadows. In Oxford these sites have national and European-level protection, having rich and diverse flora and fauna. The plant species are highly dependent on soil moisture conditions and any change to these conditions can have significant implications for species composition. Therefore any change to the design or operation of structures in similar settings could have implications for the status of these types of ecosystem.

The fluxes of water and nitrate were the focus of Chapter 2. The Oxford floodplain aquifer is isolated from regional aquifers as it is underlain almost completely by very low permeability bedrock. As a result, high nitrate groundwater does not flow laterally from outside of the floodplain. It is notable that in the River Thames catchment, extensive areas of alluvial gravels are generally underlain by mudstone bedrock as these form wide, shallow, low-gradient zones in which alluvial sediments most readily accumulate (Section 1.3, Figure 3). The hydrogeological setting of Oxford is therefore not atypical. In the study area the input of high nitrate concentration water to the floodplain aquifer is a combination of local sources, primarily urban, and water from the River Thames. A conceptual model was proposed focussing on the latter of these two inputs. Two mechanisms were identified for the recharge of the floodplain aquifer by this high nitrate water: raised river levels, resulting from river management structures, causing a flux of water from river to aquifer through the river bed; and overbanking of river water infiltrating through the floodplain surface during periods of flooding. Simple models used evidence from measurements of groundwater chemistry to show that there must be substantial nitrate removal occurring in the river bed sediments, and in the floodplain more diffusely, as groundwater nitrate concentrations measured were generally very low. Groundwater chemistry data at three depths were available at six sites with nested boreholes (see Chapter 3) but did not show any vertical variation in nitrate concentration although stratification in redox-reactivity might be expected (Kolbe et al., 2018). (N.B., in Chapter 2 reference is made to denitrification in the hyporheic zone. It is recognised that although denitrification is likely to be the dominant process, other processes might have been contributing to the lowering of nitrate concentration in groundwater, and therefore the more generic term 'nitrate removal' should have been used. In addition, as there is debate within the peer-reviewed literature as to the definition of the hyporheic zone, it is recognised that it would have been more appropriate to have referred to 'river bed sediments' as the location for much of the nitrate removal.)

The insights from research reported in Chapter 5 raised some issues with the calculations presented in Chapter 2. The calibration of the groundwater flow component of the model presented in Chapter 5 indicated that the hydraulic conductivity of the bed of the River Thames may be substantially higher than that used in calculations in Chapter 2; the river bed hydraulic conductivity used in the baseline model in Chapter 5 was 10 m d⁻¹, and in Chapter 2 was 1 m d⁻¹. This would have increased the estimate of the amount of nitrate attenuation occurring within the river bed sediments, calculated in Chapter 2 as 1×10^3 kg/a/km, by an order of magnitude. The modelling undertaken in Chapter 5 also improved the understanding of water fluxes during flood events. The modelling indicated that the simple approach taken to estimate groundwater recharge in Chapter 2 may not be valid, i.e. equating recharge with the average volume of the unsaturated zone in winter months. The modelling in Chapter 5 showed groundwater discharge, rather than recharge, occurring over most of the floodplain during the main period of the July 2007 flood event. In this modelling study, net groundwater recharge was shown to occur during the latter period of the flood and was of a similar order to that produced by the simple model in Chapter 2; the modelled estimate of recharge resulting from a single event, when averaged over the whole floodplain, was 52 mm in Chapter 2, compared with the equivalent in Chapter 5 of 29 and 41 mm, using surface layer hydraulic conductivities of 0.03 and 0.3 m d⁻¹, respectively. However, the recharge from the linked model system in Chapter 5 was focussed primarily in the south of Port Meadow. Groundwater from boreholes in this area of the floodplain (PTM27 and PTM30 in Table 3 in Chapter 3) show a greater range in nitrate concentrations than others in the study, indicating a higher influx of nitrate from fluvial flood waters. However, the median values of groundwater nitrate concentrations are still very low in these boreholes, showing almost complete nitrate removal is occurring.

These improvements to the conceptual model do not change the conclusions of this aspect of the thesis, that the flux of nitrate into the floodplain aquifer is potentially significant for groundwater quality, but that conditions within the floodplain result in almost complete nitrate removal. The anaerobic conditions created by the shallow unsaturated zone and carbon-rich, fine-grained sediments promote this groundwater nitrate removal. It is proposed that the river management structures contribute by creating raised groundwater levels.

Although it is acknowledged that there is considerable uncertainty in the results presented in Chapter 2, the research highlights the need for environmental managers to understand the hydrogeological setting, potential nitrate sources and capacity for attenuation within floodplains when considering whether high nitrate groundwater concentrations are likely where there are sensitive receptors. There are also implications for those who are considering changes to existing river management structures as part of river restoration schemes as the resulting changes to groundwater levels may have unintended environmental consequences. The research shows that there are benefits associated with the groundwater conditions produced by these structures in terms of the potential for enhanced attenuation of pollutants as well as creating optimal settings for some types of groundwater-dependent terrestrial ecosystems.

2) Better understanding of the influence of waste dumps on water quality in floodplain aquifers and quantification of the fluxes of associated pollutants to rivers via the subsurface

Data presented in Chapters 1 and 3 highlighted that there are substantial areas of floodplain in the UK that are within urban or peri-urban areas, and that many legacy waste dumps are located here. As these waste dumps may have no containment measures, this creates a serious risk to the environment that it was argued is not sufficiently recognised. The research presented in Chapter 3 helps to characterise the plume of pollution produced by the largest of the historical waste dumps in the River

Thames floodplain in Oxford. The study of nitrogen dynamics through the use of N-species chemistry, nitrogen isotopes and dissolved nitrous oxide revealed that the ammonium emanating from the waste dump is not significantly retarded by sorption to the aquifer sediments. It is hypothesised in Chapter 3 that this is because the floodplain sediments down-gradient of the historic waste dump have reached their capacity for adsorption. The lack of dissolved oxygen in the floodplain sediments which might otherwise have facilitated nitrification, is noted as a key factor in the conservation of the ammonium.

In Chapter 3, estimates are made of the potential flux of ammonium into the River Thames. This assumes that the impact of the waste dump studied was representative of others located on the floodplain in Oxford. The flux of ammonium estimated was substantial at 27.5 tonnes per annum, which, assuming no attenuation within the hyporheic zone or in-stream processing, would mean around 40% of the ammonium concentration measured in the river could be sourced from the waste dumps.

The understanding of the aquifer system that developed since the research in Chapter 3 was undertaken has some implications for the calculations made and the interpretation. The aquifer hydraulic conductivity estimated through particle size distribution calculations, reported in Chapter 3, is lower than the values resulting from the calibration of the groundwater flow model in Chapter 5. If the higher value is used, the groundwater flows, and therefore ammonium fluxes, would be much greater; the influx to the River Thames from waste dumps on the Oxford valley floodplain would be a factor of 2.5 greater than the 75 kg per day calculated. However, an assumption was made in Chapter 3 that the river bed hydraulic conductivity was the same as the aquifer hydraulic conductivity, based on the correlation between the river level and groundwater levels in a borehole very close to the river. Subsequent analysis presented in Chapters 2 and 5 showed that although possibly not as low as reported in other studies in the Thames catchment, the river bed hydraulic conductivity is likely to be at least an order of magnitude lower than the floodplain gravel aquifer. Data presented in Chapter 2 also showed that there is likely to be some flow west beneath the River Thames into an area of the floodplain aquifer with a deeper unsaturated zone where attenuation of ammonium could occur prior to discharge to the Seacourt Stream. All of this, together with the evidence of the attenuation capacity of the river bed sediments of the River Thames in Chapter 2, suggests the flux into the river may be lower than proposed in Chapter 3.

Recognising that there large uncertainties in the calculations made, it is argued that the implications of the research presented in Chapter 3 are still highly significant. Pollution from legacy waste dumps on our floodplains is likely to be substantial and this could be having a major impact on environmental receptors. The mitigation options are limited due to the setting and the cost of removal or containment. The requirement of environmental managers is, initially at least, risk assessment, including monitoring.

3) Conditions and mechanisms that control the occurrence of groundwater flooding in urbanised floodplains

The research presented in Chapter 4 was one of the first detailed studies undertaken internationally on groundwater flooding in a permeable floodplain setting. The study identified that although rapid groundwater level rise could occur in response to direct rainfall recharge, the primary driver for groundwater flooding in this type of setting is the stage in the associated river. Based on local knowledge, survey evidence and data from an extensive water level monitoring network, including temporary flood gauges, it was shown that in urbanised areas groundwater flooding was only apparent where property was protected from fluvial flood waters. These were located within residential areas surrounded by high ground mainly associated with transport routes. Research presented in Chapter 5 indicates that groundwater discharge would likely occur throughout the floodplain but is masked by fluvial flood waters. Property can be impacted by groundwater flooding at ground floor level where it is built on low-lying ground in areas protected from fluvial flooding. However, the greatest impact on property has been on the basements of houses that have been converted into living areas. Where land for development was known historically to be particularly vulnerable to flooding, houses were built with the ground floor higher above the floodplain than elsewhere. This created a tall void beneath the ground floor that was perceived during the second half of the 20th Century, a period of relatively infrequent flooding and rising house prices, as a risk-free and cost-effective option for adding living space. In addition to property, Chapter 4 also reported that infrastructure was vulnerable to groundwater flooding. High groundwater levels resulted in: inflow to the sewer network compromising its operation; and inundation of telecommunication wiring. As highlighted in Chapter 5, persistent low level nuisance flooding, which is influenced by subsurface conditions, also caused issues with the transport network.

It was commonly thought that the subsurface would soak up overbanked water during periods of flooding. In Chapter 4, the capacity of the subsurface to accept water was calculated and shown to be insignificant in relation to the volumes of river water flowing through the floodplain during major events. This calculation used gravel porosity values but it may have been more appropriate to use the equivalent for the alluvium, although this would not have changed the conclusion.

The research reported in Chapter 4 was undertaken in collaboration with the Environment Agency, as part of the Oxford Flood Risk Management Study (OFRMS). General measures to reduce groundwater flood risk were discussed in Chapter 4. These were considered as part of the OFRMS, but at a city scale the only cost-effective option to address groundwater flooding was to reduce the period and height of the fluvial flood peak. This is likely to be the case in other similar urban settings. This is best done by increasing conveyance of river water through urbanised flood plain areas. This was also the optimum option identified by the OFRMS to reduce fluvial flood risk in Oxford. The scheme designed did not get implemented because the economic case was not strong enough to get Government funding. However, following further flooding in 2014, additional analysis was undertaken (I acted as an advisor to this project), and as a consequence the Oxford Flood Alleviation Scheme was designed, on the same principles, and has received funding.

The research undertaken as part of this thesis occurred at a time when flood policy was recognising the need to assess groundwater flood risk as part of a holistic approach to flood management. However, perhaps contradictory to this ethos, as discussed in Section 1.1.5, the UK response to the EU Floods Directive requirements and the concerns raised as part of the Pitt Review (Pitt, 2008), was to devolve responsibility on groundwater flooding, as well as pluvial flooding, to local authorities (in the flood context, referred to as Lead Local Flood Authorities; LLFAs). The understanding developed through research undertaken here is an aid to those responsible for assessing the significance of groundwater flooding. The insights into the potential speed of response of groundwater levels to rainfall recharge is also of relevance to the design of SuDS as measures to aid pluvial flood alleviation. The groundwater flooding research reported in this thesis has informed the development by the British Geological Survey of national maps of groundwater flood susceptibility at 1:50,000 scale (that I also have been involved in producing). These have been provided to LLFAs to help identify areas of high groundwater flood risk, a requirement of the EU Floods Directive.

4) A flood model system incorporating groundwater processes, used to understand the role groundwater and floodplain properties play in controlling the persistence of flooding in urbanised floodplains

Observations from the study area and reference to the international research literature has highlighted the impact of persistent, shallow flood waters that extend the length of flood events and have tangible economic consequences. The study presented in Chapter 5 explored the extent to which groundwater levels and the hydraulic conductivity of the surface alluvium, control the duration of this type of flooding in permeable floodplains. Flood inundation models that incorporate subsurface processes are not common. Those models that have been used to simulate floodplain-aquifer

interactions, have generally been applied within research studies to understand physical processes. The models are commonly complex, physics-based codes that require long run-times. As part of the study, to address the lack of routinely used codes for flood simulation and prediction that represent groundwater, a model system was developed that links a groundwater flow model, ZOOMQ3D, to the widely-applied, computationally-efficient, cellular flood inundation model, LISFLOOD-FP. The models are linked via a low hydraulic conductivity layer that allows vertical fluxes to be quantified and passed between the two models during run time.

It is recognised that there are limitations to the linked model system; these are described in the Section 4 of Chapter 5. For example, there is an inconsistency in the model system, in that groundwater recharge is calculated in advance of the linked model run in locations that are inundated by flood waters. The lack of capture of direct rainfall input to, and evaporation from, flood waters on the floodplain may be significant under certain circumstances, as may ignoring unsaturated zone flow processes. In addition, the model does not incorporate the effects of subsurface infrastructure on the recession of groundwater and flood waters, which limits its application in urban environments. However, the case study application does allow the performance of the model system to be tested in a moderately modified floodplain setting.

The linked model system was applied to the Oxford floodplain, simulating the alluvial gravel aquifer, surface flows on the floodplain, and flows between the two via the shallow alluvium. The linked model system was used to simulate flooding processes in Port Meadow, a well-monitored area of the River Thames floodplain that regularly floods, and where flood waters are retained due to natural levees and anthropogenically-modified topography.

There were issues with simulating groundwater and flood water levels in the study that would be experienced with any application of the model. Floodplain aquifers are highly heterogeneous systems. It is challenging to both set-up a model structure that adequately captures all the key components, and for it to be parameterised. Observed water levels, including flood waters, are required to assess the performance of the model system, and monitoring networks in these settings are rare. Gradients are low and even minor errors in topography, datums and water level measurements can be problematic. Some of these issues were apparent in the modelling study presented in Chapter 5 but nevertheless the output of the model system produced a good fit to the observed groundwater and surface water level time-series, and allowed the research aims to be met.

The linked model system provided a good match to the flood extent when compared with photographs taken close to the peak of the event and to water level time-series at specific locations. The insights from the fluxes produced by the model were of particular interest, as already discussed in this Section. As shown in Chapter 5, where the alluvial aquifer is confined and where three dimensions are considered, water fluxes can be complex. The modelling of Port Meadow during the July 2007 flood event, simulated net groundwater discharge over most of the floodplain, as the groundwater heads in the high diffusivity aquifer closely followed the River Thames levels, which were generally higher than the flood water levels on the floodplain. Net recharge only occurred where flood waters accumulated due to topographical controls. This is a significant contribution to the international scientific literature on this topic, which generally reports large and uniform volumes of groundwater recharge occurring across floodplains as a result of fluvial flooding.

In the case of Port Meadow in the July 2007 event, the vertically discharging groundwater was a minor component of the floodplain water balance. As discussed in Chapter 1, the use of floodplains to store flood waters upstream of vulnerable urban centres is gaining interest as one of the options for flood risk management that does not rely on hard flood defences. This is achieved with embankments on floodplains and managed river inlets. Questions have been posed as to whether groundwater might compromise the performance of these systems. The insights from Chapter 5 are of relevance as Port Meadow acts in a similar way to an upstream flood store. In this case groundwater inflow to the floodplain is not a high proportion of the floodplain storage capacity and would not have compromised

its performance if it had been used to store flood waters. However, the research does highlight that groundwater can interact with floodplain stores and it is important for flood managers to understand floodplain hydrogeology when designing them.

Chapter 5 confirms that shallow groundwater and superficial deposits could have a significant bearing on flood duration if the depth threshold of impactful flooding considered for flood risk assessment is lowered. As flooding becomes more frequent in the future, the economic consequences of this aspect of flooding may become substantial and consideration should be given to it by the relevant agencies when planning the gathering of evidence during floods. When undertaking flood risk modelling it should also be assessed whether the circumstances require the simulation of the interaction of flood waters with groundwater, as well as the nature of the superficial geology to be taken into consideration. The model developed in Chapter 5 provides an excellent tool in this context. In terms of mitigation, improved drainage for the removal of flood waters is the primary option. However, this needs to be designed recognising that drainage routes during recession can be flood water pathways during main flood events.

6.2 Further work

Recommendations for further work relate to the study area, to further understand aspects of the floodplain system, as well as more generic applications:

• The Port Meadow study involved a limited amount of field measurement of parameters used within the linked model system, in particular surface layer and gravel thickness, and aquifer, river bed and surface layer hydraulic conductivities. Better quantification of these parameters would help to constrain the ranges used within the calibration stage of the model development and the sensitivity analysis. These parameters need to be measured at a density that matches their likely spatial variability. This should include a combination of point measurements and geophysical surveying. Geophysical surveying, using a technique such as 3D electrical resistivity tomography, combined with a high-density intrusive methodology such as penetration with a steel rod to identify the alluvium base, and further shallow drilling (Chambers et al., 2014), would allow the thickness of both the alluvium and the gravels to be better mapped. Aquifer tests, both pumping tests and falling head tests, could be undertaken on newly drilled and existing boreholes within Port Meadow to estimate hydraulic conductivity and specific storage of the gravel aquifer. Permeameter tests at a range of depths within the alluvium would help to quantify its hydraulic conductivity (MacDonald et al., 2012).

Tests within the river bed sediments could be undertaken to quantify its hydraulic conductivity, including methods using such permeameters, seepage meters, and freeze coring along with lab permeability measurement (Brunner et al., 2017). It is recognised that this involves significant challenges given the length, width and depth of the River Thames in the Port Meadow area, and the public access that makes it problematic if equipment needs to be left in situ. An additional approach to parameter quantification would be to use available analytical numerical methods on measured river and groundwater level time-series data (Barlow & Moench, 1998).

Further and higher density high-frequency monitoring of flood water and groundwater levels would also improve the calibration of the model. The linked model system provides a tool for exploring processes within floodplains, in relation to pollutant processing, ecosystem status as well as flooding.

 It was suggested in the thesis that the complex groundwater flow pathways, resulting from managed river systems and their associated bypass channels, could result in longer residence times for groundwaters in floodplain aquifers, and that this could affect pollutant attenuation and possibly recovery from flooding. Research to explore this aspect of floodplain hydrology would be of value. This would involve focussed monitoring of zones where these complex pathways occur, including higher spatial density of nested piezometers, to allow groundwater heads, temperature and water chemistry to be measured. Additional spatial characterisation of the floodplain aquifer and associated water courses, using techniques described above, would also be necessary.

- The research in this thesis identified that interventions to improve ecological status of aquatic and terrestrial ecosystems, that change groundwater levels, could have potential unintended consequences, in particular for pollutant attenuation capacity of the subsurface, and soil moisture conditions to which floodplain vegetation can be so sensitive. Few studies have been undertaken which adequately compare pre- and post-intervention conditions to assess the benefits and disadvantages. This research is of value in the context of river restoration schemes, wetland development and natural flood management measures.
- At a national scale, the monitoring of floodplain aquifers needs to be improved. Our national groundwater level monitoring network has been developed primarily for groundwater resource assessment. Water supplied from floodplain aquifers is not insignificant, however, the network of monitoring boreholes in this setting is relatively small. It could be argued that the dense river monitoring network provides a good measure of the storage in floodplain aquifers, however, there is a need for a greater number of boreholes that are issue- and receptor-specific. These would be located to provide warning of flooding to vulnerable communities or monitor pollution plumes in relation to important ecological or water supply sources.
- The study identified that a large number of legacy waste dumps are located on floodplain aquifers, with possibly limited containment measures. It is suggested that further investigation is undertaken in the Oxford floodplain to examine the cumulative impact of waste dumps on groundwater and river quality. The flux of ammonium into the River Thames in the Port Meadow study area was estimated to be potentially substantial, however the degree of attenuation in the river bed sediments was not measured. In the context of a wider floodplain study, further work to examine processes in this zone would be of value as would investigation of in-river attenuation.
- This research would be valuable in the context of a national scale assessment of the impact of legacy waste dumps on our floodplain aquifers and associated aquatic and terrestrial ecosystems. An initial screening study could be relatively easily undertaken as collated datasets on licensed historical landfills are available from the environment regulators.
- The linked model system developed in this study could be applied to urbanised areas of the Oxford floodplain where, additionally, the impact of drainage and the built environment on the alluvium layer would need to be assessed. Model systems such as this should include interaction of surface and groundwater flows with subsurface infrastructure, to allow a holistic simulation of flood mechanisms. Although research is being undertaken to incorporate these processes in flood models, there is still much work to be done to produce models for operational use.
- The linked model system could also be applied to other groundwater flood settings, in particular on the outcrop of the Chalk outcrop. This is work that BGS plans to undertake.
- The issue of nuisance flooding has been raised in this thesis. This issue needs to be considered further and evidence gathered to assess if it is a significant aspect of flooding elsewhere in urban floodplain settings. An economic analysis in the context of urbanised UK floodplains needs to be undertaken. Models, such as the one developed as part of this study, may need to be applied more routinely within flood risk management assessments.

References

- Abboud, J. M., Ryan, M. C., & Osborn, G. D. (2018). Groundwater flooding in a river-connected alluvial aquifer. Journal of Flood Risk Management, 11(4), e12334.
- Abbott, B. W., Baranov, V., Mendoza-Lera, C., Nikolakopoulou, M., Harjung, A., Kolbe, T., ... & Wallin, M. B. (2016). Using multi-tracer inference to move beyond single-catchment ecohydrology. *Earth-Science Reviews*, 160, 19-42. http://dx.doi.org/10.1016/j.earscirev.2016.06.014
- Addiscott, T. M., Whitmore, A. P., & Powlson, D. S. (1991). Farming, fertilizers and the nitrate problem. *CAB International*, 170.
- Anderson, M. P., Woessner, W. W., & Hunt, R. J. (2015). Applied groundwater modeling: simulation of flow and advective transport. Academic press.
- Antiguedad, I., Zabaleta, A., Martinez-Santos, M., Ruiz, E., Uriarte, J., Morales, T., . . . Sanchez-Perez, J.-M. (2017). A simple multi-criteria approach to delimitate nitrate attenuation zones in alluvial floodplains. Four cases in south-western Europe. *Ecological Engineering*, 103, 315-331. https://doi.org/10.1016/j.ecoleng.2016.09.007
- Arnell, N. W., & Gosling, S. N. (2016). The impacts of climate change on river flood risk at the global scale. *Climatic Change*, 134(3), 387-401. doi:10.1007/s10584-014-1084-5
- Arthington, A. H., Naiman, R. J., McClain, M. E., & Nilsson, C. (2010). Preserving the biodiversity and ecological services of rivers: new challenges and research opportunities. *Freshwater Biology*, 55(1), 1-16. doi:10.1111/j.1365-2427.2009.02340.x
- Ascott, M. J., Lapworth, D. J., Gooddy, D. C., Sage, R. C., & Karapanos, I. (2016). Impacts of extreme flooding on riverbank filtration water quality. *Science of the Total Environment, 554-555*, 89-101. https://doi.org/10.1016/j.scitotenv.2016.02.169
- Ascott, M. J., Marchant, B. P., Macdonald, D., McKenzie, A. A., & Bloomfield, J. P. (2017). Improved understanding of spatio-temporal controls on regional scale groundwater flooding using hydrograph analysis and impulse response functions. *Hydrological Processes*, *31*(25), 4586-4599. doi:10.1002/hyp.11380
- Back, J.O., Rivett, M.O., Hinz, L.B., Mackay, N., Wanangwa, G.J., Phiri, O.L., Songola, C.E., Thomas, M.A., Kumwenda, S., Nhlema, M. & Miller, A.V. (2018). Risk assessment to groundwater of pit latrine rural sanitation policy in developing country settings. *Science of the Total Environment*, 613, 592-610. doi.org/10.1016/j.scitotenv.2017.09.071
- Baillieux, A., Campisi, D., Jammet, N., Bucher, S., & Hunkeler, D. (2014). Regional water quality patterns in an alluvial aquifer: Direct and indirect influences of rivers. *Journal of Contaminant Hydrology*, *169*, 123-131. https://doi.org/10.1016/j.jconhyd.2014.09.002
- Ball, L., Clegg, M., Lewis, L., & Bell, G. (2009). Finding a long term solution to flooding in Oxford: the challenges faced. *Proceedings of the European Conference on Flood Risk Management: Research into Practice*. 30 September-2 October 2008, Keble College, Oxford, UK. CRC Press, The Netherlands.
- Barkwith, A., Hurst, M. D., Jackson, C. R., Wang, L., Ellis, M. A., & Coulthard, T. J. (2015). Simulating the influences of groundwater on regional geomorphology using a distributed, dynamic, landscape evolution modelling platform. *Environmental Modelling & Software, 74*, 1-20. doi:10.1016/j.envsoft.2015.09.001
- Barlow, P. M., & Moench, A. F. (1998). Analytical solutions and computer programs for hydraulic interaction of stream-aquifer systems (No. 98-415-A). US Dept. of the Interior, US Geological Survey; Information Services.
- Barlow J., Moore F., & Burgess-Gamble, G. (2104). Working with natural processes to reduce flood risk. In: *R&D Framework: Science Report SC130004/R2*. Environment Agency: Bristol.
- Bates, P. D., & De Roo, A. P. J. (2000). A simple raster-based model for flood inundation simulation. Journal of Hydrology, 236(1-2), 54-77. doi:10.1016/s0022-1694(00)00278-x
- Bates, P. D., Horritt, M. S., & Fewtrell, T. J. (2010). A simple inertial formulation of the shallow water equations for efficient two-dimensional flood inundation modelling. *Journal of Hydrology*, 387(1-2), 33-45. doi:10.1016/j.jhydrol.2010.03.027

- Bates, P. D., Stewart, M. D., Desitter, A., Anderson, M. G., Renaud, J. P., & Smith, J. A. (2000).
 Numerical simulation of floodplain hydrology. *Water Resources Research*, *36*(9), 2517-2529. doi:10.1029/2000wr900102
- Battin, T. J., & Sengschmitt, D. (1999). Linking Sediment Biofilms, Hydrodynamics, and River Bed Clogging: Evidence from a Large River. *Microbial Ecology*, 37(3), 185-196. doi:10.1007/s002489900142
- Bellmore, J. R., Benjamin, J. R., Newsom, M., Bountry, J. A., & Dombroski, D. (2017a). Incorporating food web dynamics into ecological restoration: a modeling approach for river ecosystems. *Ecological Applications*, 27(3), 814-832. doi:10.1002/eap.1486
- Bellmore, J. R., Duda, J. J., Craig, L. S., Greene, S. L., Torgersen, C. E., Collins, M. J., & Vittum, K. (2017b). Status and trends of dam removal research in the United States. *Wiley Interdisciplinary Reviews: Water, 4*(2), e1164. doi:10.1002/wat2.1164
- Benke, A. C., Chaubey, I., Ward, G. M., & Dunn, E. L. (2000). Flood pulse dynamics of an unregulated river floodplain in the southeastern U.S. coastal plain. *Ecology*, *81*(10), 2730-2741. doi:10.1890/0012-9658(2000)081[2730:FPDOAU]2.0.CO;2
- Bernard-Jannin, L., Brito, D., Sun, X., Jauch, E., Neves, R., Sauvage, S., & Sánchez-Pérez, J.-M. (2016).
 Spatially distributed modelling of surface water-groundwater exchanges during overbank
 flood events a case study at the Garonne River. *Advances in Water Resources, 94*, 146-159.
 https://doi.org/10.1016/j.advwatres.2016.05.008
- Bernard-Jannin, L., Sun, X., Teissier, S., Sauvage, S., & Sánchez-Pérez, J.-M. (2017). Spatio-temporal analysis of factors controlling nitrate dynamics and potential denitrification hot spots and hot moments in groundwater of an alluvial floodplain. *Ecological Engineering*, 103, 372-384. https://doi.org/10.1016/j.ecoleng.2015.12.031
- Blann, K. L., Anderson, J. L., Sands, G. R., & Vondracek, B. (2009). Effects of Agricultural Drainage on Aquatic Ecosystems: A Review. *Critical Reviews in Environmental Science and Technology*, 39(11), 909-1001. doi:10.1080/10643380801977966
- Blasch, K. W., Constantz, J., & Stonestrom, D. A. (2007). Thermal methods for investigating groundwater recharge (2330-7102). *United States Geological Survey Report*.
- Boano, F., Harvey, J. W., Marion, A., Packman, A. I., Revelli, R., Ridolfi, L., & Wörman, A. (2014).
 Hyporheic flow and transport processes: Mechanisms, models, and biogeochemical implications. *Reviews of Geophysics*, 52(4), 603-679. doi.org/10.1002/2012RG000417
- Böhlke, J. K., Smith, R. L., & Miller, D. N. (2006). Ammonium transport and reaction in contaminated groundwater: Application of isotope tracers and isotope fractionation studies. *Water Resources Research*, 42(5). doi:10.1029/2005WR004349
- Boonstra, J., & de Ridder, N. A. (1981). Numerical modelling of groundwater basins: a user-oriented manual: *ILRI Publication*, 29. Wageningen: International Institute for Land Reclamation and Improvement.
- Booth, E. G., Zipper, S. C., Loheide, S. P., & Kucharik, C. J. (2016). Is groundwater recharge always serving us well? Water supply provisioning, crop production, and flood attenuation in conflict in Wisconsin, USA. *Ecosystem Services*, *21*, 153-165. https://doi.org/10.1016/j.ecoser.2016.08.007
- Booth, K. A., & Linley, K. A. (2010). Geological indicators of flooding: user guidance notes. *British Geological Survey Open Report*, OR/10/064.
- Böttcher, J., Strebel, O., Voerkelius, S., & Schmidt, H. L. (1990). Using isotope fractionation of nitratenitrogen and nitrate-oxygen for evaluation of microbial denitrification in a sandy aquifer. *Journal of Hydrology*, *114*(3), 413-424. https://doi.org/10.1016/0022-1694(90)90068-9
- Boulton, A. J., Findlay, S., Marmonier, P., Stanley, E. H., & Valett, H. M. (1998). The functional significance of the hyporheic zone in streams and rivers. *Annual Review of Ecology and Systematics*, 29(1), 59-81. doi:10.1146/annurev.ecolsys.29.1.59
- Bouraoui, F., & Grizzetti, B. (2011). Long term change of nutrient concentrations of rivers discharging in European seas. *Science of the Total Environment, 409*(23), 4899-4916.

https://doi.org/10.1016/j.scitotenv.2011.08.015

- Bradbrook, K. (2006). JFLOW: a multiscale two-dimensional dynamic flood model. *Water and Environment Journal, 20*(2), 79-86. doi:10.1111/j.1747-6593.2005.00011.x
- Bradbrook, K. F., Lane, S. N., Waller, S. G., & Bates, P. D. (2004). Two dimensional diffusion wave modelling of flood inundation using a simplified channel representation. *International Journal of River Basin Management*, *2*(3), 211-223. doi:10.1080/15715124.2004.9635233
- Brand, J. H., Spencer, K. L., O'shea, F. T., & Lindsay, J. E. (2018). Potential pollution risks of historic landfills on low-lying coasts and estuaries. *Wiley Interdisciplinary Reviews: Water*, 5(1), e1264. doi:10.1002/wat2.1264
- Bricker, S. H., & Bloomfield, J. P. (2014). Controls on the basin-scale distribution of hydraulic conductivity of superficial deposits: a case study from the Thames Basin, UK. *Quarterly Journal of Engineering Geology and Hydrogeology*, 47(3), 223-236. doi:10.1144/qjegh2013-072
- Bridge, J. S. (2009). *Rivers and floodplains: forms, processes, and sedimentary record*: John Wiley & Sons.
- Brown, A. G., Lespez, L., Sear, D. A., Macaire, J.-J., Houben, P., Klimek, K., . . . Pears, B. (2018). Natural vs anthropogenic streams in Europe: History, ecology and implications for restoration, river-rewilding and riverine ecosystem services. *Earth-Science Reviews, 180*, 185-205. https://doi.org/10.1016/j.earscirev.2018.02.001
- Brunke, M. (1999). Colmation and depth filtration within streambeds: Retention of particles in hyporheic interstices. *International Review of Hydrobiology*, *84*(2), 99-117.
- Brunner, P., Therrien, R., Renard, P., Simmons, C. T., & Franssen, H. J. H. (2017). Advances in understanding river-groundwater interactions. *Reviews of Geophysics*, 55(3), 818-854. doi.org/10.1002/2017RG000556
- Buffin-Bélanger, T., Cloutier, C.-A., Tremblay, C., Chaillou, G., & Larocque, M. (2016). Dynamics of groundwater floodwaves and groundwater flood events in an alluvial aquifer. *Canadian Water Resources Journal / Revue canadienne des ressources hydriques, 41*(4), 469-483. doi:10.1080/07011784.2015.1102651
- Bunn, S. E., & Arthington, A. H. (2002). Basic Principles and Ecological Consequences of Altered Flow Regimes for Aquatic Biodiversity. *Environmental Management, 30*(4), 492-507. doi:10.1007/s00267-002-2737-0
- Burt, T., Pinay, G., Grimm, N., & Harms, T. (2013). Between the land and the river: river conservation and the riparian zone. *River Conservation: Challenges and Opportunities*, 217-240.
- Burt, T. P., Bates, P. D., Stewart, M. D., Claxton, A. J., Anderson, M. G., & Price, D. A. (2002). Water table fluctuations within the floodplain of the River Severn, England. *Journal of Hydrology*, 262(1-4), 1-20. doi:10.1016/s0022-1694(01)00567-4
- Burt, T. P., Howden, N. J. K., Worrall, F., Whelan, M. J., & Bieroza, M. (2011). Nitrate in United Kingdom Rivers: Policy and Its Outcomes Since 1970. *Environmental Science & Technology*, 45(1), 175-181. doi:10.1021/es101395s
- Burt, T. P., Matchett, L. S., Goulding, K. W. T., Webster, C. P., & Haycock, N. E. (1999). Denitrification in riparian buffer zones: the role of floodplain hydrology. *Hydrological Processes*, 13(10), 1451-1463. doi:10.1002/(SICI)1099-1085(199907)13:10<1451::AID-HYP822>3.0.CO;2-W
- Burt, T. P., & Pinay, G. (2005). Linking hydrology and biogeochemistry in complex landscapes. Progress in Physical Geography: Earth and Environment, 29(3), 297-316. doi:10.1191/0309133305pp450ra
- Burt, T. P., Pinay, G., Matheson, F. E., Haycock, N. E., Butturini, A., Clement, J. C., . . . Maitre, V. (2002). Water table fluctuations in the riparian zone: comparative results from a pan-European experiment. *Journal of Hydrology*, 265(1), 129-148. https://doi.org/10.1016/S0022-1694(02)00102-6
- Burt, T. P., & Shahgedanova, M. (1998). An historical record of evaporation losses since 1815 calculated using long-term observations from the Radcliffe Meteorological Station, Oxford,

England. *Journal of Hydrology, 205*(1), 101-111. https://doi.org/10.1016/S0022-1694(97)00143-1

- Buss, S. R., Herbert, A. W., Morgan, P., Thornton, S. F., & Smith, J. W. N. (2004). A review of ammonium attenuation in soil and groundwater. *Quarterly Journal of Engineering Geology* and Hydrogeology, 37(4), 347-359. doi:10.1144/1470-9236/04-005
- Cabrera, F., Clemente, L., Díaz Barrientos, E., López, R., & Murillo, J. M. (1999). Heavy metal pollution of soils affected by the Guadiamar toxic flood. *Science of the Total Environment, 242*(1), 117-129. https://doi.org/10.1016/S0048-9697(99)00379-4
- Calver, A. (2001). Riverbed Permeabilities: Information from Pooled Data. *Ground Water, 39*(4), 546-553. doi:10.1111/j.1745-6584.2001.tb02343.x
- Cardenas, M. B. (2015). Hyporheic zone hydrologic science: A historical account of its emergence and a prospectus. *Water Resources Research*, *51*(5), 3601-3616. doi:10.1002/2015WR017028
- Cardenas, M. B., & Zlotnik, V. A. (2003). A Simple Constant-Head Injection Test for Streambed Hydraulic Conductivity Estimation. *Groundwater*, *41*(6), 867-871. doi:10.1111/j.1745-6584.2003.tb02428.x
- Ceazan, M. L., Thurman, E. M., & Smith, R. L. (1989). Retardation of ammonium and potassium transport through a contaminated sand and gravel aquifer: the role of cation exchange. *Environmental Science & Technology, 23*(11), 1402-1408. doi:10.1021/es00069a012
- Chambers, J. E., Wilkinson, P. B., Uhlemann, S., Sorensen, J. P. R., Roberts, C., Newell, A. J., Ward, W.
 O. C., Binley, A., Williams, P. J., Gooddy, D. C. & Old, G. (2014). Derivation of lowland riparian wetland deposit architecture using geophysical image analysis and interface detection.
 Water Resources Research, 50(7), 5886-5905. doi:10.1002/2014WR015643
- Chave, P. (2001). The EU water framework directive. IWA publishing.
- Chow, V. T. (1998). Applied Hydrology. McGraw-Hill, Inc., New York.
- Christensen, T. H., Kjeldsen, P., Bjerg, P. L., Jensen, D. L., Christensen, J. B., Baun, A., . . . Heron, G. (2001). Biogeochemistry of landfill leachate plumes. *Applied Geochemistry*, *16*(7), 659-718. https://doi.org/10.1016/S0883-2927(00)00082-2

CIA (2002). The world fact book. *Central Intelligence Agency*, Office of Public Affairs, Washington DC.

- Cisowska, I., & Hutchins, M. G. (2016). The effect of weirs on nutrient concentrations. *Science of the Total Environment, 542*, 997-1003. https://doi.org/10.1016/j.scitotenv.2015.10.064
- Ciszewski, D., & Grygar, T. M. (2016). A Review of Flood-Related Storage and Remobilization of Heavy Metal Pollutants in River Systems. *Water, Air, & Soil Pollution, 227*(7), 239. doi:10.1007/s11270-016-2934-8
- Claxton, A. J., Bates, P. D., & Cloke, H. L. (2003). Mixing of Hillslope, River, and Alluvial Ground Waters in Lowland Floodplains. *Ground Water*, *41*(7), 926-936. doi:10.1111/j.1745-6584.2003.tb02435.x
- Clément, J.-C., Aquilina, L., Bour, O., Plaine, K., Burt, T. P., & Pinay, G. (2003). Hydrological flowpaths and nitrate removal rates within a riparian floodplain along a fourth-order stream in Brittany (France). *Hydrological Processes*, *17*(6), 1177-1195. doi:10.1002/hyp.1192
- Clilverd, H. M., Thompson, J. R., Heppell, C. M., Sayer, C. D., & Axmacher, J. C. (2016). Coupled Hydrological/Hydraulic Modelling of River Restoration Impacts and Floodplain Hydrodynamics. *River Research and Applications*, *32*(9), 1927-1948. doi:10.1002/rra.3036
- Cloutier, C.-A., Buffin-Bélanger, T., & Larocque, M. (2014). Controls of groundwater floodwave propagation in a gravelly floodplain. *Journal of Hydrology*, *511*, 423-431. https://doi.org/10.1016/j.jhydrol.2014.02.014
- Cobby, D., Morris, S., Parkes, A., & Robinson, V. (2009). Groundwater flood risk management: advances towards meeting the requirements of the EU floods directive. *Journal of Flood Risk Management, 2*(2), 111-119. doi:10.1111/j.1753-318X.2009.01025.x
- Corniello, A., Ducci, D., & Ruggieri, G. (2007). Areal identification of groundwater nitrate contamination sources in periurban areas. *Journal of Soils and Sediments*, 7(3), 159-166. doi:10.1065/jss2007.03.213

- Correll, D. L., Jordan, T. E., & Weller, D. E. (1997). Failure of agricultural riparian buffers to protect surface waters from groundwater nitrate contamination. *Forest, 22*, 23.
- Coulthard, T. J., Neal, J. C., Bates, P. D., Ramirez, J., de Almeida, G. A. M., & Hancock, G. R. (2013). Integrating the LISFLOOD-FP 2D hydrodynamic model with the CAESAR model: implications for modelling landscape evolution. *Earth Surface Processes and Landforms, 38*(15), 1897-1906. doi:10.1002/esp.3478

Cracknell, A. P. (2007). Introduction to remote sensing: CRC press.

- Dahl, M., Nilsson, B., Langhoff, J. H., & Refsgaard, J. C. (2007). Review of classification systems and new multi-scale typology of groundwater–surface water interaction. *Journal of Hydrology*, 344(1), 1-16. https://doi.org/10.1016/j.jhydrol.2007.06.027
- Dahm, C. N., Grimm, N. B., Marmonier, P., Valett, H. M., & Vervier, P. (1998). Nutrient dynamics at the interface between surface waters and groundwaters. *Freshwater Biology*, *40*(3), 427-451. doi:10.1046/j.1365-2427.1998.00367.x
- Dahm, C. N., Grimm, N. B., Marmonier, P., Valett, H. M., & Vervier, P. (1998). Nutrient dynamics at the interface between surface waters and groundwaters. *Freshwater Biology*, *40*(3), 427-451. doi:10.1046/j.1365-2427.1998.00367.x
- Davies, B. R., & Walker, K. F. (1986). River systems as ecological units. An introduction to the ecology of river systems. In B. R. Davies & K. F. Walker (Eds.), *The Ecology of River Systems* (pp. 1-8). Dordrecht: Springer Netherlands.
- de Sosa, L. L., Williams, A. P., Orr, H. G., & Jones, D. L. (2018). Riparian research and legislation, are they working towards the same common goals? A UK case study. *Environmental Science & Policy, 82*, 126-135. https://doi.org/10.1016/j.envsci.2018.01.023
- Dearden, R. A., & Price, S. J. (2012). A proposed decision-making framework for a national infiltration SuDS map. *Management of Environmental Quality: An International Journal, 23*(5), 478-485. doi:10.1108/14777831211255060
- Defra (2005). Making Space for Water: Taking Forward a New Government Strategy for Flood and Coastal Erosion Risk Management in England: *First Government Response to the Autumn* 2004 Making Space for Water Consultation Exercise, London: Department of the Environment, Food and Rural Affairs.
- Delwiche, C. C., & Steyn, P. L. (1970). Nitrogen isotope fractionation in soils and microbial reactions. *Environmental Science & Technology*, 4(11), 929-935. doi:10.1021/es60046a004
- DeSimone, L. A., & Howes, B. L. (1998). Nitrogen transport and transformations in a shallow aquifer receiving wastewater discharge: A mass balance approach. *Water Resources Research*, 34(2), 271-285. doi:10.1029/97WR03040
- Devito, K. J., Fitzgerald, D., Hill, A. R., & Aravena, R. (2000). Nitrate Dynamics in Relation to Lithology and Hydrologic Flow Path in a River Riparian Zone. *Journal of Environmental Quality, 29*(4), 1075-1084. doi:10.2134/jeq2000.00472425002900040007x
- Díaz-Redondo, M., Egger, G., Marchamalo, M., Damm, C., de Oliveira, R. P., & Schmitt, L. (2018).
 Targeting lateral connectivity and morphodynamics in a large river-floodplain system: The upper Rhine River. *River Research and Applications*, 34(7), 734-744. doi:10.1002/rra.3287
- Dixon, A. (2004). Hydrology of the Oxford Meadows cSAC. Environment Agency Report.
- Dixon, A., Bradford, R., Cooper, D., Reeve, C., & Tucker, D. (1990). A Hydrogeological Assessment of the Impact of Proposed Sand and Gravel Extraction on an Adjacent Ecologically Sensitive
 Floodplain Environment - a Case Study. In: J.W.Gaskarth & A.C. Lumsden, eds. *Proceedings of* 6th Extractive Industry Geological Conference. University of Birmingham, 70-81.
- Doble, R., Crosbie, R., Peeters, L., Joehnk, K., & Ticehurst, C. (2014). Modelling overbank flood recharge at a continental scale. *Hydrol. Earth Syst. Sci., 18*(4), 1273-1288. doi:10.5194/hess-18-1273-2014
- Doble, R. C., Crosbie, R. S., Smerdon, B. D., Peeters, L., & Cook, F. J. (2012). Groundwater recharge from overbank floods. *Water Resources Research, 48*. doi:10.1029/2011wr011441
- Domingo, N. D. S., Refsgaard, A., Mark, O., & Paludan, B. (2010). Flood analysis in mixed-urban areas

reflecting interactions with the complete water cycle through coupled hydrologic-hydraulic modelling. *Water Science and Technology, 62*(6), 1386-1392. doi:10.2166/wst.2010.365

- Doppler, T., Franssen, H.-J. H., Kaiser, H.-P., Kuhlman, U., & Stauffer, F. (2007). Field evidence of a dynamic leakage coefficient for modelling river–aquifer interactions. *Journal of Hydrology*, 347(1), 177-187. https://doi.org/10.1016/j.jhydrol.2007.09.017
- Dottori, F., Salamon, P., Bianchi, A., Alfieri, L., Hirpa, F. A., & Feyen, L. (2016). Development and evaluation of a framework for global flood hazard mapping. *Advances in Water Resources*, 94, 87-102. doi:10.1016/j.advwatres.2016.05.002
- Douglas, I. (2012). Peri-urban ecosystems and societies: transitional zones and contrasting values. In *The peri-urban interface* (pp. 41-52). Routledge.
- Downs, P., & Gregory, K. (2014). *River channel management: towards sustainable catchment hydrosystems*: Routledge.

EEA (2018). Why should we care about floodplains? European Environment Agency. Available at: https://www.eea.europa.eu/themes/water/european-waters/water-use-andenvironmental-pressures/why-should-we-care-about-floodplains

Entwistle, N. S., Heritage, G. L., Schofield, L. A., & Williamson, R. J. (2019). Recent changes to floodplain character and functionality in England. *Catena*, *174*, 490-498. https://doi.org/10.1016/j.catena.2018.11.018

- Environment Agency (2009). Oxford Flood Risk Management Strategy: Hydraulic mode-ling–model build and calibration final report. *Environment Agency Report*, Bristol, UK.
- Environment Agency (2016). Thames River Basin District, River Basin Management Plan. *Environment Agency Report*, Bristol, UK.
- Environment Agency (2018). Oxford flood scheme. Available at: https://www.gov.uk/government/publications/oxford-flood-scheme/oxford-flood-scheme.
- Erős, T., Kuehne, L., Dolezsai, A., Sommerwerk, N., & Wolter, C. (2019). A systematic review of assessment and conservation management in large floodplain rivers Actions postponed. *Ecological Indicators, 98*, 453-461. https://doi.org/10.1016/j.ecolind.2018.11.026
- ESI (2016). Oxford Flood Alleviation Scheme: Groundwater Flood Modelling. *ESI Ltd*. Confidential Report to CH2MHill.
- European Commission (2015). Towards an EU Research and Innovation Policy Agenda for Naturebased Solutions and Re-naturing Cities: *Final Report of the Horizon 2020 Expert Group on Nature-based Solutions and Re-naturing Cities*. Brussels: Directorate-General for Research and Innovation Directorate I — Climate Action and Resource Efficiency, European Commission.

Fetter, C. W., Boving, T., & Kreamer, D. (2017). *Contaminant hydrogeology*. Waveland Press.

Finch, J. W., Bradford, R. B., & Hudson, J. A. (2004). The spatial distribution of groundwater flooding in a chalk catchment in southern England. *Hydrological Processes*, 18(5), 959-971. doi:10.1002/hyp.1340

Fleming, G. (2002). Flood risk management: learning to live with rivers: Thomas Telford.

- Flipse Jr., W. J., & Bonner, F. T. (1985). Nitrogen-Isotope Ratios of Nitrate in Ground Water Under Fertilized Fields, Long Island, New York. *Ground Water*, 23(1), 59-67. doi:10.1111/j.1745-6584.1985.tb02780.x
- Forshay, K. J., & Stanley, E. H. (2005). Rapid Nitrate Loss and Denitrification in a Temperate River Floodplain. *Biogeochemistry*, 75(1), 43-64. doi:10.1007/s10533-004-6016-4
- Forster, S. S. D., Cripps, A. C., Smith-Carington, A., Stewart, W. D. P., & Rosswall, T. (1982). Nitrate leaching to groundwater. *Philosophical Transactions of the Royal Society of London. B, Biological Sciences, 296*(1082), 477-489. doi:10.1098/rstb.1982.0021
- Foster, S. S. D. (2001). The interdependence of groundwater and urbanisation in rapidly developing cities. *Urban Water*, *3*(3), 185-192. https://doi.org/10.1016/S1462-0758(01)00043-7
- Foster, S. S. D., Cripps, A. C., Smith-Carington, A., Stewart, W. D. P., & Rosswall, T. (1982). Nitrate leaching to groundwater. *Philosophical Transactions of the Royal Society of London. B,*

Biological Sciences, 296(1082), 477-489. doi:10.1098/rstb.1982.0021

- Foster, S. D., Hirata, R., & Howard, K. W. (2011). Groundwater use in developing cities: policy issues arising from current trends. *Hydrogeology Journal*, 19(2), 271-274.
- Foster, S. S. D., Morris, B. L., & Chilton, P. J. (1999). Groundwater in urban development-a review of linkages and concerns. *IAHS Publication*, 3-12.
- Friberg, N., Harrison, L., O'Hare, M., & Tullos, D. (2017). Restoring rivers and floodplains: Hydrology and sediments as drivers of change. *Ecohydrology*, *10*(5), e1884. doi:10.1002/eco.1884

Fukada, T., Hiscock, K. M., & Dennis, P. F. (2004). A dual-isotope approach to the nitrogen hydrochemistry of an urban aquifer. *Applied Geochemistry*, 19(5), 709-719. https://doi.org/10.1016/j.apgeochem.2003.11.001

- Galloway, D.F. (1999). Addressing the global nitrogen problem. *Abstr Paper Am Chem Soc*, 218, U591.
- Galloway, J. N., Dentener, F. J., Capone, D. G., Boyer, E. W., Howarth, R. W., Seitzinger, S. P., . . . Vöosmarty, C. J. (2004). Nitrogen Cycles: Past, Present, and Future. *Biogeochemistry*, *70*(2), 153-226. doi:10.1007/s10533-004-0370-0
- García-Gil, A., Vázquez-Suñé, E., Sánchez-Navarro, J. Á., Mateo Lázaro, J., & Alcaraz, M. (2015). The propagation of complex flood-induced head wavefronts through a heterogeneous alluvial aquifer and its applicability in groundwater flood risk management. *Journal of Hydrology*, *527*, 402-419. https://doi.org/10.1016/j.jhydrol.2015.05.005
- Gardner, C. (1991). Water regime of river meadows: Yarnton Mead case study. *Report to MAFF, Institute of Hydrology, Wallingford, UK*.
- George, D. J. (1992). Rising groundwater: a problem of development in some urban areas of the Middle East. In G. J. H. McCall, D. J. C. Laming, & S. C. Scott (Eds.), *Geohazards: Natural and man-made* (pp. 171-182). Dordrecht: Springer Netherlands.
- Gilvear, D. J., Spray, C. J., & Casas-Mulet, R. (2013). River rehabilitation for the delivery of multiple ecosystem services at the river network scale. *Journal of Environmental Management, 126*, 30-43. https://doi.org/10.1016/j.jenvman.2013.03.026
- Gooddy, D. C., Clay, J. W., & Bottrell, S. H. (2002). Redox-driven changes in porewater chemistry in the unsaturated zone of the chalk aquifer beneath unlined cattle slurry lagoons. *Applied Geochemistry*, *17*(7), 903-921. https://doi.org/10.1016/S0883-2927(02)00055-0
- Gooddy, D. C., & Hinsby, K. (2008). Organic quality of groundwaters. *Natural groundwater quality*, 59-70.
- Gooddy, D. C., Macdonald, D. M. J., Lapworth, D. J., Bennett, S. A., & Griffiths, K. J. (2014). Nitrogen sources, transport and processing in peri-urban floodplains. *Science of the Total Environment, 494-495*, 28-38. https://doi.org/10.1016/j.scitotenv.2014.06.123
- Gooddy, D. C., Withers, P. J. A., McDonald, H. G., & Chilton, P. J. (1998). Behaviour and Impact of Cow Slurry Beneath a Storage Lagoon: II. Chemical Composition of Chalk Porewater after 18 Years. *Water, Air, and Soil Pollution, 107*(1), 51-72. doi:10.1023/a:1004975615804
- Gormly, J. R., & Spalding, R. F. (1979). Sources and Concentrations of Nitrate-Nitrogen in Ground Water of the Central Platte Region, Nebraska. *Ground Water*, *17*(3), 291-301. doi:10.1111/j.1745-6584.1979.tb03323.x
- Gotkowitz, M., Attig, J., & McDermott, T. (2014). Groundwater flood of a river terrace in southwest Wisconsin, USA. *Hydrogeology Journal*, 1-12. doi:10.1007/s10040-014-1129-x
- Gowing, D., & Youngs, E. (2005). The requirements of Apium repens–an ecohydrological assessment. *English Nature Final Report February*.
- Gregory, K. J. (2006). The human role in changing river channels. *Geomorphology*, 79(3), 172-191. https://doi.org/10.1016/j.geomorph.2006.06.018
- Gregory, S. V., Swanson, F. J., McKee, W. A., & Cummins, K. W. (1991). An Ecosystem Perspective of Riparian Zones. *BioScience*, 41(8), 540-551. doi:10.2307/1311607
- Grizzetti, B., Pistocchi, A., Liquete, C., Udias, A., Bouraoui, F., & van de Bund, W. (2017). Human pressures and ecological status of European rivers. *Scientific Reports*, 7(1), 205.

doi:10.1038/s41598-017-00324-3

- Guida, R. J., Remo, J. W. F., & Secchi, S. (2016). Tradeoffs of strategically reconnecting rivers to their floodplains: The case of the Lower Illinois River (USA). *Science of the Total Environment, 572*, 43-55. https://doi.org/10.1016/j.scitotenv.2016.07.190
- Güneralp, B., Güneralp, İ., & Liu, Y. (2015). Changing global patterns of urban exposure to flood and drought hazards. *Global Environmental Change*, 31, 217-225. doi.org/10.1016/j.gloenvcha.2015.01.002
- Habets, F., Gascoin, S., Korkmaz, S., Thiery, D., Zribi, M., Amraoui, N., . . . Viennot, P. (2010). Multi-model comparison of a major flood in the groundwater-fed basin of the Somme River (France). *Hydrology and Earth System Sciences, 14*(1), 99-117. doi:10.5194/hess-14-99-2010
- Hammersmark, C. T., Rains, M. C., Wickland, A. C., & Mount, J. F. (2009). Vegetation and water-table relationships in a hydrologically restored riparian meadow. *Wetlands, 29*(3), 785-797. doi:10.1672/08-15.1
- Hammond, M. J., Chen, A. S., Djordjević, S., Butler, D., & Mark, O. (2015). Urban flood impact assessment: A state-of-the-art review. *Urban Water Journal, 12*(1), 14-29. doi:10.1080/1573062X.2013.857421
- Harms, T. K., & Grimm, N. B. (2008). Hot spots and hot moments of carbon and nitrogen dynamics in a semiarid riparian zone. *Journal of Geophysical Research: Biogeosciences, 113*(G1). doi:10.1029/2007JG000588
- Hatch, C. E., Fisher, A. T., Ruehl, C. R., & Stemler, G. (2010). Spatial and temporal variations in streambed hydraulic conductivity quantified with time-series thermal methods. *Journal of Hydrology*, *389*(3), 276-288. https://doi.org/10.1016/j.jhydrol.2010.05.046
- Haycock, N. E., & Burt, T. P. (1993). Role of floodplain sediments in reducing the nitrate concentration of subsurface run-off: A case study in the Cotswolds, UK. *Hydrological Processes*, 7(3), 287-295. doi:10.1002/hyp.3360070306
- Haycock, N. E., & Pinay, G. (1993). Groundwater Nitrate Dynamics in Grass and Poplar Vegetated Riparian Buffer Strips during the Winter. *Journal of Environmental Quality, 22*(2), 273-278. doi:10.2134/jeq1993.00472425002200020007x
- Heaton, T. H. E. (1986). Isotopic studies of nitrogen pollution in the hydrosphere and atmosphere: A review. *Chemical Geology: Isotope Geoscience section, 59*, 87-102. https://doi.org/10.1016/0168-9622(86)90059-X
- Heaton, T. H. E., Trick, J. K., & Williams, G. M. (2005). Isotope and dissolved gas evidence for nitrogen attenuation in landfill leachate dispersing into a chalk aquifer. *Applied Geochemistry*, 20(5), 933-945. https://doi.org/10.1016/j.apgeochem.2004.12.004
- Heeren, D. M., Fox, G. A., Fox, A. K., Storm, D. E., Miller, R. B., & Mittelstet, A. R. (2014). Divergence and flow direction as indicators of subsurface heterogeneity and stage-dependent storage in alluvial floodplains. *Hydrological Processes*, 28(3), 1307-1317. doi:10.1002/hyp.9674
- Hester, E. T., & Gooseff, M. N. (2010). Moving beyond the banks: Hyporheic restoration is fundamental to restoring ecological services and functions of streams. *Environmental Science & Technology*, 44 (5), 1521-1525. doi: 10.1021/es902988n
- Hester, E. T., Guth, C. R., Scott, D. T., & Jones, C. N. (2016). Vertical surface water–groundwater exchange processes within a headwater floodplain induced by experimental floods. *Hydrological Processes, 30*(21), 3770-3787. doi:10.1002/hyp.10884
- Hibbs, B. J., & Sharp, J. J. M. (2012). Hydrogeological Impacts of Urbanization. *Environmental and Engineering Geoscience*, 18(1), 3-24. doi:10.2113/gseegeosci.18.1.3
- Hill, A. R., & Duval, T. P. (2009). Beaver dams along an agricultural stream in southern Ontario, Canada: their impact on riparian zone hydrology and nitrogen chemistry. *Hydrological Processes*, 23(9), 1324-1336. doi:10.1002/hyp.7249
- Hiscock, K.M. & Grischek, T. (2002). Attenuation of groundwater pollution by bank filtration. *Journal* of Hydrology, 266(3-4), 139-144. doi.org/10.1016/S0022-1694(02)00158-0
- Hodgson, R. (2008). Groundwater and flooding in Oxford: examining key soil processes. University of

Reading MSc dissertation.

- Hoehn, E. (2002). Hydrogeological issues of riverbank filtration–a review. Riverbank filtration: understanding contaminant biogeochemistry and pathogen removal, edited by: Ray, C., NATO Sci. Ser., IV. *Earth Environ. Sci, 14,* 17-42.
- House, A. R., Thompson, J. R., Sorensen, J. P. R., Roberts, C., & Acreman, M. C. (2016). Modelling groundwater/surface water interaction in a managed riparian chalk valley wetland. *Hydrological Processes*, *30*(3), 447-462. doi:10.1002/hyp.10625
- Howard, K. W. F., & Gelo, K. K. (2002). Intensive groundwater use in urban areas: the case of megacities. *Intensive use of groundwater: challenges and opportunities*, 484.
- Hucks Sawyer, A., Bayani Cardenas, M., Bomar, A., & Mackey, M. (2009). Impact of dam operations on hyporheic exchange in the riparian zone of a regulated river. *Hydrological Processes*, 23(15), 2129-2137. doi:10.1002/hyp.7324
- Hughes, A. G., Vounaki, T., Peach, D. W., Ireson, A. M., Jackson, C. R., Butler, A. P., . . . Wheater, H. S. (2011). Flood risk from groundwater: examples from a Chalk catchment in southern England. *Journal of Flood Risk Management*, 4(3), 143-155. doi:10.1111/j.1753-318X.2011.01095.x
- Hunter, N. M., Bates, P. D., Horritt, M. S., & Wilson, M. D. (2007). Simple spatially-distributed models for predicting flood inundation: A review. *Geomorphology*, *90*(3–4), 208-225. http://dx.doi.org/10.1016/j.geomorph.2006.10.021
- Institute of Hydrology (1987). Worton Rectory Farm (Oxon) feasibility study on the likely effects of proposed gravel scheme on Yarnton and Pixey Mead SSSI. *Institute of Hydrology Report*.
- Irvine, D. J., Brunner, P., Franssen, H.-J. H., & Simmons, C. T. (2012). Heterogeneous or homogeneous? Implications of simplifying heterogeneous streambeds in models of losing streams. *Journal of Hydrology*, 424-425, 16-23. https://doi.org/10.1016/j.jhydrol.2011.11.051
- Jackson, B. M., Browne, C. A., Butler, A. P., Peach, D., Wade, A. J., & Wheater, H. S. (2008). Nitrate transport in Chalk catchments: monitoring, modelling and policy implications. *Environmental Science & Policy*, *11*(2), 125-135. https://doi.org/10.1016/j.envsci.2007.10.006
- Jackson, B. M., Wheater, H. S., Wade, A. J., Butterfield, D., Mathias, S. A., Ireson, A. M., . . . Whitehead, P. G. (2007). Catchment-scale modelling of flow and nutrient transport in the Chalk unsaturated zone. *Ecological Modelling*, *209*(1), 41-52. https://doi.org/10.1016/j.ecolmodel.2007.07.005
- Jackson, C., & Spink, A. (2004). User's manual for the groundwater flow model ZOOMQ3D. *British Geological Survey Internal Report*, IR/04/140.
- Jackson, D. L., & McLeod, C. R. (Eds.). (2000). Handbook on the UK status of EC habitats directive interest features: provisional data on the UK distribution and extent of Annex I habitats and the UK distribution and population size of Annex II species: version 1. Joint Nature Conservation Committee.
- Jain, S. K., Mani, P., Jain, S. K., Prakash, P., Singh, V. P., Tullos, D., . . . Dimri, A. P. (2018). A Brief review of flood forecasting techniques and their applications. *International Journal of River Basin Management*, *16*(3), 329-344. doi:10.1080/15715124.2017.1411920
- Jakovljev, V. V., Svirenko, L. P., Chebanov, O. J., & Spirin, O. I. (2002). Rising Groundwater Levels in North-Eastern Ukraine: Hazardous Trends in Urban Areas. In K. W. F. Howard & R. G. Israfilov (Eds.), Current Problems of Hydrogeology in Urban Areas, Urban Agglomerates and Industrial Centres (pp. 221-241). Dordrecht: Springer Netherlands.
- Jarvie, H. P., Neal, C., & Withers, P. J. A. (2006). Sewage-effluent phosphorus: A greater risk to river eutrophication than agricultural phosphorus? *Science of the Total Environment, 360*(1), 246-253. https://doi.org/10.1016/j.scitotenv.2005.08.038
- Jha, A. K., Bloch, R., & Lamond, J. (2012). Cities and flooding: a guide to integrated urban flood risk management for the 21st century. The World Bank.
- Johnston Associates (1989). A Status Report on the Nation's Floodplain Management Activity: An Interim Report Prepared for Interagency Task Force on Floodplain Management. US

Government Printing Office, Washington, DC.

Jones, M. (2007). Rising groundwater in central London. Water and Sewerage Journal 4: 35-36.

- Jongman, B., Ward, P. J., & Aerts, J. C. J. H. (2012). Global exposure to river and coastal flooding: Long term trends and changes. *Global Environmental Change*, *22*(4), 823-835. https://doi.org/10.1016/j.gloenvcha.2012.07.004
- Jung, M., Burt, T. P., & Bates, P. D. (2004). Toward a conceptual model of floodplain water table response. *Water Resources Research*, *40*(12). doi:10.1029/2003wr002619
- Junk, W. J., Bayley, P. B., & Sparks, R. E. (1989). The flood pulse concept in river-floodplain systems. *Canadian special publication of fisheries and aquatic sciences*, 106(1), 110-127.
- Jurado, A., Borges, A. V., & Brouyère, S. (2017). Dynamics and emissions of N2O in groundwater: A review. *Science of the Total Environment, 584-585*, 207-218. https://doi.org/10.1016/j.scitotenv.2017.01.127
- Karamanos, R. E., & Rennie, D. A. (1978). NITROGEN ISOTOPE FRACTIONATION DURING AMMONIUM EXCHANGE REACTIONS WITH SOIL CLAY. *Canadian Journal of Soil Science*, *58*(1), 53-60. doi:10.4141/cjss78-005
- Karpf, C., & Krebs, P. (2011). Quantification of groundwater infiltration and surface water inflows in urban sewer networks based on a multiple model approach. *Water research*, 45(10), 3129-3136.
- Kellogg, D. Q., Gold, A. J., Groffman, P. M., Addy, K., Stolt, M. H., & Blazejewski, G. (2005). In Situ Ground Water Denitrification in Stratified, Permeable Soils Underlying Riparian Wetlands. *Journal of Environmental Quality*, 34(2), 524-533. doi:10.2134/jeq2005.0524
- Kendall, C. (1998). Chapter 16 Tracing Nitrogen Sources and Cycling in Catchments. In C. Kendall & J. J. McDonnell (Eds.), *Isotope Tracers in Catchment Hydrology* (pp. 519-576). Amsterdam: Elsevier.
- Kessler, H., Mathers, S., & Sobisch, H.-G. (2009). The capture and dissemination of integrated 3D geospatial knowledge at the British Geological Survey using GSI3D software and methodology. *Computers & Geosciences*, 35(6), 1311-1321. https://doi.org/10.1016/j.cageo.2008.04.005
- Kidmose, J., Troldborg, L., Refsgaard, J. C., & Bischoff, N. (2015). Coupling of a distributed hydrological model with an urban storm water model for impact analysis of forced infiltration. *Journal of Hydrology*, *525*, 506-520. https://doi.org/10.1016/j.jhydrol.2015.04.007
- Kingsford, R. T. (2000). Ecological impacts of dams, water diversions and river management on floodplain wetlands in Australia. *Austral Ecology*, *25*(2), 109-127. doi:10.1046/j.1442-9993.2000.01036.x
- Klijn, F., Samuels, P., & Van Os, A. (2008). Towards flood risk management in the EU: State of affairs with examples from various European countries. *International Journal of River Basin Management, 6*(4), 307-321. doi:10.1080/15715124.2008.9635358
- Krause, B., Culmsee, H., Wesche, K., Bergmeier, E., & Leuschner, C. (2011). Habitat loss of floodplain meadows in north Germany since the 1950s. *Biodiversity and Conservation*, 20(11), 2347-2364. doi 10.1007/s10531-011-9988-0
- Kolbe, T., De Dreuzy, J. R., Abbott, B. W., Aquilina, L., Babey, T., Green, C. T., Fleckenstein, J.H., Labasque, T., Laverman, A.M., Marçais, J. & Peiffer, S. (2019). Stratification of reactivity determines nitrate removal in groundwater. Proceedings of the National Academy of Sciences, 116(7), 2494-2499. doi.org/10.1073/pnas.1816892116
- Kollet, S. J., & Maxwell, R. M. (2006). Integrated surface–groundwater flow modeling: A free-surface overland flow boundary condition in a parallel groundwater flow model. Advances in Water Resources, 29(7), 945-958.
- Krause, B., Culmsee, H., Wesche, K., Bergmeier, E., & Leuschner, C. (2011). Habitat loss of floodplain meadows in north Germany since the 1950s. *Biodiversity and Conservation*, 20(11), 2347-2364. doi:10.1007/s10531-011-9988-0

- Krause, S., Bronstert, A., & Zehe, E. (2007). Groundwater–surface water interactions in a North German lowland floodplain – Implications for the river discharge dynamics and riparian water balance. *Journal of Hydrology*, 347(3), 404-417. https://doi.org/10.1016/j.jhydrol.2007.09.028
- Krause, S., Hannah, D. M., Fleckenstein, J. H., Heppell, C. M., Kaeser, D., Pickup, R., . . . Wood, P. J. (2011). Inter-disciplinary perspectives on processes in the hyporheic zone. *Ecohydrology*, 4(4), 481-499. doi:10.1002/eco.176
- Krause, S., Jacobs, J., Voss, A., Bronstert, A., & Zehe, E. (2008). Assessing the impact of changes in landuse and management practices on the diffuse pollution and retention of nitrate in a riparian floodplain. *Science of the Total Environment, 389*(1), 149-164. https://doi.org/10.1016/j.scitotenv.2007.08.057
- Kreibich, H., & Thieken, A. H. (2008). Assessment of damage caused by high groundwater inundation. *Water Resources Research*, 44(9). doi:10.1029/2007WR006621
- Kummu, M., de Moel, H., Ward, P. J., & Varis, O. (2011). How Close Do We Live to Water? A Global Analysis of Population Distance to Freshwater Bodies. *PLOS ONE, 6*(6), e20578. doi:10.1371/journal.pone.0020578
- Lane, S. N. (2017). Natural flood management. *Wiley Interdisciplinary Reviews: Water, 4*(3), e1211. doi:10.1002/wat2.1211
- Laner, D., Fellner, J., & Brunner, P. H. (2009). Flooding of municipal solid waste landfills An environmental hazard? *Science of the Total Environment, 407*(12), 3674-3680. https://doi.org/10.1016/j.scitotenv.2009.03.006
- Lapworth, D. J., Gooddy, D. C., Allen, D., & Old, G. H. (2009). Understanding groundwater, surface water, and hyporheic zone biogeochemical processes in a Chalk catchment using fluorescence properties of dissolved and colloidal organic matter. *Journal of Geophysical Research: Biogeosciences, 114*(G3). doi:10.1029/2009JG000921
- Lapworth, D. J., Gooddy, D. C., Kent, F., Heaton, T. H. E., Cole, S. J., & Allen, D. (2013). A combined geochemical and hydrological approach for understanding macronutrient sources. *Journal of Hydrology, 500*, 226-242. https://doi.org/10.1016/j.jhydrol.2013.07.006
- Lawrence, A. R., Gooddy, D. C., Kanatharana, P., Meesilp, W., & Ramnarong, V. (2000). Groundwater evolution beneath Hat Yai, a rapidly developing city in Thailand. *Hydrogeology Journal*, 8(5), 564-575. doi:10.1007/s10040000098
- Lee, M. H., Lee, S.-H., Kim, E.-S., Eoh, W., Chung, S.-S., & Lee, C.-S. (2015). Survival-Related Factors of Spinal Metastasis with Hepatocellular Carcinoma in Current Surgical Treatment Modalities: A Single Institute Experience. *J Korean Neurosurg Soc, 58*(5), 448-453. doi:10.3340/jkns.2015.58.5.448
- Lerner, D. N., & Barrett, M. H. (1996). Urban Groundwater Issues In The United Kingdom. *Hydrogeology Journal*, 4(1), 80-89. doi:10.1007/s100400050096
- Lewandowski, J., Lischeid, G., & Nützmann, G. (2009). Drivers of water level fluctuations and hydrological exchange between groundwater and surface water at the lowland River Spree (Germany): field study and statistical analyses. *Hydrological Processes, 23*(15), 2117-2128. doi:10.1002/hyp.7277
- Lewin, J., & Ashworth, P. J. (2014). The negative relief of large river floodplains. *Earth-Science Reviews, 129*, 1-23. https://doi.org/10.1016/j.earscirev.2013.10.014
- Leyer, I. (2005). Predicting plant species' responses to river regulation: the role of water level fluctuations. *Journal of Applied Ecology*, *42*(2), 239-250. doi:10.1111/j.1365-2664.2005.01009.x
- Liang, D., Falconer, R. A., & Lin, B. (2007). Coupling surface and subsurface flows in a depth averaged flood wave model. *Journal of Hydrology, 337*(1), 147-158. https://doi.org/10.1016/j.jhydrol.2007.01.045
- Liu, B., & Coulthard, T. J. (2017). Modelling the interaction of aeolian and fluvial processes with a combined cellular model of sand dunes and river systems. *Computers & Geosciences, 106*, 1-

9. doi:10.1016/j.cageo.2017.05.003

- Locatelli, L., Mark, O., Mikkelsen, P. S., Arnbjerg-Nielsen, K., Deletic, A., Roldin, M., & Binning, P. J. (2017). Hydrologic impact of urbanization with extensive stormwater infiltration. *Journal of Hydrology*, *544*, 524-537. https://doi.org/10.1016/j.jhydrol.2016.11.030
- Loheide II, S. P., & Gorelick, S. M. (2007). Riparian hydroecology: A coupled model of the observed interactions between groundwater flow and meadow vegetation patterning. *Water Resources Research*, *43*(7). doi:10.1029/2006WR005233
- Lorah, M. M., Cozzarelli, I. M., & Böhlke, J. K. (2009). Biogeochemistry at a wetland sediment–alluvial aquifer interface in a landfill leachate plume. *Journal of Contaminant Hydrology*, *105*(3), 99-117. https://doi.org/10.1016/j.jconhyd.2008.11.008
- Lyngkild, J., & Christensen, T. H. (1992). Redox zones of a landfill leachate pollution plume (Vejen, Denmark). *Journal of Contaminant Hydrology*, *10*(4), 273-289. https://doi.org/10.1016/0169-7722(92)90011-3
- MacDonald, A., Lapworth, D., Hughes, A., Auton, C., Maurice, L., Finlayson, A., & Gooddy, D. (2014). Groundwater, flooding and hydrological functioning in the Findhorn floodplain, Scotland. *Hydrology Research*, *45*(6), 755-773.
- MacDonald, A. M., Maurice, L., Dobbs, M. R., Reeves, H. J., & Auton, C. A. (2012). Relating in situ hydraulic conductivity, particle size and relative density of superficial deposits in a heterogeneous catchment. *Journal of Hydrology*, 434, 130-14. doi.org/10.1016/j.jhydrol.2012.01.018
- Macdonald, D.M.J., Bloomfield, J., Hughes, A., MacDonald, A., Adams, B., & McKenzie, A. (2008). Improving the understanding of the risk from groundwater flooding in the UK. *Proceedings of the European Conference on Flood Risk Management: Research into Practice*. 30 September-2 October 2008, Keble College, Oxford, UK. CRC Press, The Netherlands.
- Macdonald, D. M. J., Dixon, A. J., & Gooddy, D. C. (2018a). Water and nitrate exchange between a managed river and peri-urban floodplain aquifer: Quantification and management implications. *Ecological Engineering*, *123*, 226-237. https://doi.org/10.1016/j.ecoleng.2018.09.005
- Macdonald, D., Dixon, A., Newell, A., & Hallaways, A. (2012). Groundwater flooding within an urbanised flood plain. *Journal of Flood Risk Management, 5*(1), 68-80. doi:10.1111/j.1753-318X.2011.01127.x
- Macdonald, D.M.J., Griffiths, K.J., Lapworth, D.J., Williams, P.J., Stuart, M.E., Gooddy, D.C. (2012). Hydrogeochemical characterisation of a peri-urban floodplain: initial findings. *British Geological Survey Internal Report*, IR/12/049.
- Macdonald, D.M.J., Hall, R., Carden, D., Dixon, A., Cheetham, M., Cornick, S., & Clegg, M. (2007). Investigating the interdependencies between surface and groundwater in the Oxford area to help predict the timing and location of groundwater flooding and to optimise flood mitigation measures. *Proceedings of 42nd Defra Flood and Coastal Management Conference*.
- Macdonald, D. M. J., Jackson, C. R., & Mansour, M. M. (2018b). A groundwater flow model of the alluvial sands and gravels aquifer of the River Thames floodplain in the vicinity of Oxford, UK. *British Geological Survey Open Report*, OR/18/067.
- MacDougall, A. S., & Turkington, R. (2005). ARE INVASIVE SPECIES THE DRIVERS OR PASSENGERS OF CHANGE IN DEGRADED ECOSYSTEMS? *Ecology*, *86*(1), 42-55. doi:10.1890/04-0669
- Macklin, M. G., Jones, A. F., & Lewin, J. (2010). River response to rapid Holocene environmental change: evidence and explanation in British catchments. *Quaternary Science Reviews*, 29(13), 1555-1576. https://doi.org/10.1016/j.quascirev.2009.06.010
- Mainstone, C. P., & Parr, W. (2002). Phosphorus in rivers ecology and management. *Science of the Total Environment, 282-283,* 25-47. https://doi.org/10.1016/S0048-9697(01)00937-8
- Malzone, J. M., Lowry, C. S., & Ward, A. S. (2016). Response of the hyporheic zone to transient groundwater fluctuations on the annual and storm event time scales. Water Resources Research, 52(7), 5301-5321.

Mansour, M., & Hughes, A. (2004). User's manual for the distributed recharge model ZOODRM.

- Marinos, P., & Kavvadas, M. (1997). Rise of the groundwater table when flow is obstructed by shallow tunnels. Groundwater in the urban environment: problems, processes and management. Balkema, Rotterdam, 49-54.
- Mariotti, A., Landreau, A., & Simon, B. (1988). 15N isotope biogeochemistry and natural denitrification process in groundwater: Application to the chalk aquifer of northern France. *Geochimica Et Cosmochimica Acta, 52*(7), 1869-1878. https://doi.org/10.1016/0016-7037(88)90010-5
- Marsh, T. (2008). A hydrological overview of the summer 2007 floods in England and Wales. *Weather, 63*(9), 274-279. doi:10.1002/wea.305
- Marsh, T., & Hannaford, J. (2007). The summer 2007 floods in England and Wales–a hydrological appraisal. *Centre for Ecology & Hydrology Report*.
- Marsh, T., & Hannaford, J. (2008). UK hydrometric register. *Hydrological data UK series, Centre for Ecology & Hydrology, Wallingford, UK*.
- Matula, S., Mekonnen, G. B., Báťková, K., & Nešetřil, K. (2014). Simulations of groundwater-surface water interaction and particle movement due to the effect of weir construction in the subwatershed of the river Labe in the town of Děčín. *Environmental Monitoring and Assessment, 186*(11), 7755-7770. doi:10.1007/s10661-014-3964-6
- McCarty, G. W., Mookherji, S., & Angier, J. T. (2007). Characterization of denitrification activity in zones of groundwater exfiltration within a riparian wetland ecosystem. *Biology and Fertility of Soils*, *43*(6), 691-698. doi:10.1007/s00374-006-0151-0
- McClain, M. E., Boyer, E. W., Dent, C. L., Gergel, S. E., Grimm, N. B., Groffman, P. M., . . . Pinay, G. (2003). Biogeochemical Hot Spots and Hot Moments at the Interface of Terrestrial and Aquatic Ecosystems. *Ecosystems*, 6(4), 301-312. doi:10.1007/s10021-003-0161-9
- McGranahan, G., Satterthwaite, D., & Tacoli, C. (2004). Rural-urban change, boundary problems and environmental burdens. *International Institute for Environment and Development Human Settlements Working Paper Series: Rural-Urban Interactions and Livelihood Strategies*, No. 10.
- McKenzie, A., Rutter, H., & Hulbert, A. (2010). The use of elevation models to predict areas at risk of groundwater flooding. *Geological Society, London, Special Publications, 345*(1), 75-79.
- McKenzie, A.A., & Ward, R.S. (2015). Estimating numbers of properties susceptible to groundwater flooding in England. *British Geological Survey Open Report*, OR/15/016.
- McMahon, T. A., & Finlayson, B. L. (2003). Droughts and anti-droughts: the low flow hydrology of Australian rivers. *Freshwater Biology, 48*(7), 1147-1160. doi:10.1046/j.1365-2427.2003.01098.x
- McMillan, H. K., & Brasington, J. (2007). Reduced complexity strategies for modelling urban floodplain inundation. *Geomorphology*, *90*(3), 226-243. https://doi.org/10.1016/j.geomorph.2006.10.031
- Miller, J. D., & Hutchins, M. (2017). The impacts of urbanisation and climate change on urban flooding and urban water quality: A review of the evidence concerning the United Kingdom. *Journal of Hydrology: Regional Studies, 12*, 345-362. https://doi.org/10.1016/j.ejrh.2017.06.006

Mitsch, W.J. & Gosselink, J.G. (2000). *Wetlands*. New York, USA: Wiley.

- Moftakhari, H. R., AghaKouchak, A., Sanders, B. F., Allaire, M., & Matthew, R. A. (2018). What Is Nuisance Flooding? Defining and Monitoring an Emerging Challenge. *Water Resources Research*, *54*(7), 4218-4227. doi:10.1029/2018wr022828
- Moftakhari, H. R., AghaKouchak, A., Sanders, B. F., & Matthew, R. A. (2017). Cumulative hazard: The case of nuisance flooding. *Earth's Future*, *5*(2), 214-223. doi:10.1002/2016EF000494
- Monk, W. A., Compson, Z. G., Choung, C. B., Korbel, K. L., Rideout, N. K., & Baird, D. J. (2019). Urbanisation of floodplain ecosystems: Weight-of-evidence and network meta-analysis elucidate multiple stressor pathways. Science of The Total Environment.

- Montz, B. (2000). The generation of flood hazards and disasters by urban development of floodplains. *Floods*, *1*, 116-127.
- Morris, S. E., Cobby, D., Zaidman, M., & Fisher, K. (2018). Modelling and mapping groundwater flooding at the ground surface in Chalk catchments. *Journal of Flood Risk Management*, *11*, S251-S268. doi:10.1111/jfr3.12201
- Murray-Hudson, M., Wolski, P., & Ringrose, S. (2006). Scenarios of the impact of local and upstream changes in climate and water use on hydro-ecology in the Okavango Delta, Botswana. *Journal of Hydrology, 331*(1), 73-84. https://doi.org/10.1016/j.jhydrol.2006.04.041
- Naganna, S. R., Deka, P. C., Ch, S., & Hansen, W. F. (2017). Factors influencing streambed hydraulic conductivity and their implications on stream–aquifer interaction: a conceptual review. *Environmental Science and Pollution Research*, 24(32), 24765-24789. doi:10.1007/s11356-017-0393-4
- Naganna, S. R., Deka, P. C., Ch, S., & Hansen, W. F. (2017). Factors influencing streambed hydraulic conductivity and their implications on stream–aquifer interaction: a conceptual review. *Environmental Science and Pollution Research*, 24(32), 24765-24789. doi:10.1007/s11356-017-0393-4
- Naiman, R. J., Bilby, R. E., & Bisson, P. A. (2000). Riparian Ecology and Management in the Pacific Coastal Rain Forest. *BioScience*, *50*(11), 996-1011. doi:10.1641/0006-3568(2000)050[0996:REAMIT]2.0.CO;2
- Nat, D. v. d., Schmidt, A. P., Tockner, K., Edwards, P. J., & Ward, J. V. (2002). Inundation Dynamics in Braided Floodplains: Tagliamento River, Northeast Italy. *Ecosystems*, *5*(7), 0636-0647. doi:10.1007/s10021-002-0170-0
- Naughton, O., Johnston, P. M., McCormack, T., & Gill, L. W. (2017). Groundwater flood risk mapping and management: examples from a lowland karst catchment in Ireland. *Journal of Flood Risk Management, 10*(1), 53-64. doi:10.1111/jfr3.12145
- Neal, C., Jarvie, H. P., Neal, M., Hill, L., & Wickham, H. (2006). Nitrate concentrations in river waters of the upper Thames and its tributaries. *Science of the Total Environment, 365*(1), 15-32. https://doi.org/10.1016/j.scitotenv.2006.02.031
- Neal, J., Schumann, G., Fewtrell, T., Budimir, M., Bates, P., & Mason, D. (2011). Evaluating a new LISFLOOD-FP formulation with data from the summer 2007 floods in Tewkesbury, UK. *Journal of Flood Risk Management, 4*(2), 88-95. doi:10.1111/j.1753-318X.2011.01093.x
- Négrel, P., & Petelet-Giraud, E. (2005). Strontium isotopes as tracers of groundwater-induced floods: the Somme case study (France). *Journal of Hydrology, 305*(1), 99-119. https://doi.org/10.1016/j.jhydrol.2004.08.031
- NERC (2017). Groundwater: the threat beneath our feet. Planet Earth NERC, Winter 2016-17.
- Newell, A. J. (2008). Morphology and Quaternary geology of the Thames floodplain around Oxford. *British Geological Survey Open Report*, OR/08/030.
- Nones, M. (2015). Implementation of the floods directive in selected EU member states. *Water and Environment Journal, 29*(3), 412-418. doi:10.1111/wej.12129
- Ó Dochartaigh, B. É., Archer, N. A. L., Peskett, L., MacDonald, A. M., Black, A. R., Auton, C. A., . . . Bonell, M. (2018). Geological structure as a control on floodplain groundwater dynamics. *Hydrogeology Journal*. doi:10.1007/s10040-018-1885-0
- Oakes, D. B., Young, C. P., & Foster, S. S. D. (1981). The effects of farming practices on groundwater quality in the United Kingdom. *Science of the Total Environment, 21*, 17-30. https://doi.org/10.1016/0048-9697(81)90132-7
- Palmer, M. A., Filoso, S., & Fanelli, R. M. (2014). From ecosystems to ecosystem services: Stream restoration as ecological engineering. *Ecological Engineering*, 65, 62-70. https://doi.org/10.1016/j.ecoleng.2013.07.059
- Panday, S., & Huyakorn, P. S. (2004). A fully coupled physically-based spatially-distributed model for evaluating surface/subsurface flow. *Advances in Water Resources*, 27(4), 361-382. https://doi.org/10.1016/j.advwatres.2004.02.016

- Paris, A., Teatini, P., Venturini, S., Gambolati, G., & Bernstein, A. G. (2010). Hydrological Effects of Bounding the Venice (Italy) Industrial Harbor by a Protection Cutoff Wall: Modeling Study. *Journal of Hydrologic Engineering*, 15(11), 882-891. doi:10.1061/(ASCE)HE.1943-5584.0000258
- Pärn, J., Pinay, G., & Mander, Ü. (2012). Indicators of nutrients transport from agricultural catchments under temperate climate: A review. *Ecological Indicators*, 22, 4-15. https://doi.org/10.1016/j.ecolind.2011.10.002
- Petts, G. E. (1988). Accumulation of fine sediment within substrate gravels along two regulated rivers, UK. *Regulated Rivers: Research & Management, 2*(2), 141-153. doi:10.1002/rrr.3450020208
- Pfeiffer, S. M., Bahr, J. M., & Beilfuss, R. D. (2006). Identification of groundwater flowpaths and denitrification zones in a dynamic floodpain aquifier. *Journal of Hydrology, 325*(1), 262-272. https://doi.org/10.1016/j.jhydrol.2005.10.019
- Pinault, J.-L., Amraoui, N., & Golaz, C. (2005). Groundwater-induced flooding in macroporedominated hydrological system in the context of climate changes. *Water Resources Research*, *41*(5). doi:10.1029/2004WR003169
- Pinay, G., Bernal, S., Abbott, B. W., Lupon, A., Marti, E., Sabater, F., & Krause, S. (2018). Riparian corridors: A new conceptual framework for assessing nitrogen buffering across biomes. *Frontiers in Environmental Science*, 6, 47. doi.org/10.3389/fenvs.2018.00047
- Pinay, G., Black, V. J., Planty-Tabacchi, A. M., Gumiero, B., & Décamps, H. (2000). Geomorphic control of denitrification in large river floodplain soils. *Biogeochemistry*, *50*(2), 163-182. doi:10.1023/a:1006317004639
- Pinay, G., Ruffinoni, C., Wondzell, S., & Gazelle, F. (1998). Change in Groundwater Nitrate Concentration in a Large River Floodplain: Denitrification, Uptake, or Mixing? *Journal of the North American Benthological Society*, *17*(2), 179-189. doi:10.2307/1467961
- Pinder, G. F., Bredehoeft, J. D., & Cooper Jr., H. H. (1969). Determination of Aquifer Diffusivity from Aquifer Response to Fluctuations in River Stage. *Water Resources Research*, 5(4), 850-855. doi:10.1029/WR005i004p00850
- Pinter, N. (2005). One Step Forward, Two Steps Back on U.S. Floodplains. *Science*, 308(5719), 207-208. doi:10.1126/science.1108411
- Pitt, M. (2008). Learning lessons from the 2007 floods (p. 505). London: Cabinet Office.
- Poff, N. L., Allan, J. D., Bain, M. B., Karr, J. R., Prestegaard, K. L., Richter, B. D., . . . Stromberg, J. C. (1997). The Natural Flow Regime. *BioScience*, *47*(11), 769-784. doi:10.2307/1313099
- Potter, K. W. (2006). Small-scale, spatially distributed water management practices: Implications for research in the hydrologic sciences. *Water Resources Research*, *42*(3). doi:10.1029/2005WR004295
- Punalekar, S., Verhoef, A., Tatarenko, I., van der Tol, C., Macdonald, D., Marchant, B., . . . Gowing, D. (2016). Characterization of a Highly Biodiverse Floodplain Meadow Using Hyperspectral Remote Sensing within a Plant Functional Trait Framework. *Remote Sensing*, 8(2), 112.
- Pysek, P., & Prach, K. (1993). Plant Invasions and the Role of Riparian Habitats: A Comparison of Four Species Alien to Central Europe. *Journal of Biogeography, 20*(4), 413-420. doi:10.2307/2845589
- Ramberg, L., Wolski, P. & Krah, M. (2006). Wetlands 26: 677. doi.org/10.1672/0277-5212(2006)26[677:WBAIIA]2.0.CO;2
- Ramli, M.F., Aris, A.Z., Sulaiman, W.N.A., Kura, N.U. & Tukur, A.I. (2017). An overview assessment of the effectiveness and global popularity of some methods used in measuring riverbank filtration. *Journal of hydrology*, 550, 497-515. doi.org/10.1016/j.jhydrol.2017.05.021
- Ramsar & IUCN (1999). Wetlands and global change. Available at: http://www.ramsar.org/key_unfccc_bkgd.htm
- Ranalli, A. J., & Macalady, D. L. (2010). The importance of the riparian zone and in-stream processes in nitrate attenuation in undisturbed and agricultural watersheds A review of the scientific

literature. Journal of Hydrology, 389(3), 406-415.

https://doi.org/10.1016/j.jhydrol.2010.05.045

- Reynolds, R. J. (1987). Diffusivity of Glacial-Outwash Aquifer by the Floodwave-Response Technique. *Groundwater, 25*(3), 290-299. doi:10.1111/j.1745-6584.1987.tb02133.x
- Rivett, M. O., Buss, S. R., Morgan, P., Smith, J. W. N., & Bemment, C. D. (2008). Nitrate attenuation in groundwater: A review of biogeochemical controlling processes. *Water Research*, 42(16), 4215-4232. https://doi.org/10.1016/j.watres.2008.07.020
- Roni, P., & Beechie, T. (Eds.). (2012). Stream and watershed restoration: a guide to restoring riverine processes and habitats. John Wiley & Sons.
- Rushton, K. R., & Al-Othman, A. A. R. (1994). Control of rising groundwater levels in Riyadh, Saudi
 Arabia. In Groundwater problems in urban areas: Proceedings of the International
 Conference organized by the Institution of Civil Engineers and held in London, 2–3 June 1993
 (pp. 299-309). Thomas Telford Publishing.
- Salvadore, E., Bronders, J., & Batelaan, O. (2015). Hydrological modelling of urbanized catchments: A review and future directions. *Journal of Hydrology, 529*, 62-81. https://doi.org/10.1016/j.jhydrol.2015.06.028
- Sampson, C. C., Fewtrell, T. J., Duncan, A., Shaad, K., Horritt, M. S., & Bates, P. D. (2012). Use of terrestrial laser scanning data to drive decimetric resolution urban inundation models. *Advances in Water Resources*, 41, 1-17. doi:10.1016/j.advwatres.2012.02.010
- Sayers, P. B., Horritt, M., Penning-Rowsell, E., & McKenzie, A. (2015). Climate Change Risk Assessment 2017: Projections of future flood risk in the UK. London: Sayers and Partners LLP report for the Committee on Climate Change.
- Schumann, G. J. P., Neal, J. C., Voisin, N., Andreadis, K. M., Pappenberger, F., Phanthuwongpakdee, N., . . Bates, P. D. (2013). A first large-scale flood inundation forecasting model. *Water Resources Research*, 49(10), 6248-6257. doi:10.1002/wrcr.20521
- Sebilo, M., Mayer, B., Grably, M., Billiou, D., & Mariotti, A. (2004). The Use of the Ammonium Diffusion Method for 15N-NH4+ and 15N-NO3<sup/> Measurements: Comparison with Other Techniques. *Environmental Chemistry*, 1(2), 99-103. https://doi.org/10.1071/EN04037
- Silva, S. R., Kendall, C., Wilkison, D. H., Ziegler, A. C., Chang, C. C. Y., & Avanzino, R. J. (2000). A new method for collection of nitrate from fresh water and the analysis of nitrogen and oxygen isotope ratios. *Journal of Hydrology, 228*(1), 22-36. https://doi.org/10.1016/S0022-1694(99)00205-X
- Silvertown, J., Araya, Y., & Gowing, D. (2015). Hydrological niches in terrestrial plant communities: a review. *Journal of Ecology*, *103*(1), 93-108. doi:10.1111/1365-2745.12332
- Sommer, T., Karpf, C., Ettrich, N., Haase, D., Weichel, T., Peetz, J. V., . . . Ullrich, K. (2009). Coupled modelling of subsurface water flux for an integrated flood risk management. *Nat. Hazards Earth Syst. Sci.*, *9*(4), 1277-1290. doi:10.5194/nhess-9-1277-2009
- Sommerwerk, N., Bloesch, J., Paunovi, #x, , M., Baumgartner, C., . . . Tockner, K. (2010). Managing the worlds most international river: the Danube River Basin. *Marine and Freshwater Research*, *61*(7), 736-748. https://doi.org/10.1071/MF09229
- Sophocleous, M. (2002). Interactions between groundwater and surface water: the state of the science. *Hydrogeology Journal*, *10*(1), 52-67. doi:10.1007/s10040-001-0170-8
- Soren, J. (1976). Basement Flooding and Foundation Damage from Water-table Rise in the East New York Section of Brooklyn, Long Island, New York. US Department of the Interior, Geological Survey.
- Stewart, M. D., Bates, P. D., Price, D. A., & Burt, T. P. (1998). Modelling the spatial variability in floodplain soil contamination during flood events to improve chemical mass balance estimates. *Hydrological Processes*, *12*(8), 1233-1255. doi:10.1002/(SICI)1099-1085(19980630)12:8<1233::AID-HYP614>3.0.CO;2-2
- Stromberg, J. C., Lite, S. J., Marler, R., Paradzick, C., Shafroth, P. B., Shorrock, D., . . . White, M. S. (2007). Altered stream-flow regimes and invasive plant species: the Tamarix case. *Global*

Ecology and Biogeography, 16(3), 381-393. doi:10.1111/j.1466-8238.2007.00297.x

- Stuart, M., & Lapworth, D. (2011). A review of processes important in the floodplain setting. *British Geological Survey Open Report*, OR/11/030.
- Stuart, M. E., Gooddy, D. C., Bloomfield, J. P., & Williams, A. T. (2011). A review of the impact of climate change on future nitrate concentrations in groundwater of the UK. *Science of the Total Environment*, 409(15), 2859-2873. https://doi.org/10.1016/j.scitotenv.2011.04.016
- Teng, J., Jakeman, A. J., Vaze, J., Croke, B. F. W., Dutta, D., & Kim, S. (2017). Flood inundation modelling: A review of methods, recent advances and uncertainty analysis. *Environmental Modelling & Software, 90*, 201-216. doi:10.1016/j.envsoft.2017.01.006
- Thacker, F. S. (1968). *The Thames Highway* (Vol. 2): David & Charles.
- Thieken, A. H., Müller, M., Kreibich, H., & Merz, B. (2005). Flood damage and influencing factors: New insights from the August 2002 flood in Germany. *Water Resources Research*, *41*(12). doi:10.1029/2005WR004177
- Thomas, K. L., Lloyd, D., & Boddy, L. (1994). Effects of oxygen, pH and nitrate concentration on denitrification by Pseudomonas species. *FEMS Microbiology Letters*, *118*(1), 181-186.
- Thomaz, S. M., Kovalenko, K. E., Havel, J. E., & Kats, L. B. (2015). Aquatic invasive species: general trends in the literature and introduction to the special issue. *Hydrobiologia*, 746(1), 1-12. doi:10.1007/s10750-014-2150-8
- Thompson, J. R., Iravani, H., Clilverd, H. M., Sayer, C. D., Heppell, C. M., & Axmacher, J. C. (2017). Simulation of the hydrological impacts of climate change on a restored floodplain. *Hydrological Sciences Journal-Journal Des Sciences Hydrologiques*, 62(15), 2482-2510. doi:10.1080/02626667.2017.1390316
- Thorp, J. H., Thoms, M. C., & Delong, M. D. (2006). The riverine ecosystem synthesis: biocomplexity in river networks across space and time. *River Research and Applications, 22*(2), 123-147. doi:10.1002/rra.901
- Tockner, K., Pusch, M., Borchardt, D., & Lorang, M. S. (2010). Multiple stressors in coupled river– floodplain ecosystems. *Freshwater Biology*, *55*(s1), 135-151. doi:10.1111/j.1365-2427.2009.02371.x
- Tockner, K., & Stanford, J. A. (2002). Riverine flood plains: Present state and future trends. *Environmental Conservation, 29*(3), 308-330. doi:10.1017/S037689290200022X
- VanderKwaak, J. E., & Loague, K. (2001). Hydrologic-Response simulations for the R-5 catchment with a comprehensive physics-based model. *Water Resources Research*, *37*(4), 999-1013. doi:10.1029/2000WR900272
- Vázquez-Suñé, E., Sánchez-Vila, X., & Carrera, J. (2005). Introductory review of specific factors influencing urban groundwater, an emerging branch of hydrogeology, with reference to Barcelona, Spain. *Hydrogeology Journal, 13*(3), 522-533. doi:10.1007/s10040-004-0360-2
- Verhoef, A., Fernandez-Galvez, J., Diaz-Espejo, A., Main, B. E., & El-Bishti, M. (2006). The diurnal course of soil moisture as measured by various dielectric sensors: Effects of soil temperature and the implications for evaporation estimates. Journal of Hydrology, 321(1-4), 147-162.
- Vidon, P. (2012). Towards a better understanding of riparian zone water table response to precipitation: surface water infiltration, hillslope contribution or pressure wave processes? *Hydrological Processes, 26*(21), 3207-3215. doi:10.1002/hyp.8258
- Vidon, P., Allan, C., Burns, D., Duval, T. P., Gurwick, N., Inamdar, S., . . . Sebestyen, S. (2010). Hot Spots and Hot Moments in Riparian Zones: Potential for Improved Water Quality Management1. *JAWRA Journal of the American Water Resources Association*, 46(2), 278-298. doi:10.1111/j.1752-1688.2010.00420.x
- Wakida, F. T., & Lerner, D. N. (2005). Non-agricultural sources of groundwater nitrate: a review and case study. *Water Research*, *39*(1), 3-16. https://doi.org/10.1016/j.watres.2004.07.026
- Walling, D. E., & Fang, D. (2003). Recent trends in the suspended sediment loads of the world's rivers. *Global and Planetary Change*, *39*(1), 111-126. https://doi.org/10.1016/S0921-8181(03)00020-1

- Wang, H. F. & Anderson, M. P. (1982). Introduction to Groundwater Modeling: Finite Difference and Finite Element Methods. W. H. Freeman, San Francisco. 237.
- Wang, L., Stuart, M. E., Bloomfield, J. P., Butcher, A. S., Gooddy, D. C., McKenzie, A. A., . . . Williams, A. T. (2012). Prediction of the arrival of peak nitrate concentrations at the water table at the regional scale in Great Britain. *Hydrological Processes*, 26(2), 226-239. doi:10.1002/hyp.8164
- Wang, X., Zhang, G., & Xu, Y. (2015). Impacts of the 2013 Extreme Flood in Northeast China on Regional Groundwater Depth and Quality. *Water*, 7(8), 4575.
- Ward, J. V., & Stanford, J. A. (1995). The serial discontinuity concept: Extending the model to floodplain rivers. *Regulated Rivers: Research & Management, 10*(2-4), 159-168. doi:10.1002/rrr.3450100211
- Ward, P. J., Jongman, B., Aerts, J. C. J. H., Bates, P. D., Botzen, W. J. W., Diaz Loaiza, A., . . .
 Winsemius, H. C. (2017). A global framework for future costs and benefits of river-flood protection in urban areas. *Nature Climate Change*, *7*, 642. doi:10.1038/nclimate3350
- Warner, J., & van Buuren, A. (2011). Implementing Room for the River: narratives of success and failure in Kampen, the Netherlands. *International Review of Administrative Sciences*, 77(4), 779-801. doi:10.1177/0020852311419387
- Wassenaar, L. I. (1995). Evaluation of the origin and fate of nitrate in the Abbotsford Aquifer using the isotopes of 15N and 18O in NO3–. *Applied Geochemistry*, *10*(4), 391-405. https://doi.org/10.1016/0883-2927(95)00013-A
- Weber, R., Watson, A., Forter, M., & Oliaei, F. (2011). Review Article: Persistent organic pollutants and landfills - a review of past experiences and future challenges. *Waste Management & Research, 29*(1), 107-121. doi:10.1177/0734242x10390730
- Werner, M. G. F., Hunter, N. M., & Bates, P. D. (2005). Identifiability of distributed floodplain roughness values in flood extent estimation. *Journal of Hydrology*, 314(1), 139-157. https://doi.org/10.1016/j.jhydrol.2005.03.012
- Werritty, A. (2006). Sustainable flood management: oxymoron or new paradigm? *Area, 38*(1), 16-23. doi:10.1111/j.1475-4762.2006.00658.x
- Wheeler, B. D., Gowing, D. J. G., Shaw, S. C., Mountford, J. O., & Money, R. P. (2004). Ecohydrological guidelines for lowland wetland plant communities. *Environment Agency Report*.
- White, D., & Macdonald, D.M.J. (2015). Review of data relevant to the characterisation of leachate from Burgess Field waste dump, Oxford. *British Geological Survey Internal Report*, IR/15/037.
- Whitehead, P. G., & Crossman, J. (2012). Macronutrient cycles and climate change: Key science areas and an international perspective. *Science of the Total Environment, 434*, 13-17. https://doi.org/10.1016/j.scitotenv.2011.08.046
- Willner, S. N., Otto, C., & Levermann, A. (2018). Global economic response to river floods. *Nature Climate Change*, *8*(7), 594-598. doi:10.1038/s41558-018-0173-2
- Withers, P. J. A., & Lord, E. I. (2002). Agricultural nutrient inputs to rivers and groundwaters in the UK: policy, environmental management and research needs. *Science of the Total Environment, 282-283*, 9-24. https://doi.org/10.1016/S0048-9697(01)00935-4
- Woessner, W. W. (2000). Stream and Fluvial Plain Ground Water Interactions: Rescaling Hydrogeologic Thought. *Ground Water*, *38*(3), 423-429. doi:10.1111/j.1745-6584.2000.tb00228.x
- Wohl, E., Bledsoe, B. P., Jacobson, R. B., Poff, N. L., Rathburn, S. L., Walters, D. M., & Wilcox, A. C. (2015). The Natural Sediment Regime in Rivers: Broadening the Foundation for Ecosystem Management. *BioScience*, 65(4), 358-371. doi:10.1093/biosci/biv002
- Wong, J. S., Freer, J. E., Bates, P. D., Sear, D. A., & Stephens, E. M. (2015). Sensitivity of a hydraulic model to channel erosion uncertainty during extreme flooding. *Hydrological Processes*, 29(2), 261-279. doi:10.1002/hyp.10148
- Younger, P. L., Mackay, R., & Connorton, B. J. (1993). Streambed Sediment as a Barrier to Groundwater Pollution: Insights from Fieldwork and Modelling in the River Thames Basin. *Water and Environment Journal, 7*(6), 577-585. doi:10.1111/j.1747-6593.1993.tb00890.x

- Yu, D., & Lane, S. N. (2006). Urban fluvial flood modelling using a two-dimensional diffusion-wave treatment, part 1: mesh resolution effects. *Hydrological Processes*, 20(7), 1541-1565. doi:10.1002/hyp.5935
- Zheng, Z., Gao, J., Ma, Z., Wang, Z., Yang, X., Luo, X., . . . Fu, G. (2016). Urban flooding in China: main causes and policy recommendations. *Hydrological Processes*, 30(7), 1149-1152. doi:10.1002/hyp.10717